OTIS FLE COPY

WRDC-TR-90-3027

AD-A230 223



90 12 21 008

A Theoretical Approach to Analysis and Design of Efficient Reduced Control for Space Structures

Dr Hayrani Oz

The Ohio State University Dept of Aeronautical & Astronautical Engineering Columbus, Ohio 43210



October 1990

Final Report for the Period October 1986 to October 1990

Approved for public release; distribution unlimited.

FLIGHT DYNAMICS LABORATORY WRIGHT RESEARCH AND DEVELOPMENT CENTER AIR FORCE SYSTEMS COMMAND WRIGHT PATTERSON AIR FORCF BASE, OHIO 45433-6553

# NOTICE

When Government drawings, specifications, or other data are used for any purpose other than in connection with a definitely related Government procurement operation, the United States Government thereby incurs no responsibility or any obligation whatsoever. The fact that the Government may have formulated or in any way supplied the said drawings, specifications, or other data, is not to be regarded by implication, or otherwise as in any manner, as licensing the holder or any other person or corporation; or as conveying any rights or permission to manufacture, use, or sell any patented invention that may in any way be related thereto.

This report is releasable to the National Technical Information Service (NTIS). At NTIS, it will be available to the general public, including foreign nations.

This technical report has been reviewed and is approved for publication.

I are adapted to the

NARENDRA S. KHOT Project Engineer Design & Analysis Methods Group

NELSON D. WOLF, Technical Manager Design & Analysis Methods Group Analysis & Optimization Branch

FOR THE COMMANDER

DAVID K. MILLER, Lt Col, USAF Chief, Analysis & Optimization Branch Structures Division

"If your address has changed, if you wish to be removed from our mailing list, or if the addressee is no longer employed by your organization please notify WRDC/FIBRA, Wright-Patterson AFB OH 45433-6553 to help us maintain a current mailing list".

Copies of this report should not be returned unless return is required by security considerations, contractual obligations, or notice on a specific doucment.

	CUMENTATION PAG	GE	Form Approved OMB No. 0704-0188
Public reporting burden for this collection of info	rmation is estimated to average 1 hour per resp completing and reviewing the collection of info	ponse, including the time for re	viewing instructions, searching existing data sources, refine this burden estimate or any other apert of this
ollection of information, including suggestions fo Javis Highway, Suite 1204, Arlington, VA, 22202-4	or reducing this burden, to Washington Headqu 1302, and to the Office of Management and Bud	uarters Services, Directorate for Iget, Paperwork Reduction Proj	ect (0704-0188), Washington, DC 20503
I. AGENCY USE ONLY (Leave blarik	) 2. REPORT DATE	3. REPORT TYPE ANI	D DATES COVERED
TITLE AND SUBTITLE	15 Oct 90	Final Oct 8	6 - Oct 90
A Theoretical Approach	) to Analysis and Desig	an of	5. FUNDING NUMBERS
Efficient Reduced Cont	rol for Space Structur	ces	PR 2302
. AUTHOR(S)			TA N5
			w0 02
Dr Hayrani Oz			
PERFORMING ORGANIZATION NAM	ME(S) AND ADDRESS(ES)		8. PERFORMING ORGANIZATION
Department of Aeronaut	ical & Astronautical F	Engineering	REFORT NUMBER
The Ohio State Univers	ity		N/A
0010.0003, 0110 40210			
SPONSORING MONITONING AGEN	ICY NAME(S) AND ADDRESS(ES)		10. SPONSORING / MONITORING
Flight Dynamics Labora	tory (WRDC/FIBRA)		AGENCY REPORT NUMBER
Wright Research Develo	opment Center		WRDC-TR-90-3027
wright-ratterson AFB (	/n 40433-6003		
1. SUPPLEMENTARY NOTES			
Narendra S. Khot, WRDO	C/FIBRA, (513) 255-6992	2	
2a. DISTRIBUTION (AVAILABILITY ST	TATEMENT		126. DISTRIBUTION CODE
Approved for public re	lease; distribution ur	limited.	
·			
3. ABSTRACT (Maximum 200 words) The report develops	concepts of efficienc	v for the cont	rol of structural dynamic
systems. The efficienc	ies defined are unique	e nondimensiona.	1 measures of structure
control system performa	nce forming a common b	asis of evalua	tion of such systems.
leveropments are presen	ite element models of	structural svs	ystems and spatially tems. Internal charac-
liscrete, typically fin	ficiencies of structur	serecterar byb	tems are given by devel-
discrete, typically fin terizations of power ef		al control sys	
discrete, typically fin terizations of power ef oping an efficiency mod	es analysis of control	al control systems. The sign	nificance of relationship
discrete, typically fin terizations of power ef oping an efficiency mod of controller efficienc cepts, efficient model/	es analysis of control y modes to structural controller reduction a	al control systems. The sign modes is studio approaches are of	nificance of relationship ed. Based on these con-
discrete, typically fin terizations of power ef oping an efficiency mod of controller efficienc cepts, efficient model/ ries of systems with dy	les analysis of control y modes to structural controller reduction a namic compensators inc	al control sys lers. The sign modes is studio pproaches are of luding reduced	nificance of relationship ed. Based on these con- developed. The efficien- order efficiency-state
discrete, typically fin terizations of power ef oping an efficiency mod of controller efficienc cepts, efficient model/ cies of systems with dy estimators are also dev	les analysis of control y modes to structural controller reduction a namic compensators inc eloped.	al control systems. The sign modes is studio opproaches are de luding reduced	nificance of relationship ed. Based on these con- developed. The efficien- order efficiency-state
discrete, typically fin terizations of power ef oping an efficiency mod of controller efficienc cepts, efficient model/ cies of systems with dy estimators are also dev illustrative exampl CSDL-ACOSS-6 (Model 2)	les analysis of control y modes to structural controller reduction a namic compensators inc eloped. es based on the ACOSS- as typical large space	al control systems. The sign modes is studio approaches are luding reduced tetrahedral	nificance of relationship ed. Based on these con- developed. The efficien- order efficiency-state structure and the
discrete, typically fin terizations of power ef oping an efficiency mod of controller efficienc cepts, efficient model/ cies of systems with dy estimators are also dev illustrative exampl CSDL-ACOSS-6 (Model 2) establishes that effici	les analysis of control y modes to structural controller reduction a namic compensators inc eloped. es based on the ACOSS- as typical large space ency of structural com	al control systems. The sign modes is studie opproaches are of cluding reduced 4 tetrahedral se structures, and atrol systems is	nificance of relationship ed. Based on these con- developed. The efficien- order efficiency-state structure and the re given. The research s a fundamental concept
discrete, typically fin terizations of power ef oping an efficiency mod of controller efficienc cepts, efficient model/ cies of systems with dy estimators are also dev illustrative exampl CSDL-ACOSS-6 (Model 2) establishes that effici that must be addressed	les analysis of control y modes to structural controller reduction a namic compensators inc eloped. es based on the ACOSS- as typical large space ency of structural con in any structure contr	al control systems is studied modes is studied opproaches are de cluding reduced 4 tetrahedral se structures, a tool systems is	nificance of relationship ed. Based on these con- developed. The efficien- order efficiency-state structure and the re given. The research s a fundamental concept gn. The efficiency
discrete, typically fin terizations of power ef oping an efficiency mod of controller efficienc cepts, efficient model/ cies of systems with dy estimators are also dev illustrative exampl CSDL-ACOSS-6 (Model 2) establishes that effici that must be addressed measures also establish finite dimensional cont	les analysis of control y modes to structural controller reduction a namic compensators inc eloped. es based on the ACOSS- as typical large space ency of structural con in any structure contr the performance of ∞- rol design models	al control system lers. The sign modes is studio approaches are of luding reduced 4 tetrahedral structures, and trol systems is ol system design dimensional system	nificance of relationship ed. Based on these con- developed. The efficien- order efficiency-state structure and the re given. The research s a fundamental concept gn. The efficiency stems on the basis of
discrete, typically fin terizations of power ef oping an efficiency mod of controller efficienc cepts, efficient model/ cies of systems with dy estimators are also dev illustrative exampl CSDL-ACOSS-6 (Model 2) establishes that effici that must be addressed neasures also establish finite dimensional cont 4. SUBJECT TERMS	es analysis of control y modes to structural controller reduction a namic compensators inc eloped. es based on the ACOSS- as typical large space ency of structural con in any structure contr the performance of ∞- rol design models.	al control systems is studied modes is studied opproaches are de luding reduced tetrahedral a structures, a trol systems is col system designed imensional systems	nificance of relationship ed. Based on these con- developed. The efficien- order efficiency-state structure and the re given. The research s a fundamental concept gn. The efficiency stems on the basis of
discrete, typically fin terizations of power ef oping an efficiency mod of controller efficienc cepts, efficient model/ cies of systems with dy estimators are also dev illustrative exampl CSDL-ACOSS-6 (Model 2) establishes that effici that must be addressed measures also establish finite dimensional cont 4. SUBJECT TERMS fultivariable Controls,	es analysis of control y modes to structural controller reduction a namic compensators inc eloped. es based on the ACOSS- as typical large space ency of structural con in any structure contr the performance of ∞- rol design models.	al control systems is studio modes is studio opproaches are opproaches are opproa	nificance of relationship ed. Based on these con- developed. The efficien- order efficiency-state structure and the re given. The research s a fundamental concept gn. The efficiency stems on the basis of 15. NUMBER OF PAGES 270
discrete, typically fin terizations of power ef oping an efficiency mod of controller efficienc cepts, efficient model/ cies of systems with dy estimators are also dev illustrative exampl CSDL-ACOSS-6 (Model 2) establishes that effici that must be addressed measures also establish finite dimensional cont 4. SUBJECT TERMS fultivariable Controls, Reduced Order System, E	les analysis of control y modes to structural controller reduction a namic compensators inc eloped. es based on the ACOSS- as typical large space ency of structural contr in any structure contr the performance of $\infty$ - rol design models. A Control Structure Int fficiency	al control systems is studied modes is studied opproaches are opproaches are oppr	nificance of relationship ed. Based on these con- developed. The efficien- order efficiency-state structure and the re given. The research s a fundamental concept gn. The efficiency stems on the basis of 15. NUMBER OF PAGES 270 16. PRICE CODE
discrete, typically fin terizations of power ef oping an efficiency mod of controller efficienc cepts, efficient model/ cies of systems with dy estimators are also dev illustrative exampl CSDL-ACOSS-6 (Model 2) establishes that effici that must be addressed measures also establish finite dimensional cont 4. SUBJECT TERMS fultivariable Controls, deduced Order System, E	es analysis of control y modes to structural controller reduction a namic compensators inc eloped. es based on the ACOSS- as typical large space ency of structural con in any structure contr the performance of ∞- rol design models. // Control Structure Int fficiency // (CASSIFICATION 19 OF THIS PAGE	al control systems is studie modes is studie opproaches are opproaches are opproa	nificance of relationship ed. Based on these con- developed. The efficien- order efficiency-state structure and the re given. The research s a fundamental concept gn. The efficiency stems on the basis of 15. NUMBER OF PAGES 270 16. PRICE CODE
discrete, typically fin terizations of power ef oping an efficiency mod of controller efficienc cepts, efficient model/ cies of systems with dy estimators are also dev illustrative exampl CSDL-ACOSS-6 (Model 2) establishes that effici that must be addressed neasures also establish finite dimensional cont 4. SUBJECT TERMS fultivariable Controls, teduced Order System, E OF REPORT Inclassified	es analysis of control y modes to structural controller reduction a namic compensators inc eloped. es based on the ACOSS- as typical large space ency of structural con in any structure contr the performance of $\infty$ - rol design models. Control Structure Int fficiency (CLASSIFICATION 19 OF THIS PAGE Unclassified	al control systems is studie modes is studie opproaches are of luding reduced tetrahedral structures, and trol systems is ol system designed dimensional system eraction, d SECURITY CLASSIFIC OF ABSTRACT Unclassified	nificance of relationship ed. Based on these con- developed. The efficien- order efficiency-state structure and the re given. The research s a fundamental concept gn. The efficiency stems on the basis of 15. NUMBER OF PAGES 270 16. PRICE CODE

#### FOREWORD

This final report was prepared by the Ohio State University, for the Analysis and Optimization Branch (FIBR) of the Wright Research and Development Center. The work was performed under Contract F33615-86-C-3212 which was initiated under Project No. 2032, "Structures", Task No. N5, "Structural Dynamics and Controls". The objective of this contract was to develop mathematical algorithms for designing efficient reduced order controls with particular reference to the large space structures. The program manager was Dr H. Oz of the Department of Aeronautical and Astronautical Engineering. In FIBR, Dr N. S. Khot was the project monitor.

			1	
	Accesi	on For	1	
	NTIS	CRASI	J_	
	DIIC	i AB	[]	
	J din	0:1 Ced 221101:	بب	
)	By D <sub>i</sub> t ib	tio /		-
	A	vailability C	Codes	
	Di.t	Avail a d Special	or	
	A-1			

# TABLE OF CONTENTS

r

Section	Page
1.0 INTRODUCTION	1
2.0 EFFICIENCY OF STRUCTURE-CONTROL SYSTEMS	5 5 8 11 11 14 16 20 26 26 26 26 30
<ul> <li>3.0 EFFICIENCY MODES ANALYSIS OF STRUCTURE CONTROL SYSTEMS</li></ul>	44 44 44 46 46 50 53 53 56 56 56 56 57
<ul> <li>4.0 EFFICIENCY MODEL/CONTROLLER REDUCTION</li></ul>	. 71 . 71 . 72 . 80 . 84 . 87 on 90 . 94 . 94 . 95 . 96
<ul> <li>5.0 EFFICIENCY OF SYSTEMS WITH COMPENSATOR DYNAMICS</li> <li>5.1 Introduction</li> <li>5.2 Efficiency of Structure-Control Systems With Compensator Dynamics-Compensator Efficiency</li> <li>5.3 Reduced-Order Efficiency State Estimators</li> <li>5.4 Residual Dynamic Response to Control Power Spillover</li> </ul>	106 106 107 113

# TABLE OF CONTENTS (Concluded)

Section	1 .	Page
6	OTHER APPLICATIONS OF CONTROLLER EFFICIENCY AND APPLICATION TO CSDL-ACOSS-6	122 122 122 124 126 127
7	CONCLUSIONS	153
	REFERENCES	155
	APPENDIX A: PROOF OF FUNCTIONAL S <sup>R</sup> AS CONTROL POWER	157 160 162
	APPENDIX B: CLOSED-FORM SOLUTION FOR S*	164
	APPENDIX C: COMPUTER RESULTS OF EXAMPLES	166
	Nomenclature for Computer Outputs	166
	Examples 3.1 and 4.1	169
	Example 3.2	176
	Examples 3.3 and 3.6	183
	Examples 3.4 and 4.2	190
	Example 4.4	204
	Example 4.5	221
	Example 6.3	234
	Example 6.4	246

### LIST OF ILLUSTRATIONS

Figure

Figure 2.1:	ACOSS-4 Tetrahedral Structure
Figure 2.2a:	Percent Global efficiencies for ACOSS-4
Figure 2.2b:	Percent Relative Model efficiencies for ACOSS-4
Figure 2.2c:	Model efficiency coefficients for ACOSS-4
Figure 2.3.a:	Control Powers for n=2
Figure 2.3b:	Control Powers for $n=8$
Figure 2.4a:	Responses for n=2 of natural control (continuous input), suboptimal control and evaluation model with m=2, e*=1.27%, e=1.27% $\mu$ =1, sq*=77.67%, sq=98.73%
Figure 2.4b:	Responses for n=8 of natural control (continuous input), and suboptimal control and evaluation model with m=2, e*=5.06%, e=56.8% $\mu$ =11.2, sq*=8.3%, sq=43.2% 40
Figure 2.4c:	Responses for n=12 of natural control (continuous input) and suboptimal control with m=6, e*=31.8%, e=100% $\mu$ =3.13, sq=sq*=0
Figure 2.5a:	Response of evaluation model with efficient model reduction applied for n=8, m=2, e*=14.7%, e=96.58%, $\mu$ =6.57, sq*=0.24, sq=3.45%
Figure 2.5b:	Response of evaluation model with efficient model reduction applied for n=8, m=4, e*=44.7%, e=93.93%, $\mu$ =2.1, sq*=0.14, sq=6.07%
Figure 3.1:	Globally optimal system LOS for n=2 {modes: 11,12}; example 3.1
Figure 3.2:	Suboptimal system LOS for n=2 {modes: 11,12}, m=1 (input #4); example 3.1
Figure 3.3:	Evaluation model LOS for n=2 {modes: 11,12} m=1 (input #4); example 3.1
Figure 3.4:	Globally optimal system LOS for n=2 (modes: 11,12); example 3.3

Figure 3.5:	Suboptimal system LOS for n=2 {modes: 11,12}; m=4 (#1-4); example 3.3
Figure 3.6:	Evaluation model LOS for n=2 (modes: 11,12), m=4 (#1-4); example 3.3
Figure 3.7:	Globally optimal system LOS for n=8 {modes: 1-8} m=2 (#1-2); example 3.4
Figure 3.8:	Suboptimal system LOS for n=8 (modes: 1-8), m=2 (#1-2); example 3.4
Figure 3.9:	Evaluation model LOS for n=8 {modes 1-8}, m=2 (#1-2); example 3.4
Figure 3.10:	Globally optimal system LOS for n=8 {modes: 1-8}, m=4 (#1-4); example 3.5
Figure 3.11:	Suboptimal system LOS for n=8 (modes: 1-8), m=4 (#1-4); example 3.5
Figure 3.12:	Evaluation model LOS for n=8 {modes: 1-8}, m=4 (#1-4); . 4); example 3.5
Figure 4.1:	Globally optimal system LOS for n=8 (modes: 1-8); example 4.4; controller reduction via efficiency filtering for $n_{Pr}=4$
Figure 4.2:	Supoptimal system LOS for n=8 (modes: 1-8), m=4 (#1-4); example 4.4; controller reduction via efficiency filtering for $n_p=4$
Figure 4.3:	Evaluation model LOS for n=8 {modes: 1-8}, m=4 (#1-4), example 4.4; controller reduction via efficiency filtering for $n_P=4$
Figure 4.4:	Globally optimal system LOS, n=8 {modes: 1-8}, example 4.4; controller reduction via efficiency filtering for $n_p=8$ . 100
Figure 4.5:	Suboptimal system LOS, n=8 (modes: 1-8), m=4 (#1-4), example 4.4; controller reduction via efficiency filtering for $n_p=8$
Figure 4.6:	Evaluation model LOS, n=8 {modes: 1-8}, m=4 (#1-4), example 4.4; controller reduction via efficiency filtering for $n_p=8$
Figure 4.7:	Globally optimal system LOS, n=8 (modes: 3-6, 9-12); example 4.5; efficient model selection reduction 103
Figure 4.8:	Suboptimal System LOS, n=8 {modes: 3-6, 9-12}; m=4 (#1-4), example 4.5; efficient model selection (reduction) 104

i T

Figure 4.9:	Evaluation model LOS, n=8 {modes: 3-6, 9-12} m=4 (#1-4), example 4.5; efficient model selection (reduction)	105
Figure 6.1:	Globally optimal system LOS, n=4 {modes: 9-12}, example 6.1; response of an inefficient system e*=8.39%, e=45.67%	130
Figure 6.2:	Suboptimal system LOS, n=4 {modes: 9-12} m=2 (#1,2), example 6.1; response of an inefficent system e*=8.39%, e=45.67%	131
Figure 6.3:	Evaluation for Model LOS, n=4 (modes: 9-12), m=2 (#1-2), example 6.1; response of an inefficient system e*=8.39%, e=45.67%	132
Figure 6.4:	Globally optimal system LOS, n=4 {modes: 9-12}, example 6.2; response of an inefficient system e*=0.2%, e=21.33%	133
Figure 6.5:	Suboptimal system LOS, n=4 (modes: 9-12), m=1 (#5), example 6.2; response of an inefficient system, e*=0.2%, e=21.33%	134
Figure 6.6:	Evaluation model LOS, n=4 (modes: 9-12) m=1 (#5), example 6.2; response of an inefficient system, e*=0.2%, e=21.33%	135
Figure 6.7:	Globally optimal system LOS, n=8 {modes: 1-8} m=1 (#1) for actuator selection via efficiencies	136
Figure 6.8:	Suboptimal system LOS, n=8 {modes: 1-8}, m=1 ( 1) for actuator selection via efficiencies	137
Figure 6.9:	Evaluation model LOS, n=8 {modes: 1-8}, m=1 ( 1) for actuator selection via efficiencies	138
Figure 6.10:	Globally optimal system LOS, n=8 (modes: 1-8), for actuator selection via efficiencies	139
Figure 6.11:	Suboptimal system LOS, n=8 (modes: 1-8), m=12; for actuator selection via efficiencies	140
Figure 6.12:	Evaluation model LOS, n=8 (modes: 1-8), m=12; for actuator selection via efficiencies	141
Figure 6.13:	Globally optimal system LOS, n=8 (modes: 1-8), effect of LQR weighting on efficiency, r=0.05, e*=16.89%, e=56.93%	142
Figure 6.14:	Suboptimal system LOS, n=8 {modes: 1-8}, m=1; effect of LQR weighting on efficiency r=0.05, e*=16.89%, e=56.93%	143

Figure 6.15:	Evaluation mode. LOS, n=8 (modes: 1-8), m=1; effect of LQR weighting on efficiency r=0.05, e*=16.89%, e=56.93% .	144
Figure 6.16:	Globally optimal system LOS, n=8 (modes: 1-8); effect of LQR weighting on efficiency, q=100, $e*=7.81$ %, $e=56.93$ %.	145
Figure 6.17:	Suboptimal system LOS, n=8 (modes: 1-8); effect of LQR weighting on efficiency, q=100, e*=7.81%, e=56.93%	146
Figure 6.18:	Evaluation model LOS, n=8 (modes: 1-8); effect of LQR weighting on efficiency, q=100, e*=7.81%, e=56.93%	147
Figure 6.19:	CSDL - ACOSS - 6 (model 2) structure	148
Figure 6.20:	Globally optimal system, square x-y plane deflection at node 37, n=8, {modes: 12,13,17,21,22,24,28,30}, example 6.3	149
Figure 6.21:	Suboptimal system, square x-y plane deflection at node 37, n=8, {modes: 12,13,17,21,22,24,28,30}, m=4 (#18-21), example 6.3	150
Figure 6.22:	Globally optimal system, square x-y plane deflection at node 37, n=10, {modes: 7-16}, example 6.4	151
Figure 6.23:	Suboptimal system, square x-y plane deflection at node 37, $n=10$ , {modes: 7-16}, $m=10$ (#1-10), example 6.4	152

### LIST OF TABLES

Table	Pa	ge
Table 2.1:	Efficiencies for ACOSS-4 with n=2	32
Table 2.2:	Efficiencies for ACOSS-4 with $n=8$	32
Table 2.3	Efficiencies and Control Powers for ACOSS-4 with an 8th order model obtained via Efficient Model Reduction Approach	r 32

#### 1.0 INTRODUCTION

The purpose of this research is to study a nondimensional measure of structure-control system performance which has the potential to characterize both quantitatively and qualitatively the designer's ability in dealing with some of the problem areas such as assessment of control spillover effects, model/controller order reduction, input configuration, and the interaction between structural and control variables from the structure-control system point of view. The nondimensional measure is defined as the efficiency of the system.

The concept of efficiency is widely encourtered in thermal and thermomechanical sciences. In these fields an efficiency is defined as a nondimensional ratio of two scalars which represent energy or energy related quantities. Typically, one of the scalars characterizes a theoretically ideal, but physically irrealizable process, and the other characterizes an actual physically realized process. The difference between these two scalars represents waste of the total available energy or energy related quantity in realizing the actual physical process and is regarded as an irreversibility inherent in the physical system. Many different forms of efficiency are defined in thermomechanical sciences depending on how the theoretically ideal and the physically realized actual processes are identified in a certain application. Some examples are propulsion efficiency, heat engine and heat pump efficiency, adiabatic compressor efficiency, brake efficiency, overall efficiency, etc. (Ref. 9). In thermomechanical disciplines an important objective is to design a system with high or maximum efficiency consistent with the physical constraints. In practical engineering terms, the thermomechanical systems which use (or convert) the largest fraction of the total available energy in realizing the physical function of that system, thus causing the minimum waste of resources, are desired.

The utility of efficiency concepts as analysis and design tools in thermomechanical systems is well established and provides valuable physical insight to the working of the system. The question arises whether such a time tested concept in thermomechanical discipline can be extended to the field of distributed parameter structure-control systems and yield comparable practical value and physical insight for the analysis and design of such systems. This investigation is a quest in that direction and presents a conceptual framework to establish the usefulness of efficiency concept for structure-control systems.

For structure-control systems, the efficiency concept is defined as the ratio of two control functionals pertaining to the particular structure-control design. The control functionals are judiciously defined to represent the average control powers consumed during the control period. For the purpose of defining efficiencies, four quadratic control power functionals are relevant. Out of these four control powers a relative model efficiency and a global efficiency can be defined for a structure-control system with maximum possible percent efficiency of 100 for each. The two efficiencies are related by a modal efficiency coefficient.

Relative model efficiency is defined as the ratio of a quadratic modal control power functional to a quadratic real control power functional associated with any control design model. Physically, relative model efficiency represents the fraction of real control power expended on the distributed-parameter system (DPS) usefully absorbed by the control design model. The relative model efficiency can be defined for control of the DPS by both spatially discontinuous (discrete) and spatially continuous input configurations. If the dimension of the control design model and t<sup>1</sup> DPS system are identical then the relative model efficiency by definition becomes 100%. Consequently, the relative model

efficiency is predeminantly an indicator of the effect of finite dimensionalization of the control design model in using the available control power and configuration.

On the other hand, global efficiency is defined as the ratio of the real control power of a globally optimal control based on a spatially continuous input profile to the real control power of achieving the same control objectives with a spatially discontinuous input profile. The definition is based on dynamically similar control systems (Ref. 1). Physically, the global efficiency is related to the degree of extra control power associated with a spatially discontinuous input configuration in controlling a DPS versus accomplishing the same task with a spatially continuous input profile and therefore, it should be a predominant indicator of the effectiveness of the input configuration to control DPS.

Modal efficiency coefficient is a proportionality constant between the global efficiency and the relative model efficiency. Thus, with these quantities, the effect of finite dimensionalization of the control design model and the effect of nature of spatial discretization of the input profile in controlling a DPS can be studied.

The concept of efficiency is introduced to address particular issues of DPS control. A good physical understanding of the concept is best brought about by considering a DPS formulation. Therefore, the basic aspects of the concept will be presented from the perspective of a DPS formulation. On the other hand, in practice, for a complex structure only a spatially discrete formulation is explicitly available. Since such a spatially discrete formulation acts as a surrogate for DPS formulation, the transition of the definitions of efficiency to spatially-discrete structural dynamics is also presented. In particular, since spatially discrete models are almost always obtained via the Finite Element

method (FEM), specific interpretations of the concepts for the finite element models are given.

The premise of the concepts presented in this report is that a structurecontrol system must strive to achieve the highest possible efficiencies. In that, the control system should be designed such that as much of the real control power as possible should be channeled to the reduced-order control design model leaving little, or, of possible, no control power spillover for the truncated dynamics. This objective requires maximization of the relative model efficiency. Furthermore, any control design, characterized by a spatially discontinuous input profile, should try to approximate the performance of the globally optimal solution characterized by a dynamically similar spatially continuous input profile. This object requires maximization of the global efficiency. Based on the proposed concepts, an efficient model reduction approach can be proposed in that for a given input configuration, the components which contribute least to the efficiencies can be discarded.

The efficiencies will be shown to be functions of control parameters such as the number of control inputs and locations, method of control, and model order; as well as structural parameters such as natural frequencies and mode shapes. All relevant parameters can be altered to change the efficiency of hybrid structure-control system. An important aspect of the results is that the performance of the entire infinite-dimensional ( $\infty$ -D) DPS can be studied based on quantities computed from the finite-dimensional control design model alone without any knowledge of the truncated dynamics. The concepts introduced are valid irrespective of the theory or method by which the controls are designed. The efficiency concept can be used both for analysis and design of the structurecontrol system.

#### 2.0 EFFICIENCY OF STRUCTURE-CONTROL SYSTEMS

#### 2.1 <u>Introduction</u>

The concept of efficiency for structure-control systems is introduced in this chapter. The developments presented in this chapter constitute the central foundation and concepts around which this investigation revolves. A model efficiency and a global efficiency are defined for the structure-control system from the point of view of control power consumed during control. The efficiency concept is illustrated to be a useful tool in understanding the interaction between the control variables and structure variables and is shown to provide insight to the behavior of the structure-control system. This chapter and the illustrations of the theory presented herein focus on the analysis of a structure-control system. The illustrative examples evaluate performances of several linear quadratic regulator designs for the ACOSS-4 structure (Ref. 10).

In Section 2.2, control objectives for a distributed parameter system (DPS) are stated. Section 2.3 defines a global control power for the DPS and Section 2.4 presents the globally optimal control for the DPS. Suboptimal control of the DPS is discussed in Section 2.5. Section 2.6 generalizes the concepts of control power to spatially discrete systems, in particular finite element models (FEM) of structural systems are noted. The definition of efficiency of a structurecontrol system is introduced in Section 2.7. Also, presented in Section 2.7 is the methodology for the use of efficiency as an analysis and design tool. Finally illustrative examples are given in Section 2.8 for the control of ACOSS-4 tetrahedral truss structure.

## 2.2 <u>Control Objectives for the DPS</u>

We consider the DPS equation of motion

$$m(p)\hat{u}(p,t) + \mathcal{I}[u(p,t)] = f(p,t)$$
(2.1)

where m, u, f, and f are mass distribution,  $dis_{F}lacement$  field vector, stiffness operator matrix, and the input field vector, respectively. p denotes a position vector, which will be suppressed in the following for convenience. In general, (2.1) will represent the three-dimensional partial differential equations of motion. For convenience, we shall assume f > 0.

The eigenvalue problem of (2.1) is

$$\omega_{\mathbf{r}}^2 m \phi_{\mathbf{r}} = \mathcal{L} \phi_{\mathbf{r}}$$
(2.2)

with the orthogonality relations

$$\int_{D(p)} \phi_{\mathbf{r}}^{\mathrm{T}} m \phi_{\mathbf{s}} dD(p) = \delta_{\mathbf{r}\mathbf{s}} , \quad \int \phi_{\mathbf{r}}^{\mathrm{T}} \mathcal{I} \phi_{\mathbf{s}} dD(p) = \omega_{\mathbf{r}}^{2} \delta_{\mathbf{r}\mathbf{s}} , \quad r, \ s = 1, 2, \dots$$

$$(2.3)$$

where  $\omega_r$  is the natural frequency,  $\phi_r$  is the corresponding vector of eigenfunctions, and D(p) is the structural domain. Introducing the modal expansion

$$u = \sum_{\mathbf{r}} \phi_{\mathbf{r}}(p)\xi_{\mathbf{r}}(t) \tag{2.4}$$

(2.1) is transformed to

$$\hat{\xi}(t) + \omega_r^2 \xi_r(t) = f_r(t)$$
  $r = 1, 2, ...$  (2.5)

in the modal configuration space where  $\xi_r$  is a modal configuration coordinate and  $f_r(t)$  represents a modal input coordinate. Similar to modal expansion (2.4) of the displacement field vector, a modal expansion for the input field vector can be written

$$f(p,t) = \sum_{r} m(p)\phi_{r}(p)f_{r}(t)$$
 (2.6)

From (3) and (6) the modal input coordinate  $f_r$  can be obtained as

$$f_{\mathbf{r}}(t) = \int \phi_{\mathbf{r}}^{T}(p) f(p,t) dD \qquad (2.7)$$

The control objective on the  $\infty$ -D DPS is to insure proper allocation of n pairs of eigenvalues of a set of  $\{n\}$  modes from Equation (2.5). Another objective is to minimize control spillover effects so that the response of the control design will not be degraded by excessive control spillover. In improving stability characteristics of time-invariant linear systems, the premise of all control methods is to obtain desirable eigenvalue locations either directly or indirectly. The phrase "eigenvalue allocation" is used here in a general sense to address both direct and indirect means of allocation. As examples, a direct eigenvalue allocation technique is the Simon-Mitter Algorithm or any other technique in which desired eigenvalue locations are imposed as explicit constraints in obtaining the control gains; on the other hand, an indirect eigenvalue allocation technique is to use the linear quadratic regulator (LQR) theory by which eigenvalue positions are obtained indirectly as a by-product of optimizing a performance measure. Hence, the LQR approach is sometimes categorized as an optimal eigenvalue allocation technique. No implication is intended in this paper as to the use or necessity of direct eigenvalue allocation techniques in understanding and applying the proposed concepts of efficiency. In the sequel, the emphasis will be on the generic meaning of the phrase "eigenvalue allocation" as a qualifying phrase for the function of the control system which may have been designed by either a direct or indirect allocation whatever the case may be.

From the control objectives stated above, one would infer the ideal control to be the one by which {n} pairs of closed-loop eigenvalues are located as

desired with the minimum control power, and control spillover is eliminated completely. The solution for such an ideal control for the  $\infty$ -D DPS (2.1) will be stated in Sec. (2.4).

#### 2.3 <u>Global Control Power for the DPS</u>

Whatever one's favorite control design may be to satisfy the control objectives, one can define and compute a <u>global control evaluation functional for the</u>  $\infty$ -D DPS in the form

$$S = \int_{0}^{\infty} \int_{D} m^{-1}(p) f^{T}(p,t) f(p,t) dD dt$$
 (2.8)

(2.8) represents the total quadratic control power expended on the actual DPS. The control power (2.8) is dimensionally the average power consumed by the control design over the control period. Because (2.8) is the control power on the entire DPS, it is recognized as a global quantity. Since the global control power defined by (2.8) is computed from the physical input field vector, it will also be referred to as the real control power  $S^R$  where superscript R denotes "real," hence

$$S^{R} = \iint m^{-1}(p) f^{T}(p,t) f(p,t) dD dt$$
 (2.9)

 $S^R$  can be computed for any given f(p,t) regardless of the details of the control design technique by which it is computed. By a similar motivation, we define a global modal control power functional

$$S_{\alpha}^{\mathsf{M}} = \sum_{r=1}^{\infty} \int_{0}^{\infty} f_{r}^{2}(t) dt \qquad (2.10)$$

where M denotes that the quadratic controls are the modal control input coordinates and the subscript  $\infty$  implies that all modes, hence the entire DPS, are considered in the computation. In contrast to S<sup>R</sup>, which involves the real input field, S<sup>M</sup> seems to represent an abstract quantity since modal inputs are used in its definition. However, S<sup>R</sup> and S<sup>M</sup> are related.

Substituting the modal expansion (2.7) into definition (2.9) and using the orthogonality relations one obtains

$$S^{R} = \iint m^{-1} \sum_{r=1}^{\infty} m \phi_{r} f_{r} \sum_{s=1}^{\infty} m \phi_{s} f_{s} \, dD \, dt = \sum_{r=1}^{\infty} \int f_{r}^{2}(t) \, dt \qquad (2.11)$$

hence

$$S^{\mathsf{R}} = S^{\mathsf{M}}_{\boldsymbol{\varpi}} \tag{2.12}$$

Identity (2.12) defines an invariance property. The global control power for the DPS is frame indifferent; it is the same whether one studies it in the real-space of (2.9) or the modal-space of (2.10).

The global modal control power  $S^{M}_{\varpi}$  can be decomposed into

$$S_{\infty}^{M} = \sum_{r=1}^{n} \int_{0}^{\infty} f_{r}^{2}(t) dt + \sum_{r=n+1}^{\infty} \int_{0}^{\infty} f_{r}^{2}(t) dt \qquad (2.13)$$

$$S_{\bullet}^{\mathsf{M}} = S_{\mathsf{C}}^{\mathsf{M}} + S_{\mathsf{U}}^{\mathsf{M}} \tag{2.14}$$

and from (2.12)

$$S^{\mathsf{R}} = S_{\mathsf{C}}^{\mathsf{M}} + S_{\mathsf{T}}^{\mathsf{M}} \tag{2.15}$$

where the definitions of  $S_C^M$  and  $S_U^M$  should be evident from (2.13) and (2.14).  $S_C^M$  is the portion of the control power  $S^R$  channeled for control of the (n) modes and  $S_U^M$  is the remaining control power channeled into, or better said, spilled over to the uncontrolled modes. If the set of modes (n) are referred to as the control design model,  $S_C^M$  will be termed <u>design model control power</u>. Similarly,  $S_C^m$  will be referred to as <u>control power spillover</u>.

Since  $S^R$ ,  $S^M_U$ ,  $S^M_U$  are positive definite quantities:

$$S^{\mathsf{R}} \ge S_{\mathsf{C}}^{\mathsf{M}} \tag{2.16}$$

where the equality is satisfied if and only if  $S_U^M = 0$ ; that is, when there is no control power spillover.

An important feature of the control of the  $\infty$ -D DPS is imbedded in inequality (2.16). Because S<sup>R</sup> is computed by using the real input f(p,t) applied to the actual structure, and S<sup>M</sup><sub>C</sub> is computed by using the modal inputs to the finite-dimensional control design model, inequality (2.16) relates how the control design model performance stands relative to the actual DPS. It is clear that any mismatch between S<sup>R</sup> and S<sup>M</sup><sub>C</sub> would automatically mean that some of the control power is lost to the residual modes, S<sup>M</sup><sub>C</sub>  $\neq$  0. On the other hand, in accordance with the control objectives, the ideal control system for the DPS would yield S<sup>M</sup><sub>C</sub> = 0, that is, it would minimize the power spillover It remains to address the specifics of how one might realize a minimum global control power to achieve the control objectives. To this end, we shall assume linear state-feedback control.

#### 2.4 Globally Optimal Control for the DPS

In the absence of other objectives and design constraints, it is reasonable to try to achieve the control objectives stated in Sec. (2.2) with the minimum amount of global control power. Hence one can state the optimization problem

Minimize  $S_{\infty}^{M}$ , or equivalently,  $S^{R}$ 

subject to:

$$\rho\{\pm i\omega_{\mathbf{r}}\} \rightarrow \overline{\rho}\{\overline{\alpha}_{\mathbf{r}} \pm i\beta_{\mathbf{r}}\} \qquad r = 1, 2, \dots, \infty$$
(2.17)

This is an optimization problem for the  $\infty$ -D DPS. The uncontrolled eigenvalues  $\pm i\omega_r$  are relocated to specified positions  $\bar{\alpha}_r \pm i\bar{\beta}_r$  where  $\rho\{$  represents an eigenvalue spectrum. There are no restrictions on the constraint values  $\bar{\alpha}_r$  and  $\bar{\beta}_r$ . The case where only n pairs of eigenvalues are relocated to new positions and the remaining residual pairs are not moved, is a special case of the above formulation since we can always write

$$\rho(\pm i\omega_r) \to \overline{\rho}(\overline{\alpha_r} \pm i\overline{\beta_r}) \qquad r = 1, 2, \dots, n$$

(2.18)

$$\rho\{\pm i\omega_r\} \to \overline{\rho}\{\pm i\omega_r\} \qquad r = n+1, \ldots, \infty$$

The solution of the minimum global control power is known to be (Refs. 1,2)

$$f_{\mathbf{x}}^{*}(t) = \{g_{\mathbf{x}_{1}}^{*}\xi_{\mathbf{x}}^{*}(t) + g_{\mathbf{x}_{2}}^{*}\xi_{\mathbf{x}}^{*}(t)\}$$
(2.19)

$$r = 1, 2, ..., \infty$$
  
 $g_{r_1}^* = \omega_r^2 - (\overline{\alpha}_r^2 + \overline{\beta}_r^2), \quad g_{r_2}^* = 2\overline{\alpha}_r$  (2.20)

where (\*) denotes the optimum quantities. Substituting the solution (2.19) into (2.6) one obtains the optimum input field vector which can achieve the desired eigenvalue locations with minimum control power (Ref. 3):

$$f^{*}(p,t) = \int \{G_{1}^{*}(p,p') \ u(p',t) + G_{2}^{*}(p,p') \ \dot{u}(p',t) dD(p')$$
(2.21)

where  $G_1^*$  and  $G_2^*$  are identified as symmetric optimum distributed control influence (Kernel) functions

$$G_{1}^{*}(p,p') = \sum_{r=1}^{\infty} g_{r_{1}}^{*}m(p)\phi_{r}(p)\phi_{r}(p')m(p') = G_{1}^{*}(p',p) \qquad (2.22)$$

$$G_{2}^{*}(p,p') = \sum_{r=1}^{\infty} g_{r}^{*} m(p) \phi_{r}(p) \phi_{r}(p') m(p') = G_{2}^{*}(p',p)$$

We observe the following characteristics of the optimal control solution (2.21): the optimal modal input coordinate  $f_r^*$  is a feedback of only the corresponding r-th modal coordinate, therefore optimal modal control coordinates are independent of each other. This feature of feedback control has come to be known as independent modal-space control (Ref. 4). The corresponding optimal input

field  $f^*(p,t)$  is spatially continuous since the modal synthesis of spatially continuous functions  $m(p)\phi_r(p)$  is a spatially continuous function. Without proof, we also state that controlled DPS under the optimal spatially continuous feedback input has the same eigenfunctions as the uncontrolled DPS, preserving its natural properties (Refs. 1,2). Therefore, the optimal control (2.19-2.21) has also been referred to as Natural Control (Ref. 2).

Specifically, if the desired closed-loop eigenvalues are given as the set (2.18) from (2.19, 2.20) we compute  $g_{r1}^* = g_{r2}^* = 0$  for r=n+1, ...,  $\infty$  which yields  $f_r^* = 0$ , r=n+1, n+2,...,  $\infty$ .

Upon substituting this result into (2.21) we get

$$f^{*}(p,t) = \int \{G_{1_{n}}^{*}(p,p')u(p',t) + G_{2_{n}}^{*}(p,p')\dot{u}(p',t)\} dD(p')$$
(2.23)

where  $G_{1n}^*$  and  $G_{2n}^*$  are the same as (2.22) except that the summation ends at n. The point is that the solution (2.23) does not represent a model truncation, instead the required summations end at n because the remaining terms have been computed to be zero.

The above procedure indicated that optimal control can be found with virtually no effort for any eigenvalue set for the  $\infty$ -D DPS. It remains to check the control spillover effect over the uncontrolled modes r=n+1,... To this end, we substitute the form of f\*(p,t) given by (2.23) into (2.7) which yields upon recognizing the first orthogonality in (2.3),

$$f_{s}(t) = 0 \qquad s = n+1, \dots, \infty$$

Perfect spillover elimination from residual modes is also achieved by the optimal control (2.23). We must point out that the spillover inputs  $f_s(t)$  vanish due to

appearance of  $\phi_r(p)$  in f(p,t) regardless of the functional form of the modal control inputs  $f_r(t)$ , r-1,2,...,n; that is whether modal inputs are independent or not. Therefore, spillover control is ultimately not a matter of what the temporal behavior of the control inputs is, but is a matter of spatial distribution of control. Any other spatial distribution of input would not yield perfect spillover elimination at least theoretically.

Last but not the least important feature of the optimal control is that the solution is unique (Refs. 1,2), therefore it is globally optimal, and it controls the  $\infty$ -D DPS accomplishing the control objectives ideally.

The global control power for the optimal control for the DPS can be evaluated by substituting (2.21) or (2.23) into (2.8)

$$S* = \int m^{-1} f^{*T}(p,t) f^{*}(p,t) dD \qquad (2.24)$$

which yields from (2.12)

$$S^{R} = S_{\varpi}^{M} = S^{*}$$
 (2.25)

### 2.5 <u>Suboptimal Control for the DPS</u>

By definition, any control input field f(p,t) of the form

$$f(p,t) \neq f^{*}(p,t)$$

will have a higher real control power than S\*. The most common suboptimal control is the one that seems most practical to implement; it is the point (or localized) input distribution

$$f(p,t) = \sum_{k=1}^{m} \delta(p - \Delta p_{k}) F_{k}(t)$$
 (2.26)

where  $\delta$  is the spatial Dirac delta function  $\Delta p_k$  is the domain of the influence of the local k-th input and  $F_k(t)$  is the total input over  $\Delta p_k$ .

The real control power for the suboptimal control is again computed by using (2.26) in (2.9). Denoting the total input vector of m inputs by  $F=[F_1, F_2, \ldots, F_m]^T$ , it can be shown that (Ref. 5)

$$S^{R} = \int F(t)^{T}R F(t) dt \qquad R = diag[m(p_{k})\Delta p_{k}]^{-1} \qquad (2.27)$$

and from identity (2.12) the total modal concrol power corresponding to suboptimal control profile (2.26) is

$$S_{\infty}^{\mathsf{M}} = \int F^{\mathsf{T}} R F dt$$

Furthermore, the design model control power for (n) pairs of relocated eigenvalues is

$$S_{C}^{M} = \sum_{r=1}^{n} \int f_{r}^{2}(t) dt = \int_{n}^{n} (t) f_{n}(t) dt = \int F^{T} B_{n}^{T} B_{n} F dt \qquad (2.28)$$

where  $f_n(t)$  is the n-component design model modal input vector generated by the control (2.26) and

$$f_{n}(t) = B_{n}F \quad B_{nrk} = [\phi_{rk}(p_{k})] \quad r = 1, 2, ..., n$$

k=1,2,...,m  $\phi_{rk}(p_k)$  is the area under  $\phi_k$  over  $\overline{\Delta}p_k$ .

The control power spillover due to localized inputs F can be evaluated as

$$S_{U}^{M} = \sum_{r=n+1}^{\infty} \int f_{r}^{2}(t) dt = \int f_{U}^{T}(t) f_{U}(t) dt = \int F^{T} B_{U}^{T} B_{U} F dt \qquad (2.29)$$

where

$$f_{u}(t) = B_{U}F, B_{U_{rk}} = [\phi_{rk}(p_{k})] \qquad r = n+1, \dots, k = 1, 2, \dots, m$$

 $B_n$  and  $B_u$  are the respective partitions of the B matrix for the n-th order control design model and the uncontrolled dynamics.

One does not need to compute the control spillover performance according to (2.29) if  $S^R$  and  $S^M_C$  are already available. For, from (2.27) and (2.28)

$$S_{\rm U}^{\rm M} = S^{\rm R} - S_{\rm C}^{\rm M} = \int F^{\rm T} [R - B_{\rm n}^{\rm T} B_{\rm n}] F dt \qquad (2.30)$$

in which F(t) is directly available from the control design model once it is selected. Equation (2.30) indicates that for a given DPS (hence the mass distribution and therefore the matrix R are known) when a control design model is selected, the control spillover performance can be determined solely on the basis of the control design model. No knowledge of the uncontrolled dynamics is needed. This points out the usefulness of the judicious definition of the control evaluation functional S in the form of (2.9).

The control power spillover for the suboptimal point input profile (2.26) cannot vanish, therefore from (2.15) we deduce

 $S^R > S^M_C$ 

and since  $S^R$  as given by (2.27) is suboptimal

 $S^{R} > S^{*}$ 

### 2.6 <u>Control Powers for Discrete-Systems</u>

Quite often, instead of partial differential equations, a large set of spatially discretized ordinary differential equations are assumed to describe the

dynamics of the DPS adequately, such as the finite element models (FEM) of complex structural systems. Extension of the previous definitions and results to such cases will be useful. We assume that the structural system is described by

$$M\ddot{q}(t) + Kq(t) = Q(t)$$
 (2.31)

instead of (2.1), where q(t) and Q(t) are each N-component generalized coordinates and the generalized forces vectors, respectively. M and K are N x N symmetric, positive definite mass and stiffness matrices, respectively. Equation (2.31) is usually referred to as the N-dimensional evaluation model replacing the  $\infty$ -D DPS, (2.1).

Denoting by E the modal matrix associated with the system (2.31), the modal transformations and the orthogonality relations

$$q = E\xi$$
,  $f(t) = E^{T}Q(t)$   $E^{T}ME = I$ ,  $E^{T}KE = [\omega^{2}]_{N}$  (2.32)

again yield the modal equations of motion (2.5) with the exception that this time  $r=1,2,\ldots,$  N represents the complete system. Again, by definition the total modal control power is

$$S_{\rm N}^{\rm M} = \sum_{\rm r=1}^{\rm N} \int f_{\rm r}^2(t) \ dt = \int f^{\rm T}(t) \ f(t) \ dt \qquad (2.33)$$

where N denotes the total system as opposed to  $\infty$  in (2.13) for the DPS Equation (2.1). The corresponding real control power, after recognizing the invariance property (2.12) is

$$S^{\mathsf{R}} = S^{\mathsf{M}}_{\mathsf{N}} = \int f^{\mathsf{T}} f \, dt = \int Q^{\mathsf{T}} E E^{\mathsf{T}} Q \, dt \qquad (2.34)$$

Noting from the orthogonality relations that  $EE^{T} = M^{-1}$ , for a discrete-system the equivalent of S<sup>R</sup> in (2.9) is

$$S^{\mathsf{R}} = \int Q^{\mathsf{T}} \mathcal{M}^{-1} Q \, dt \tag{2.35}$$

Hence, the generalized input vector Q(t) plays the role of f(p,t).

In particular, if (2.31) represents the FEM equations of motion, the generalized loads vector Q(t) is the vector of joint loadings. If  $F(t)=[F_1 \ F_2 \ \dots \ F_N]^T$  is the real joint inputs vector of f(p,t) one can write

$$Q = DF$$
(2.36)

where D is the joint loads distribution matrix. If f(p,t) is spatially continuous over the whole structural domain, then by necessity, it will yield an input at each joint along every joint degree of freedom. It follows that the equivalent of a spatially continuous input f(p,t) in a FEM setting is tantamount to having a full generalized loads vector Q. On the other hand, if the input f(p,t) is spatially discontinuous, such as in (2.26), that will be tantamount to having a Q with some zero components. If there are m independent joint inputs  $F_1, \ldots, F_m$  as in (2.26) substitution of (2.36) into (2.34) yields

$$S^{R} = \int F(t)^{T} R F(t) dt \qquad R = D^{T} M^{-1} D$$
 (2.37)

Hence, the weighting matrix R in (2.37) is the FEM equivalent of the weighting matrix R in (2.27) for the partial differential equations of motion.

It is clear from the form of R that, given any F(t),  $S^R$  describes the control power for the entire evaluation model. Even if F(t) may have been designed by considering only a reduced (n)th order modal model of the N-th order system (2.31), when  $S^R$  is computed, it will be, according to (2.37), the global

control power for the total N-th order evaluation model, not for the n-th order reduced-control design model.

For an n-th order control design model, it is easy to see that the counterparts of (2.13) and (2.27)-(2.29) are

$$S_{\mathbf{N}}^{\mathbf{M}} = S_{\mathbf{C}}^{\mathbf{M}} + S_{\mathbf{U}}^{\mathbf{M}}$$
(2.38)

where

$$S_{\rm C}^{\rm M} = \int F^{\rm T} D^{\rm T} E_{\rm n} E_{\rm n}^{\rm T} D F dt , \quad S_{\rm U}^{\rm M} = \int F^{\rm T} D^{\rm T} E_{\rm u} E_{\rm u}^{\rm T} D F dt \qquad (2.39)$$

$$S_{\rm U}^{\rm M} = S^{\rm R} - S_{\rm C}^{\rm M} = \int F^{\rm T} D^{\rm T} [M^{-1} - E_{\rm n} E_{\rm n}^{\rm T}] D F dt \qquad (2.40)$$

and  $E_n$  and  $E_u$  are the control design model and the uncontrolled model modal matrices, respectively. Here again, from (2.40), for a given physical system (hence the evaluation model mass matrix M is known) and the n-th order control design model, control spillover performance can be ascertained solely on the basis of the control design model. Specifically, if a FEM is used, the modes that are poorly computed will be inconsequential from the control point of view as long as those modes are in the uncontrolled set.

Finally, it remains to ascertain the counterparts of the globally optimal control and the control power (2.23)-(2.25) for the system of (2.31). From the previous discussions and the nature of the globally optimal control, it is straight forward to obtain

$$S^{\bullet} = \int Q^{\bullet T} M^{-1} Q^{\bullet} dt = \int F^{\bullet T} D^{\bullet T} M^{-1} D^{\bullet} F^{\bullet} dt \qquad (2.41)$$

$$Q^* = M E f^*(t) = D^* F^* , F^* = D^{*-1} M E f^*(t)$$
(2.42)

where f\*(t) is the vector of N-independent modal inputs  $f_r*r=1,2,\ldots,N$  precisely as computed according to Equations (2.19), (2.20). For a FEM, Q\* is the full generalized inputs vector tantamount to having N joint inputs  $F_r*$ ,  $r=1,2,\ldots,N$ , that is as many inputs as the total number of degrees of freedom. In other words, D\* for the globally optimal control must be a full rank N joint loads distribution matrix.

The forms of Q\* and its interpretation makes it clear that a FEM not having inputs at each joint along each joint degree of freedom would correspond to a discontinuous input profile and have a suboptimal performance.

# 2.7 Efficiency of a Structure-Control System

Implementable control designs for large flexible structures such as complex truss-like configurations planned for the space station will inevitably employ spatially discontinuous suboptimal input profiles consisting of a large number of distributed point force and torque actuators. In view of the control objectives and the features of the control powers we discussed heretofore, it would be desirable for any implementable structural control system to channel as much of the real control power as possible to the control design model. In other words, the power spillover  $S_{M}^{M}$  should be minimized by the control design. An equally desirable feature of the control design would be to keep the total control power as small as possible, that is, to keep the real control power of the design as near to the globally optimal control power as possible. These aspects, by necessity, bring about the concept of efficient structure-control designs. An important element of the structure-control design process must be

to find the most efficient structure-control combination for the control objectives.

We define the percent global efficiency of a structure-control system as

$$e^* = \frac{S^*}{S^R} \times 100 \le 100$$
 (2.43)

where  $c^R$  is the real control power of any suboptimal control design with the closed-loop eigenvalue spectrum  $\{\rho\}=\{\rho\}$ . S\* is the globally optimal control power corresponding to the same eigenvalues  $\{cost_{\overline{\varrho}}\}$ . This is to say that the global efficiency is based on the comparison of dynamically similar (Ref. 1) globally optimal and suboptimal control designs for the desired closed-loop eigenvalues. Since  $S^R$  is suboptimal, the upper bound of global efficiency is 100%.

Next, we define the percent relative model efficiency of a structurecontrol system to be

$$e_{*} = \frac{S_{c}^{M}}{S^{R}} \times 100 \le 100_{*}$$
 (2.44)

The relative model efficiency is an indicator of the percentage of the real control power channeled into the control design model, the balance indicating the control power spillover. This efficiency is determined solely by using the properties of the particular control design model. There is no reference to the corresponding globally minimum control. Therefore, we refer to e as the relative model efficiency. A less than perfect model efficiency automatically implies control power wasted to uncontrolled modes. However, as a 100% e means no control spillover, it will not guarantee a 100% global efficiency as  $S^R$  and  $S^*$  may still be different.

The relative model efficiency and the global efficiency are related by

$$e = \mu e * \quad \mu = \frac{S_{\rm c}^{\rm M}}{S *}$$
 (2.45)

where  $\mu$  is defined to be the modal efficiency coefficient.

Complimentary to the above definitions, one can also introduce the global and the model spillover quotients

$$sq* = \frac{S_{U}^{M}}{S*} = \frac{(S^{R} - S_{C}^{M})}{S*} = \frac{1}{e*} - \mu , \quad sq = \frac{S_{U}^{M}}{S^{R}} = 1 - e < 1$$
(2.46)

sq indicates the portion of the real control power lost as control spillover cost. sq\* indicates the control power spillover of the suboptimal design as a fraction of the globally minimum control power that would be expended on the entire DPS. Studies show that the control power spillover a suboptimal control profile can incur, can be many times more than it would take to control the entire system with a spatially continuous optimal input.

Given an initial modal state disturbance  $x_o$  for a stable control system,

$$x_0 = [\xi_1(0) \dot{\xi}_1(0) \dots \xi_n(0) \dot{\xi}_n(0)]$$

The control power for infinite time control are given by

$$S^{R} = x_{0}^{T} P^{R} x_{0} , S_{C}^{M} = x_{0}^{T} P_{C}^{M} x_{0} , S^{*} = x_{0}^{T} P^{*} x_{0}$$
(2.47)

where  $P^{R}$ ,  $P^{M}$  and P\* are the real, modal, and the globally optimal (natural) control power matrices.  $P^{R}$  and  $P^{M}$  can be obtained as the solutions of the associated Lyapunov equations for any suboptimal control design discussed in the preceding sections. The relevant equations are presented in Chapter 3. The natural control power matrix P\* is given in closed form in Ref. (6) and in Appendix B.

The global and relative model efficiencies of any control design can now be computed by using the cost matrices

$$e^{*} = \frac{x_{0}^{\mathrm{T}}P^{*}x_{0}}{x_{0}^{\mathrm{T}}P^{\mathrm{R}}x_{0}} , e = \frac{x_{0}^{\mathrm{T}}P_{\mathrm{c}}^{\mathrm{M}}x_{0}}{x_{0}^{\mathrm{T}}P^{\mathrm{R}}x_{0}} , \mu = \frac{x_{0}^{\mathrm{T}}P_{\mathrm{c}}^{\mathrm{M}}x_{0}}{x_{0}^{\mathrm{T}}P^{*}x_{0}}$$
(2.48)

Each one of the efficiencies, through the power matrices, depends on:

- \* The number, type and locations of localized inputs  $F_k(t)$
- The particular control design technique used to compute the actual spatially discontinuous feedback input F
- \* The order n of the control design model and the closed-loop eigenvalue spectrum  $(\bar{\rho})$
- Structural parameters through the appearance of modal frequencies and mode shapes
- \* The initial modal disturbance state  $x_0$ .

For the analysis and design of structural-control systems via efficiencies, one would typically take the following steps: 1) For any set of selected system variables and parameters mentioned above (such as a given n-th order design model and a control input configuration) obtain a control law by whatever technique or theory deemed appropriate. 2) For the control inputs obtained in step 1, compute

the corresponding  $S^{R}$  and  $S^{M}_{C}$  defined by Eqs. (2.27), (2.28) or Eqs. (2.37) and (2.39), and calculate the relative model efficiency e and the model spillover quotient sq as defined by Eqs. (2.44) and (2.46), respectively. 3) For different values and/or sets of variables and parameters repeat steps 1 and 2, compare the corresponding model efficiencies, simulate if necessary, and identify satisfactory designs consistent with the designer's criteria and constraints. In applying steps 1-3, one should recognize that there is no need for explicit knowledge of the closed-loop eigenvalues if studies based on relative model efficiencies are all that is desired. However, in addition, if global efficiencies e\* and global spillover quotients sq\* are also desired for further consideration, one must then proceed with the following steps: 4) For the controls designed in step 1, compute the corresponding closed-loop eigenvalues  $(\rho)$  if they are not already available. Otherwise, this step is not needed. 5) For the spectrum  $(\bar{\rho})$  found in step 4, compute the modal control gains and the modal inputs given by Eqs. (2.19) and (2.20) of the globally optimal spatially continuous control which is dynamically similar to the control design of step 1. 6) In accordance with Eqs. (2.25), (2.11) and (2.12) obtain the globally minimum real control power S\* possible for the eigenvalue spectrum  $\{\hat{p}\}$  elicited by the control design of step 1. Closed-form solution for S\* is given in Appendix B for any defined  $(\rho)$ . 7) Calculate the global efficiency e\* and global spillover quotient sq\* of the control design of step 1, as defined by Eqs. (2.43) and (2.46) by using S<sup>R</sup> from step 2 and S\* from step 6. If desired, computed the model efficiency coefficient defined in Eq. (2.45). 8) For different values and/or sets of variables and parameters repeat steps 4-7, compare the corresponding global efficiencies and spillover quotients, simulate if necessary, and identify a satisfactory designs. 9) Study the results of steps 3-8 collectively to evaluate the control designs.

The efficiency approach to structure control will liberate the engineer from the need of detailed knowledge of the unmodeled modes. Because the behavior of an infinite dimensional system can be studied and understood by means of computing its efficiencies, which require explicit knowledge of only the finite number of modeled control modes. This feature should make the efficiency approach to control design a practical tool.

With such an approach, it is possible to determine the optimal control input distribution and even the optimum eigenvalue distribution for a given nthorder control design model. For a given input field, the efficiencies can be used to determine the order n of a control design where model orders that yield high efficiencies can be selected. In addition, because the efficiencies are dependent on the input distribution, closed-loop eigenvalues, and other structural and control parameters, different order control models could become more efficient simply by changing the structure-control system's configuration and parameters. In all of these, the objective then should be to maximize the global and relative efficiencies.

In particular, for N-th order discrete evaluation models,  $S_n^M$  represents the control power consumed in controlling n<N modes, while  $S^R$  represents the power consumed by all N modes. Therefore, the relative efficiency e becomes a valid nondimensional measure of the effects of model order reduction. A similar statement holds true for global efficiency. Based on these observations, a closed-loop <u>Efficient Model Reduction Technique</u> can be formulated in that one can propose to retain in the control design model the modes to which the relative model and/or global efficiencies ar most sensitive for any given input configuration.
In contrast to design, efficiencies can be used to evaluate the merits of a given control design since they reflect the effects of many variables of the control problem. In this chapter, we shall illustrate the analyses of some Linear Quadratic Regulator (LQR) control designs based on the efficiency concept. Also, the proposed concept of Efficient Model Reduction Technique will be demonstrated.

## 2.8 <u>Illustrative Example: Control of ACOSS-4 Tetrahedral Truss Structure</u> <u>Analysis via Efficiencies</u>

As a demonstration of the utility of the efficiency concept, the performances of various LQR control designs (step 1) for the ACOSS-4 structure shown in Fig. 2.1 were evaluated for different order modal control design models and different number of inputs (steps 3 and 8). The inputs were located at the pods of the structure. A twelfth order (N=12) evaluation model (2.31), obtained via FEM, was considered. For a given n-th order control design and number of inputs  $1 \le$ m  $\le$  6, the control designs were based on the minimization of the LQR performance measure (step 1)

$$J = \frac{1}{2} \int_{0}^{\infty} (x^{\mathrm{T}} W_{\mathrm{x}} x + F^{\mathrm{T}} W_{\mathrm{c}} F) dt$$

$$W_x = block - diag \begin{bmatrix} 1 & 0 \\ 0 & \omega_r \end{bmatrix} - 2$$
,  $W_c = [1]$ 

where the LQR control design weighting matrix  $W_c$  has no relationship to the weighting matrix R uniquely defined for efficiency purposes. The LQR design approach essentially is an indirect eigenvalue allocation. Instead of requiring

explicit eigenvalue allocation one can implicitly admit the desired eigenvalues  $(\bar{\rho})$  to be those of the LQR solutions for specific choices of weighting parameters q and r. For each LQR steady-state Riccati equation solution, the closed-loop eigenvalues were computed and assigned to be the set  $\{\bar{\rho}\}$  (step 4) where upon the corresponding globally optimal control power S\* was computed (step 6) for the set  $\{\bar{\rho}\}$ . For each LQR solution, S<sup>R</sup> and S<sup>M</sup><sub>C</sub> were computed by solving the associated Lyapunov equations for the closed-loop system (step 2). In the simulations, the 2n-th order initial modal state  $x_0$  was assumed to be  $x_0 = [1 \ \omega_1 \ 1 \ \omega_2 \dots \ 1 \ \omega_n]$  and the uncontrolled modes were initially undisturbed.

For control design models of order n=2, 4, 6, 8, 10, 12 and the input numbers m=1, 2, 3, 4, 5, 6 (steps 3 and 8); e\*, e,  $\mu$ , sq\*, and sq for the chosen LQR weighting matrices,  $W_x$  and  $W_c$ , were computed (steps 2 and 7). The results are shown in Figs. 2.2 (steps 3 and 8). The model selections were made by starting from the lowest structural modes to the higher ones. For brevity, efficiencies and spillover quotients are tabulated only for n=2 and n=8 (Tables 2.1 and 2.2). Efficiencies of other design models and actuator configurations can be inferred from the efficiency curves.

From Figs. 2.2a and b, we observe the interactions among the efficiencies, the order of the control design model, and the input configuration. For a given number of inputs the model efficiency increases with the order of the control design model. However, this is not necessarily true for the global efficiency. For one and two inputs the global efficiency seems to increase with model order, but for three inputs increasing the model order beyond n=4 decreases the global efficiency. We also observe that for a given control design model, increasing the number of inputs increases the global efficiency, but this is not necessarily true for the model efficiency. Indeed for a sixth order design model (n=6), three or more inputs cause a decrease in the model efficiency. Therefore, the third, fourth, fifth, and sixth actuators are located poorly with respect to the truncated modes (n > 7-12) such that more of control energy is lost as control power spillover to cause a drop in the model efficiency. A similar observation is made for 1 and 2 inputs in regards to model efficiency. Hence, it appears that the first 3 actuators represent a critical number of inputs for this particular structure.

The curves of sq\* and sq would describe more vividly the effect of model truncation. However, because sq\* and sq are related to e\*,  $\mu$  and e, for brevity these curves are not shown. Tables 2.1 and 2.2 list some values of the spillover quotients. From the sq\* values given in Table 2.1 one reads, for example, that for m=1 and n=2 the amount of control power lost to model truncation is 152 times the total control power that would be required to control the entire DPS with a spatially continuous optimal input profile (natural control). The control powers S<sup>R</sup>, S<sup>M</sup><sub>C</sub> for the LQR designs and the corresponding dynamically similar natural control powers S<sup>\*</sup> for n=2, 8 are shown in Figs. 2.3a and b. The power plots show that natural control power S<sup>\*</sup> can be significantly lower than the real power S<sup>R</sup> obtained by using LQR designed controls. The distances among the power curves are indicators of the global and model efficiencies and the modal efficiency coefficient.

The LQR values of  $S^R$  decrease monotonically to limit values with increasing number of inputs. The dynamically similar natural control powers  $S^*$  increase monotonically to different limit values. We conjecture that the two designs will not converge because of fundamental differences in their design concepts. The natural control is a distributed partial differential equation control solution according to Equations (2.1), (2.21) and (2.22). On the other hand, the LQR

solution is a discrete control solution based on the a priori reduced-order (truncated) model of the dynamic system. In order for the two optimal solutions to conlerge to different limits, their conceptual framework must be inherently different. This suggests that other than LQR discrete closed-loop laws can be formulated as direct approximations to the closed-loop distributed natural control solution with control powers between the LQR and natural control powers (Ref. 3).

The response profiles for a sensor colocated with the first input are given in Fig. 2.4 for different design model orders and inputs. The response profiles corresponding to the natural (globally optimal) control with continuously distributed input, suboptimal control of the n-th order control design model with m point inputs and the evaluation model, which includes the control spillover effects of the suboptimal control, are superposed in Figs. 2.4a, b and c for comparison purposes. It is seen that <u>almost identical responses</u> can be obtained with drastically different control powers. The similarity between the responses of the (suboptimal) control design model and the evaluation model for n=8, m=2in Fig 2.4b may suggest that model truncation is insignificant. This is true from an output viewpoint. However, there still exists a considerable inefficiency in the control design due to 43.2% control power wasted (sq=43.2% in Table 2.2 for n=8, m=2) to truncated modes from an input viewpoint. This inefficiency can hardly be ignored. One would also want the control design to be efficient in its control power, therefore assessment of spillover effects based on response alone without considering the control powers would be premature.

Fig. 2.4c shows the response of the evaluation model both for natural control with continuously (istributed inputs and LQR control with m=6 point inputs. Both responses are identical. Because there is no mode truncation in

the evaluation model the relative model efficiency is 100%. On the other hand, the global efficiency is about 32% reflecting the fact that the LQR solution with 6 point inputs uses about 3 times more control power than it one were to use a spatially continuous input profile designed for independent control of all 12 modes. In spite of the more control power, the LQR solution cannot produce a resp rse better than that of natural control. In the LQR solution with 6 point inputs for 12 modes intermodal coupling of the controlled responses is inevitable. In this case, it is this coupling of the controlled modal responses that causes excessive use of control effort without producing an improvement in the controlled response over that of natural control. This truly reflects the inefficiency of the control design model.

#### Efficient Model Reduction

An efficient model reduction concept would truncate the modes to which the model or global efficiency is least sensitive. We shall demonstrate the model reduc ion technique based on the model efficiency. A similar procedure can be based on the global efficiency. However, for brevity we do not demonstrate this alternate approach.

We use the model efficiency curves in Fig. 2.2b to find an eighth order reduced model with two and four point inputs. For two inputs (m=2) from Fig. ^.2b, we note that the smallest increments in the model efficiency are caused by Modes 1, 2, 7 and 8. Hence we retain Modes 3-6, 9-12 as the control design model for the given two inputs. Similarly, for the four input configuration (m=4) from Fig. 2.2b, we note that Modes 1, 2, 11 and 12 have the least contributions to the model efficiency. Hence, we truncate these modes and retain Modes 3-10. The efficiencies and control powers of the new eighth order control design models for

two and four inputs are shown in Table 2.3. A comparison of these results to the control design model, which was based on the lowest eight structural modes (Table 2.2), shows that the new control design models have significantly better efficiencies and the effect of model truncation for the new reduced modes are insig-For example, for (m=2) although the natural control power S\* has nificant. increased from 8 to 24 due to the mode selection based on model efficiency, the total actual control power  $S^R$  remained almost the same (158 vs. 162), but the modal control power  $S_{L}^{M}$  rose from 90 to 157 which indicates that the new control design model absorbs almost all of the actual control effort yielding a 97% model efficiency. The response profiles for the new eighth order control design models are shown in Figs. 2.5a and b. Again, in these figures responses of the corresponding natural control, suboptimal control, and evaluation model are superposed. They are hardly different from each other, the responses of the suboptimal control and the evaluation model had almost undetectable overshoots at the peaks in comparison to natural control. Therefore, the curves were not labeled and only the response of the evaluation models are shown in Figs. 2.5a and b. Among all responses natural control always achieved lower amplitudes than the others. Finally, one can now compare the responses of the 8th order reduced order models with m-2, 4 to the response of the 12th order evaluation model in Fig. 2.4c with m=6 inputs.

m	6*8	et	μ	sq*	sqŧ
1	0.647792	1.337612	2.06488	152.3056	98.66239
2	1.271105	1.271092	0.99999	77.67170	98.72891
3	1.8180/1	1.859323	1.02269	53.98065	98.10468
4	2.971592	2.961611	0.99664	32.65535	97.03839
5	3.606529	3.392659	0.99615	26.73133	96.40734
6	4.642842	4.642761	0.99998	20.53855	95.35720

Table 2.1: Efficiencies for ACOSS-4 with n=2

### Table 2.2: Efficiencies for ACOSS-4 with n=8

m	e*\$	eŧ	μ	sq*	sq%
1	2.70735	56.92516	21.0262	15.91033	43.07480
2	5.05710	56.82256	11.2362	8.53799	43.17744
3	19.45213	63.55967	3.2675	1.87333	36.44030
4	29.19059	69.40089	2.3775	1.04825	30.59911
5	39.41595	69.00026	1.7506	0.78648	30,99740
6	47.56904	72.06676	1.5150	0.58721	27.93324

# Table 2.3: Efficiencies and Control costs for ACOSS-4 with an 8th order model obtained via Efficient Model Reduction Approach

Control Design Mode s	m	e*\$	e¥	μ	sq*	sqŧ	S <sup>R</sup>	S℃	SUM	s*
3-6, 9-12	2	14.7	96.55	6.57	0.24	3.45	162.07	156.47	5.6	23.82
3-10	4	44.7	93.93	2.10	0.14	6.07	60.67	56.99	3.68	27.09





ACOSS-4 Tetrahedral Structure





Figure 2.2b: Percent Relative Model efficiencies for ACOSS-4







Figure 2.3.a: Control Powers for n=2





Control Powers for n=8



Figure 2.4a: Responses for n=2 of natural control (continuous input), suboptimal control and evaluation model with m=2, e\*=1.27%, e=1.27%  $\mu=1$ , sq\*=77.67%, sq=98.73%



Figure 2.4b: Responses for n-8 of natural control (continuous input), and suboptimal control and evaluation model with m-2, e\*=5.06%, e=56.8%  $\mu=11.2$ , sq\*=8.3%, sq=43.2%



Figure 2.4c: Responses for n=12 of natural control (continuous input) and suboptimal control with m=6, e\*=31.8%, e=100%  $\mu$ =3.13, sq=sq\*=0

.



Figure 2.5a: Response of evaluation model with efficient model reduction applied for n=8, m=2, e\*=14.7%, e=96.58%,  $\mu$ =6.57, sq\*=0.24, sq=3.45%



Figure 2.5b: Response of evaluation model with efficient model reduction applied for n=8, m=4, e\*=44.7%, e=93.93%,  $\mu$ =2.1, sq\*=0.14, sq=6.07%

#### 3.0 EFFICIENCY MODES ANALYSIS OF STRUCTURE CONTROL SYSTEMS

#### 3.1 Introduction

In this chapter, we delve into the structure of the efficiency quotients to gain further insight to the structure-control problem. The developments of this chapter center around the fact that the efficiency quotients presented in Chapter 2 are the ratios of quadratic functionals of positive definite symmetric control power functions. Therefore, each one of the quantities e, sq, e\* and sq\* represents a Rayleigh's quotient. It is warranted, then, to exploit the features of a Rayleigh's quotient to perform an efficiency modal analysis for the structure-control system and study its implications.

In Section 3.2, we discuss the computation of power matrices for efficiency quotients. Section 3.3 is central to this chapter. The efficiency eigenvalue problem is studied in this section, a number of spectral properties of the control power matrices are discovered, and an efficiency state-space transformation is defined. In Section 3.4, the definitions of "controller modes" and "Principal Controller directions" are given. In addition, an efficiency ellipsoid is iden tified, the surface of which represents all initial disturbances which result in a given efficiency of the system. Section 3.5 introduces the concept of principal efficiency components and discusses its role to quantify the structure of a link between the controller design and initial conditions. Illustrative examples, again using the ACOSS-4 tetrahedral structure studies in Chapter 2, are given in Section 3.6.

#### 3.2 Computation of Control Power Matrices for Efficiency Quotients

Consider the N-th order evaluation model of the structural dynamic system given by Eq. (2.31)

$$Mq + Kq = Q$$
(3.1)

where q and Q are the generalized coordinates and generalized input vectors, respectively. Noting the transformation from a physical control inputs vector F to the generalized inputs Q as

$$Q = DF$$

and the modal properties given by Eqs. (2.32) for the system (3.1), Eq. (3.1) can be replaced by the modal system

$$\xi_{\rm N} + [\omega^2]_{\rm N}\xi_{\rm N} - B_{\rm N}F$$
,  $B_{\rm N} = E_{\rm N}^{\rm T}D$  (3.2)

where  $\xi$  is the N-dimensional evaluation model displacements vector of Eq. (3.1). Typically, we shall assume that Eqs. (3.1) represents a FEM evaluation model of the structural system. Introducing the 2n-dimensional modal states vector for a control design model, 2n < 2N:

$$\mathbf{x} = [\xi_1 \ \xi_1 \ \xi_2 \ \xi_2 \dots \xi_n \ \xi_n]^{\mathrm{T}}$$
(3.3)

we have the modal-state representation of Eqs (3.1) and (3.2) for any smaller order control design model

$$\dot{\mathbf{x}} = \mathbf{A}\mathbf{x} + \mathbf{B}\mathbf{F} \tag{3.4}$$

A = block-diag  $\begin{bmatrix} 0 & 1 \\ -\omega_r^2 & 0 \end{bmatrix}$ , B =  $\begin{bmatrix} 0 & b_1 & 0 & b_2 & 0 \end{bmatrix}^T$  r=1,2...n<N (3.5)

where  $b_r$  is the r-th row vector of  $B_N$ . The control powers  $S^R$ ,  $S_C^M$  and  $S^*$  given by Eqs. ((2.37)-(2.42)), restated here for ready reference,

$$S^{R} = \int F^{T} R^{R} F dt \qquad , R^{R} = D^{T} M^{-1} D \qquad (3.6)$$

$$S_C^M = \int F^T R_C^M F dt$$
 ,  $R_C^M = D^T E_n E_n^T D$  (3.7)

$$S^* = \int F^* R^* F^T dt$$
,  $R^* = D^* M^{-1} D^*$  (3.8)

become, for a stable closed-loop system,

$$S^{R} = x_{0}^{T}P^{R}x_{0}, S_{C}^{M} = x_{0}^{T}P_{C}^{M}x_{0}, S^{*} = x_{0}^{T}P^{*}x_{0}$$
 (3.9)

for which, the state feedback control input is:

$$F = Gx$$
 ,  $F^* = G^*x$  (3.10)

The gain matrix G is the control gain matrix available from any control design method. The gain matrix G\* is the gain matrix of the globally optimal solution given by Eqs. (2.42), (2.19) and (2.20) as discussed in Section 2.6.

The control power matrices  $P^R$  and  $P^M_C$  are obtained by solving the Lyapunov Equations

$$A_{CL}^{T}P^{R} + P^{R}A_{CL} + G^{T}R^{R}G = 0$$
(3.11)

$$A_{cL}^{T}P^{M} + P^{M}A_{cL} + G^{T}R_{c}^{M}G = 0$$
(3.12)

$$A_{CL} - A + BG \tag{3.13}$$

and a closed-form solution for P\* exists, as given in Appendix B.

Noting Eq. (2.38) and the forms ((3.9), (3.11) and (3.12), we also have for the truncated modes

$$S_{U}^{M} = x_{0}^{T}(P^{R} - P_{C}^{M}) x_{0} = x_{0}^{T}P_{U}^{M}x_{0}$$
 (3.14)

Hence, the control power wasted on the uncontrolled dynamics is readily available following solutions of Eqs. (3.11) and (3.12) which are based on the controlled dynamics alone, therefore, no knowledge of truncated dynamics is required.

It must be noted that the weighting matrices R in Eqs. (3.6)-(3.8) and, hence, the control powers, are uniquely defined for computation of structuralcontrol system efficiency. Having obtained the power matrices, one can now compute e, e\*, sq and sq\* associated with any control design according to the definitions (2.43)-(2.46).

#### 3.3 <u>Efficiency Eigenvalue Problem</u>

From the definitions of e, sq, e\* and sq\*, one notes that they are nondimensional quotients involving positive definite numerator and denominator power matrices and the initial state disturbance  $x_0$ . Therefore, an efficiency quotient is also a Rayleigh's quotient, and an efficiency eigenvalue problem can be defined to analyze the structure-control system.

In the following, we only need to define a numerator and a denominator power matrix without necessarily referring to e, e\* sq\* and sq specifically and use e as a generic symbol to represent any one of them. This should not present any difficulty as the context will make it clear whether we are dealing with a specific or a general quotient. Hence, consider the general form of an efficiency quotient

$$e = \frac{x_0^T P_N x_0}{x_0^T P_D x_0} \qquad P_N, P_D > 0 \qquad (3.15)$$

where subscripts N and D denote numerator and denominator power matrices. For a particular structure control system design both  $P_N$  and  $P_D$  are readily available as given in Section 3.2.

The efficiency of the given structure-control design system is then dependent on the nature of initial state disturbance  $x_0$ . To see how the initial state interacts with the particular system, we address the eigenvalue problem associated with the efficiency quotient (3.15).

Introducing the transformation

$$x = P_{D}^{-\frac{1}{2}}z$$
 (3.16)

an efficiency quotient (3.15) becomes

$$e = \frac{z_0^{T} E z_0}{z_0^{T} z_0} , E = P_0^{-\frac{1}{2}} P_N P_0^{-\frac{1}{2}}$$
(3.17)

where E is the nondimensional symmetric positive definite efficiency matrix.

Next, we consider the standard eigenvalue problem associated with the matrix  $\boldsymbol{E}$ 

$$Ev_r = \lambda_r v_r$$
  $r=1, 2, ..., 2n$  (3.18)

where  $v_r$  is a real eigenvector with positive real eigenvalue  $\lambda_r$ , and form the modal matrix V of E

$$V = [v_1 \ v_2 \dots v_{2n}] \tag{3.19}$$

The modal matrix V is normalized to satisfy

$$V^{T}V = I$$
 ,  $V^{T}EV = \Lambda$  (3.20)

where

$$\Lambda = \operatorname{diag} \left[\lambda_1 \ \lambda_2 \dots \lambda_{2n}\right] \tag{3.21}$$

and I is a 2nx2n identity matrix.

Introducing a second transformation

$$z = V\epsilon \tag{3.22}$$

substitute it into Eq. (3.17) and use the normality conditions (3.20) to obtain the quotient e in the form

$$e = \frac{\epsilon_0^{T} \Lambda \epsilon_0}{\epsilon_0^{T} \epsilon_0}$$
(3.23)

We shall refer to the new states  $\epsilon$  in Eq. (3.23) as the <u>efficiency states</u>. Combining Eqs (3.16) and (3.22) we have the transformation

$$x = T\epsilon$$
,  $T = P_{D}^{-\frac{1}{2}}V$  (3.24)

In terms of the efficiency states, the quoticitie (3.23) can be expanded in the form

$$e - \sum_{i=1}^{2n} c_i^2 \lambda_i , c_i^2 = \frac{\epsilon_{0i}^2}{\epsilon_0^T \epsilon_0} , 0 \le c_i \le 1$$
(3.25)

in which we shall refer to  $c_i^2$  and  $\lambda_i$  as the i-th efficiency coefficient and the i-th characteristic efficiency, respectively.

Next, introducing the inverse transformation of (3.24)

$$\epsilon = Lx \qquad L = T^{-1} \tag{3.26}$$

into Eq. (3.23) and considering the original form of efficiency quotient in terms of modal states x, Eq. (3.15), we obtain the following relationships

$$P_{N} = L^{T} \Lambda L \qquad , P_{D} = L^{T} L \qquad (3.27)$$

The modal-state dynamics, Eq. (3.4) can also be represented in terms of the efficiency states by using the transformation (3.24)

$$\epsilon = A^{e}\epsilon + B^{e}F \tag{3.28}$$

$$A^{e} = LAT , B^{e} = LB$$
 (3.29)

The feedback control law for F in Eq. (3.10) becomes

$$F = G^e \epsilon$$
 ,  $G^e = GT$  (3.30)

In terms of  $\epsilon$ -states the closed-loop system becomes

$$\dot{\epsilon} = A_{CL}^{e}\epsilon$$
,  $A_{CL}^{e} = A^{e} + B^{e}G^{e} = LA_{CL}T$  (3.31)

where  $A_{CL}$  is given by Eq. (3.13).

By using Eqs. (3.27) and (3.31) in Eqs. (3.11) and (3.12), the Lyapunov Equations associated with the  $\epsilon$ -state space representation are obtained as:

$$A_{CL}^{eT} \Lambda + \Lambda A_{CL}^{e} + G^{eT} R_{C}^{M} G^{e} = 0$$
(3.32)

$$A_{CL}^{eT} I + \Lambda I_{CL}^{e} + G^{eT} R^{R} G^{e} = 0$$
(3.33)

Hence,  $\Lambda$  is recognized as the numerator power matrix and the identity matrix I is the denominator power matrix associated with the definition of an efficiency quotient in the  $\epsilon$ -state space. That is

$$\mathbf{e} = \frac{\mathbf{x}_{0}^{\mathrm{T}} \mathbf{P}_{\mathrm{N}} \mathbf{x}_{0}}{\mathbf{x}_{0}^{\mathrm{T}} \mathbf{P}_{\mathrm{D}} \mathbf{x}_{0}} = \frac{\epsilon_{0}^{\mathrm{T}} \mathbf{T}^{\mathrm{T}} \mathbf{P}_{\mathrm{N}} \mathbf{T} \epsilon_{0}}{\epsilon_{0}^{\mathrm{T}} \mathbf{T}^{\mathrm{T}} \mathbf{P}_{\mathrm{D}} \mathbf{T} \epsilon_{0}} = \frac{\epsilon_{0}^{\mathrm{T}} \Lambda \epsilon_{0}}{\epsilon_{0}^{\mathrm{T}} \epsilon_{0}}$$
(3.34)

From Eq. (3.34) or inverting Eq. (3.27) we obtain the orthonormality relationships

$$T^{T}P_{N}T = \Lambda , T^{T}P_{D}T = I$$
 (3.35)

Thus, the transformation T diagonalizes both the numerator and denominator power matrices simultaneously and matrix T is orthonormal with respect to the real

power control matrix  $P_D$ . This completes the eigenvalue problem of the efficiency quotient (3.15).

#### 3.4 <u>Efficiency Modes and Principal Controller Directions of a Structure-</u> <u>Control System</u>

In Section (3.2) we showed that the transformation matrix T defined by Eq. (3.24) can be used to transform from the modal state space (x-space) to a new efficiency state-space ( $\epsilon$ -space) to describe the system dynamics by Eq. (3.28). Furthermore, just as the structural modal matrix  $E_N$  diagonalizes the structural mass and stiffness matrices according to Eq. (2.32), the matrix T diagonalizes similarly both of the control power matrices  $P_N$  and  $P_D$  according to Eq. (3.35). By analogy, we shall refer to the T matrix as the efficiency modal matrix, and to its columns  $t_r$  as the efficiency modes of the control system.

$$T = [t_1 \ t_2 \dots t_{2n}] \tag{3.35}$$

Next, recognizing the efficiency coordinates  $c_i$  in Eq. (3.25) as the normalized efficiency coordinates associated with an initial disturbance  $\epsilon_0$  on the efficiency state

$$c_{i} = \epsilon_{i0} / |\epsilon_{0}| \tag{3.36}$$

and introducing

$$\epsilon_i = L_i^T x \tag{3.37}$$

where  $L_i$  is the i-th row vector of  $L=T^{-1}$ , the efficiency Equation (3.25) can be written in the form

Equation (3.38) represents the equation of an ellipsoid in the 2n-dimensional space with principal axes of length  $\sqrt{e/\lambda_i}$  (i=1,...2n) and  $c_i$  is the coor-

dinate along the i-th principal axis of all initial disturbances which yield a specific efficiency e. Furthermore, the direction of the i-th principal axis is given by the i-th eigenvector  $V_i$  of the efficiency matrix E where  $V_i$ 's are orthonormal in the usual sense as given by (3.20) associated with the eigenvalue problem of Eq. (3.18). With respect to the efficiency eigenvalue problem associated with the original modal coordinates x, Eq. (3.15), the i-th principal axis of the ellipsoid (3.38) is given by the i-th column  $t_i$  of the transformation matrix T, Eq. (3.24) which is the modal matrix of quotient (3.15). We note that the directions  $t_i$  are orthonormal not in the usual sense but with respect to the denominator control power matrix  $P_D$ , given in Eq. (3.35).

Since  $0 < e \le 1$  (excluding the case when e=sq\*), different initial conditions  $x_o$  yielding different efficiencies simply will result in rescaling of the lengths of the principal axes of the ellipsoid, largest length in the i-th direction given by  $\sqrt{1/\lambda_i}$ . The largest and the smallest possible length of the axes are then  $\sqrt{1/\lambda_{min}}$  and  $\sqrt{1/\lambda_{max}}$ , respectively.

We shall refer to the ellipsoid (3.38) as the <u>efficiency ellipsoid</u>, alternately, recognize the <u>efficiency modal vectors</u>  $t_i$  as the <u>principal controller</u> <u>directions</u>. We note that the efficiency modes, characteristic efficiencies and the shape of the <u>efficiency</u> ellipsoid *e*: completely and only defined by the particular control system design and the structural system design embedded in the matrices G, A and B in Eqs. (3.4) and (3.10). Initial disturbances  $x_0$  then determine the size of the efficiency ellipsoid acting merely as a scaling factor to yield a specific efficiency for the structure-control system.

Because a quotient e in the form of Eq. (3.15) represents a Rayleigh's quotient and since  $0 < e \le 1$ , we observe the following properties for efficiencies:

- a) The characteristic efficiencies  $\lambda_i$  are bounded by  $0 \le \lambda_i \le 1$  for e, e\* and sq, for the quotient sq\*  $\lambda_i \ge 0$ .
- b) Efficiencies have stationery values at  $\lambda_i$  for initial disturbances  $x_0 = t_r 1, 2, ..., 2n$ . Specifically, the minimum efficiency a structure-control system can achieve is  $e_{\min} = \lambda_{\min}$ , and will occur if  $x_0 = t_{\min}$ , that is if the initial disturbance coincides with the eigenvector (controller efficiency mode) associated with  $\lambda_{\min}$ , that is if the  $x_0$  is completely aligned with the direction of the principal axis along  $t_{\min}$ . We shall refer to  $\lambda_{\min}$  as the fundamental efficiency. Similarly, the upper bound of the efficiency of the system is given by  $e_{\max} = \lambda_{\max}$ , corresponding to  $x_0 = t_{\max}$ .

In particular, if the quotient (3.15) is the relative model spillover quotient

$$sq = 100 * \frac{S_U^M}{S^R} = \frac{x_0^T P_U^M x_0}{x_0^T P_R^R x_0}$$
 (3.39)

it then represents the percent inefficiency (fraction of control power wasted on residual dynamics) of the control system. Noting that

$$P^{R} = P^{M}_{C} + P^{M}_{U} \tag{3.40}$$

and denoting the eigenvalues of the relative efficiency quotient

$$e = \frac{x_0^T P_C^M x_0}{x_0^T P_X^R x_0}$$
(3.41)

by  $\lambda^{e}$  and the eigenvalues of the spillover quotient (3.39) by  $\lambda^{s}$  we write the eigenvalue problem

$$[\lambda^{e}P^{R} - P_{C}^{M}]t_{r} = 0$$
  $r=1, 2, ..., 2n$  (3.42)

substituting (3.40) into (3.42) we obtain

$$[(1 - \lambda_r^{e})P^{R} - P_{U}^{M}]t_r = 0$$
(3.43)

which constitutes the eigenvalue problem for the relative model spillover quotient sq. Hence, the eigenvalues of the spillover quotient sq (or equivalently phrased, controller power inefficiency) are

$$\lambda_{r}^{s} = 1 - \lambda_{r}^{e}$$
  $r=1, 2, ..., 2n$  (3.44)

with the same eigenvectors  $t_r$  as that of the relative model efficiency e in Eq. (3.42). Similar to Eq. (3.25), for the model spillover quotient we can write the expansion

which yields, after recognizing  $\sum_{i=1}^{2n} c_i^2 = 1$ 

$$sq = 1 - e$$
 (3.46)

as was given in Eq. (2.46).

Note that although there may be infinitely many truncated modes in the system, what happens from the point of view of control power used is described completely in the 2n-dimensional space spanned by the controller efficiency modes  $t_r r=1,2,\ldots,2n$ , for the infinite dimensional system.

#### 3.5 <u>Principal Efficiency Components - A Link Between Controller Design and</u> <u>Initial Conditions</u>

The ultimate efficiency of a control system is dictated together with the structure-control design embedded in the matrices A, B and G and the initial disturbance  $x_0$ . To see how these two factors interact, consider the expansion of an efficiency quotient given by (3.25) and write

$$2n$$

$$e = \Sigma e_i , e_i = c_i^2 \lambda_i$$

$$i=1$$
(3.47)

where  $e_i$  represents the efficiency of the controller in the i-th principal controller direction. We shall refer to  $\underline{e_i}$  as the i-th principal efficiency component. Obviously, from Eq. (3.47), the principal efficiency  $e_i$  is the product of the corresponding characteristic efficiency  $\lambda_i$  and the coefficient  $c_i^2$ . For a given structure, the characteristic efficiency  $\lambda_i$  is, however only a function of the control gain matrix G and the controller configuration represented in matrix B via Eqs. (3.15), (3.11), (3.12) and (3.13). On the other hand, the efficiency coordinate  $c_i$ , Eq. (3.38) is a function of both the initial state  $x_0$ and the controller design via L = T<sup>-1</sup> which in turn is a function of only G and B, again by virtue of Eqs. (3.15) and (3.11), (3.12) and (3.13). Expressing the functional dependencies of the quantities involved in the principal efficiency components in (3.47), one can write

$$e_{i} = c_{i}^{2}\lambda_{i} = \frac{x_{0}L_{i}(A, B, G)L_{i}(A, B, G)x_{0}}{L^{T}(A, B, G) L(A, B, G)} \qquad \lambda_{i} (A, B, G) \qquad (3.48)$$

and observe the effect of both structure and control design on both  $c_i$  and  $\lambda_i$ . Common sense requires that an efficient controller must have high principal efficiency components in the principal controller directions, in other words, the products  $c_i^2 \lambda_i$  must be high in each direction. Although a control design might have a large characteristic efficiencies, it will not necessarily have a high efficiency unless the projection(s)  $c_i$  of the initial disturbance  $x_o$  is (are) significant along the directions of large  $\lambda_i$ . Considering that both  $c_i$  and  $\lambda_i$ are functions of control design as shown by Eq. (3.48), it may, therefore, be the control design itself which will either enhance or degrade its efficiency. Hence, it is of significance not only to have controllers with large characteristic efficiencies but also to have their principal  $(\lambda_i)$  directions aligned favorably with the initial disturbances  $x_o$  thus extracting large projections of power so that they will yield high efficiencies in those directions.

Traditionally, in the design of a control system (such as in the LQG-type designs) there is no avenue to bring in the initial disturbance information to the computation of the control gain matrices. After having obtained a so called "optimal" controller via LQG algorithms, one does not really know how the control system will interact with any initial disturbance until a simulation is done. It is possible that an optimal controller (in the sense of the theory used to design it) will be an extremely inefficient controller if it does not "see" or "receive" the initial disturbance properly along its principal controller directions.

On the other hand, the efficiency modes analysis of a structure-control system clearly reveals the internal link by which the controller-design and the initial disturbances  $x_0$  interact. In fact, after identifying the modal matrix T and  $\lambda_1$ , that is the principal directions and the characteristic efficiencies of a controller one can readily for any initial disturbance  $x_0$  form the projections  $c_i$  via Eq. (3.38) to examine how they pair up with the respective characteristic efficiencies  $\lambda_i$  and obtain an apriori (before simulation) information about the controller performance. Certainly, the observation of the link between the controller design and the initial disturbance is a constructive one so that given an initial disturbance and a structure, the objective would be to design a control gain which will pair up significant  $\lambda_i$  with significant projections  $c_i$ .

#### 3.6 Illustrative Examples on ACOSS-4

The structure is the ACOSS-4 structure considered in Chapter 2, Fig. 2.1. In this case, the structure is subjected to an initial disturbance of unit displacement in the x direction at Node 2. Input configurations are selected from a set of twelve available actuators located at the pods of the structure. In all cases, the controls were designed by using LQR theory with unit control weighting (r=1) and state weighting  $q=\omega_r^2$  for the modal displacements and q=1 for the modal velocities. The evaluation model has N=12 modes yielding a 24th order state-space evaluation model.

#### Efficiency Modes Analysis

- Example 3.1: The control design model has two structural modes, Modes 11 and 12, n=2(Modes 11,12) one input was used m=1 (#4) which was actuator number 4. Figures labeled SIM39 (Figures 3.1 through 3.3) give the line of sight error simulation results for this case.
- Example 3.2 n=2 {Modes:11,12}, m=2(#3,#4), that is two actuators #3 and #4 were used for control.

The nomenclature for the computer outputs of the examples presented in this section and the computer outputs are given in Appendix C. Simulations of the globally optimal distributed input system, labeled "Globally Optimal System", "Suboptimal system" with m-point inputs and the "evaluation model" response with m-point inputs are presented for the examples listed below. A designation SIM (number) associated with an example indicate that figures labeled with SIM 9 Number show the line of sight error for that example. Line of sight error is computed at the vertex (Node 1) of the structure, according to the expression  $LOS=U_{x1}^2+U_{y1}^2$ , where  $U_{x1}$ ,  $U_{u1}$  are nodal displacements in the x and y directions at the vertex.

- Example 3.3: n=2 (Modes:11,12), m=4(#1-4), SIM41 (Figures 3.4 and 3.6).
- Example 3.4: n=8 (Modes:1-8 natural order), m=2 (#1,2), SIM1 (Figures 3.7, 3.8 and 3.9).
- Example 3.5: n=8 (Modes:1-2 natural order), m=4 (#1-4), SIM10 (Figures 3.10, 3.11 and 3.12).
- Example 3.6 n=2 {Modes:1-8 natural order}, m=2 (#1,2). this case presents the results of the (in) efficiency modal analysis of the quotients with  $P_N=P_U^M$ , SIM1 (Figures 3.7, 3.8, and 3.9).

#### Effect of Initial Disturbances on Efficiencies

Example 3.7 Consider the characteristic efficiencies for the global efficiency e\* and the relative model efficiency e of Example 3.4 (n=8(1-8), m=2) listed in Appendix C. For e\*, we observe that with a mere change of  $x_o$ , this system can have a maximum global efficiency of  $\lambda_{max}^*$ -53.76%, corresponding to the 2nd characteristic efficiency listed. This would occur when  $x_o=t_2$ , that is when the initial state is the second column of the efficiency modal matrix T listed. The worst efficiency of  $\lambda_{min}^*=\lambda_{14}^*=0.773$ % as listed in the computer outputs in Appendix C. For the model efficiency we note that all characteristic efficiencies are 59.79% $\geq\lambda\geq$ 54.04%. Hence, again the corresponding initial disturbances which will culminate in respective efficiencies can be identified from the columns of the modal matrix T listed in Appendix C. It is noted that regardless of  $x_o$ , the model efficiency can not change more that 5.7% within that bracket. Therefore, in this case the model efficiency is very robust to changes in  $x_o$ .



Figure 3.1: Globally optimal system LOS for n=2 (modes: 11,12); example 3.1



Figure 3.2: Suboptimal system LOS for n=2 (modes: 11,12), m=1 (input #4); example 3.1



Line of Sight Error

Figure 3.3:

Evaluation model LOS for n=2 (modes: 11,12) m=1 (input #4); example 3.1


Line of Sight Error

Figure 3.4: Globally optimal system LOS for n=2 (modes: 11,12); example 3.3



Line of Sight Error

Figure 3.5: Suboptimal system LOS for n-2 (modes: 11,12); m-4 (#1-4); example 3.3



Line of Sight Error

Figure 3.6: Evaluation model LOS for n-2 (modes: 11,12), m-4 (#1-4); example 3.3



Figure 3.7: Globally optimal system LOS for n=8 (modes: 1-8) m=2 (#1-2); example 3.4



Figure 3.8: Suboptimal system LOS for n=8 (modes: 1-8), m=2 (#1-2); example 3.4



Figure 3.9: Evaluation model LOS for n=8 (modes 1-8), m=2 (#1-2); example 3.4



Figure 3.10: Globally optimal system LOS for n=8 (modes: 1-8), m=4 (#1-4); example 3.5



Figure 3.11: Suboptimal system LOS for n=8 (modes: 1-8), m=4 (#1-4); example 3.5



Figure 3.12: Evaluation model LOS for n=8 (modes: 1-8), m=4 (#1-4); example 3.5

#### 4.0 EFFICIENCY MODEL/CONTROLLER REDUCTION

# 4.1 Introduction

In the efficiency analysis of a 2n-th order structure-control system design model, we identified 2n-controller efficiency modes which led to the efficiency state ( $\epsilon$ -state) description of the original design model in the structural modal space (x-state). The two modal descriptions are fundamentally distinct in their nature. As one is all too familiar with structural modes of the system characterized by the x-states, the controller efficiency modes characterized by the  $\epsilon$ states constitute a new concept where the latter describe principal controller directions signifying avenues of efficient use of available control power. From a structure-control system perspective, how these two sets of modes relate to each other becomes a measure of efficiency of the control task. A certain degree of misalignment of these controller modal directions and the structural modal directions result in controller's not "seeing" the structure directions retained in the model properly, thus not channeling the control power to the structure and wasting it to the unrepresented structure truncated dynamics. Conversely, certain structural directions may be "seen strongly or in full view" by the principal controller directions leading to efficient channeling of control power to the structure in that direction.

For analysis and design purposes via the efficiency concept, the above observations need to be expressed in meaningful mathematical terms. To this end, one needs to correlate the structural modal states x and the efficiency modal states  $\epsilon$ . Quantifying such correlations can help the designer to identify structural modal states with weak or strong contributions to a set of critical principal controller directions. Once this is accomplished, one can then proceed to

delete structural modal states and/or controller efficiency modal states which are weakly correlated.

The interrelationship between the structural modes and controller modes can be structured and quantified by <u>decomposition of efficiency</u> quotients by defining decomposition matrices for the numerator and denominator control power matrices. These decompositions, in turn, pave the way to proposing relevant model controller reduction criteria based on the efficiency concept.

In Section 4.2 we introduce decomposition matrices for the numerator and denominator control power matrices by using the spectral properties of the matrices given in Chapter 3. Next <u>component efficiencies</u> associated with structural wodal states are defined for the principal <u>efficiency components</u>. Certain features of these components efficiencies are discussed requiring the need to define coupled and decoupled component efficiencies. In Section 4.3, the first controller reduction method in the efficiency state space is presented. Section 4.3 leads to the subject of Section 4.4, efficiency filtering. Section 4.5 presents the methodology for efficient modal controller reduction which relies on the decomposition discussed in Section 4.2. Also, in Section 4.5 several important observations are made to be able to use the efficiency concepts properly in the analysis and design studies. Finally, in Section 4.6, we give illustrative examples using the ACOSS-4 structure.

# 4.2 <u>Decomposition of Efficiency Quotients</u>

ĺ

Associated with any of the quantities e, e\*, sq and sq\*, we define 2ndimensional numerator and denominator matrices [N] and [D] with elements

$$\{N\} = N_{ij} = [X_0 L_i L_i^T \lambda_i]_{jj} \qquad i, j=1, 2, \dots, 2n$$
(4.1)

$$D] = D_{1J} = \{X_0 L_1 L_1^T\}_{JJ}$$
(4.2)

where  $X_{0}\ is$  the initial disturbance covariance matrix

$$X_{\rm C} = X_{\rm C} X_{\rm C}^{\rm T} \tag{4.3}$$

Next by forming the total sums N and D, one can verify that

$$N = \sum_{i} \sum_{j} N_{ij} = x_{0}^{T} P_{N} x_{0}$$
(4.5)

$$\begin{array}{cccc} 2n & 2n \\ D = \Sigma & \Sigma & D_{ij} = x_0^T P_D x_0 \\ i & j \end{array}$$
(4.6)

$$e = N/D$$
  $e = \{e, e^*, sq, sq^*\}$  (4.7)

Each element  $N_{ij}$  and  $D_{ij}$  represents the power contribution of j-th structural modal state  $x_j$  in the i-th principal controller direction  $\epsilon_i$  for the respective numerator and denominator terms. We shall refer to matrices [N] and [D] as the numerator and denominator control power decomposition matrices, respectively. The quotient e in terms of sums of their elements represents an efficiency decomposition. Although through the controller design one has the power matrices  $P_N$  and  $P_D$  available in the beginning, their representations in terms of the elements  $N_{ij}$  and  $D_{ij}$  unveils the internal structure of the power associated with the controller. This exposition of the internal structure is, however, possible only through the efficiency modal analysis. Specifically through this modal analysis, each element  $N_{ij}$  and  $D_{ij}$  quantifies the correlation between the j-th structural mode  $x_i$  and the i-th controller mode.

Next, we define the partial sums

as the power contributions (extraction) of j-th structural modal state  $x_j$  to (from) all principal controller directions. Similarly, we define power contributions (extraction) of all structural modal states to (from) the i-th principal controller direction:

Considering definitions (4.9) and the definition of principal efficiency component  $e_i$ , Eq. (3.47), it follows that

$$e_{i} = \frac{N_{i}}{\sum D_{i}} = \frac{N_{i}}{D} = c_{i}^{2}\lambda_{i}$$
  $i=1,2,...,2n$  (4.10)

Where we recall that  $e_i$  represents a component with respect to the i-th principal controller direction. Noting Eq. (3.25), one can also regard  $e_i$  as the contribution to efficiency of the i-th efficiency state  $\epsilon_i$ . Similarly, one can identify efficiency components with respect to the structural modal directions and define

$$e_{j} = \frac{N_{j}}{D} = \frac{N_{j}}{\sum D_{j}}$$
,  $e = \sum e_{j}$   $j=1,2,...,2n$  (4.11)  
j

We shall refer to  $e_j$  as the j-th (structural) component efficiency and emphasize the difference in the terminology that is being used here that  $e_j$  is the <u>com-</u> <u>ponent efficiency</u> and  $e_i$  is the principal <u>efficiency component</u>, respectively, the former is with respect to structural modal components  $x_j$  and the latter with respect to controller efficiency modal components  $\epsilon_i$ .

Strictly speaking, although definitions of components  $e_i$  and  $e_j$  as in Eqs. (4.10), (4.11) are mathematically correct since their sums over respective

indices yield the total efficiency e, from a physical perspective certain features of these definitions must be considered cautiously. In both definitions, the denominator is common for all components  $e_i$  and  $e_j$ . So in effect, the significance of a component is solely determined by the numerator power contributions  $N_i$  and  $N_j$ . For that matter, the use of D which is the real control power  $S^R$  as a common denominator is of no consequence, as the relative contributions of  $N_j$  (or  $N_i$ ) terms to total efficiency would not change if we were to use any other quantity as a common denominator.

Indeed, in this case the function of efficiency as a relevant nondimensional quantity is subordinated to the contributions to the dimensional numerator control powers. Therefore, from the efficiency point  $e_j$  and  $e_i$  component descriptions may not be proper characterizations. In case of component efficiencies  $e_j$  for the structural modal states  $x_j$ , the characterization is completely out of contact with the concept and purpose of efficiency quotient as we show next.

Considering the definition of  $N_j$ , Eqs. (4.8) and (4.1) and the spectral decomposition of control powers matrices in (3.27), we can write

$$N_{j} = [X_{0} \sum_{i=1}^{N} L_{i} \lambda_{i} L_{i}^{T}]_{jj} = [X_{0} P_{N}]_{jj} = e_{j} D \qquad j=1,2,\ldots,2n \qquad (4.12)$$

summing over all component efficiencies:

$$eD = \sum_{j}^{2n} N_{j} = Trace X_{0}P_{N} = x_{0}^{T}P_{N}x_{0} = S_{C}^{M}$$
(4.13)

Thus, component efficiency characterization  $e_j$  in this sense is no different that component cost analysis of a modal power  $S_C^M$  of the control design model as defined in Ref. (7) and all the benefit and the physical reason in introducing a denominator control power matrix to yield the nondimensional efficiency is lost. Furthermore, the components  $N_j$  given by Eq. (4.12) are computed exactly as given in Ref. (7) associated with which there is certainly no concept of efficiency.

A similar situation exists for the efficiency components  $e_i$ , however, at a conceptually less critical level. Considering the definition of N<sub>i</sub>, Eqs. (4.9), (4.2) and (3.27), we can write

$$\sum_{i}^{2n} N_{i} = \sum (X_{0}L_{i}\lambda_{i}L_{i}^{T})_{jj} \qquad i=1,2,\ldots,2n \qquad (4.14)$$

$$j=1$$

which upon recognizing  $x_0^T L_i = \epsilon_{io}$  and  $\sum (x_0 L_i^T)_{jj} = \epsilon_{io}$  yields

$$N_i = \lambda_i \epsilon_{i0}^2 = e_i D.$$

Summing over i

$$\sum N_{i} = \sum \lambda_{i} \epsilon_{io}^{2} = eD = S_{C}^{M} = \epsilon_{O}^{T} \Lambda \epsilon_{O}$$
(4.15)  
i i

This is again a decomposition of the numerator modal power of the control design model regardless of the function of denominator control power D as the nondimensionalization factor. The form of  $S_C^M$  as in Eq. (4.15) is similar to the cost decoupling concept discussed in Refs. (7 and 12). Here, the efficiency states  $\epsilon$  play a role similar to the cost decoupling coordinates. Hence, principal efficiency components  $e_i$  characterization of e is also a characterization of the numerator control power  $S_C^M$  alone. However, because this characterization can be affected cally in terms of efficiency coordinates  $\epsilon_i$  after solving for the efficiency modal matrix T, the function of denominator concrol power in the concept of efficiency is accounted for indirectly. In contrast, in the  $e_j$  characterization there was no need for the efficiency concept. It must be noted, however, that the identification of the cost decoupling coordinates as in Ref. (7 and 12) is only a recognition of the algebraic similarity. The mathematical and the physical foundations under which the decoupling features are brought about in this investigation and in Ref. (7 and 12) are fundamentally different.

From the above discussion, to be consistent with the definitions of efficiency there evolves the need to define component efficiencies  $e_j$  for the contributions of structural modal states  $x_j$  which takes into account the denominator contributions  $D_j$  as well as the numerator contributions  $N_j$ . To this end, we define two component efficiencies. The first one is:

$$\mathbf{e}_{j} = \frac{\mathbf{N}_{j}\mathbf{D} + \mathbf{D}_{j}\mathbf{N}}{2\mathbf{D}^{2}} \qquad \mathbf{e} = \sum_{j} \mathbf{e}_{j} \qquad (4.16)$$

which we shall refer to as the coupled component efficiency. The coupled designation is used to note that in the definitions of  $N_j$  and  $D_j$ , terms involving products of  $x_{jo}$  with all other states appear. Hence, a state  $x_{jo}$  may have a high component efficiency contribution not because of itself alone but also because of the contribution of its coupling terms with another state or states. The second definition of component efficiency disregards all coupling terms among the states, and we refer to it as the decoupled component efficiency:

$$e_{j} = n_{j}/d_{j}$$

$$n_{j} = \bar{x}_{jo}^{T} P_{Njj} \bar{x}_{jo}$$

$$d_{j} = \bar{x}_{jo}^{T} P_{Djj} \bar{x}_{jo}$$

$$\bar{x}_{j}^{T} = [\xi_{j} \ \dot{\xi}_{j}] \qquad j=1,...,n$$

$$(4.17)$$

where  $P_{Njj}$  and  $P_{Djj}$  are the 2x2 block diagonal partitions of the numerator and denominator power matrices, respectively. In the definition of component efficiency, we retained the pair of a modal displacement and its velocity as a component for obvious reasons. This is also to be observed in all component efficiencies. The decoupled component efficiency  $e_j$  would be exactly the efficiency of the system if all initial states but the modal states j (displacement and velocity) were zero, hence, they indeed represent proper component contributions to the overall efficiency. We also recognize that  $e_j < 1$ .

Next, noting that

$$N = \sum n_{j} + \sum \sum \bar{x}_{io}^{T} P_{Nij} \bar{x}_{oj} \qquad i, j=1,2,...,n \qquad (4.18)$$

$$j \qquad i \neq j$$

and

$$D = \Sigma d_j + \Sigma \Sigma \bar{x}_{io} P_{Dij} \bar{x}_{oj}$$
$$j \qquad i \neq j$$

where the second summations are the off block-diagonal power cross-coupling terms. In general, these off-diagonal contributions are sign variant, and therefore, one might expect them to vanish in a statistical sense if sufficient modal states are disturbed initially. Thus, we can write an expected total efficiency in terms of the block-diagonal decoupled powers as

$$E(e) \approx \frac{\prod_{j=1}^{n} \Sigma n_{j}}{\prod_{j=1}^{n}}$$
(4.19)

where E denotes expectation. Further, we caution that

Many studies that have been conducted during the course of this investigation verify that the expectation of overall efficiency in terms of decoupled component contributions is quite relevant and off-block diagonal contributions to both the numerator and control powers are insignificant.

In the definitions of both the coupled and decoupled component efficiencies  $(e_j)$  one notes that the numerator and denominator component contributions  $N_j$ ,  $D_j$ ,  $n_j$  an  $d_j$  can be computed exactly directly from the numerator and denominator power matrices  $P_N$  and  $P_D$  without invoking any of the results of the efficiency modes analysis. In these definitions of the exact component contributions via the power matrices summation over all principal controller directions i=1,2,...,2n is implicit. However, in reality not all principal controller direction contributions need not be taken into account, therefore, it suffices in general to consider only the partial sums involving dominant controller modes in Eqs. (4.1), (4.2) and (3.27)

$$N_{j} \approx [X_{0}\sum_{i=1}^{n_{p}} P_{Nj}]_{jj}$$
,  $D_{j} \approx [X_{0}\sum_{i=1}^{n_{p}} P_{Di}]_{jj}$ ,  $n_{p} < 2n$   $j=1,...,2n(4.20)$ 

where

$$P_{Ni} = \lambda_i L_i L_i^T , P_{Di} = L_i L_i^T$$
(4.21)

are controller power spectral components of the power matrices available only after an efficiency modal analysis. Similarly, for approximations to  $n_j$  and  $d_j$ :

$$P_{N_{jj}} \simeq \sum_{i}^{n_{p}} [P_{N_{i}}]_{jj}$$
 and  $P_{D_{jj}} = \sum_{i}^{n_{p}} [P_{D_{i}}]_{jj}$ ,  $j=1,2,...,2n$  (4.22)

We are now in a position to propose model/controller reduction criteria based on the concept of efficiency. To this end, we shall utilize the component efficiencies  $e_j$  and principal efficiency components  $e_i$  discussed in this section.

# 4.3 <u>Controller Reduction in the Efficiency State Space - Efficiency Components</u> <u>Truncation</u>

We use the transformation (3.26) from the modal space to the efficiency space to represent the closed-loop system in the efficiency space as given by Eqs. (3.28)-(3.31).

A controller reduction in the  $\epsilon$ -space can be affected by retaining only  $n_p$  components of  $\epsilon$  associated with the dominant controller modes with highest  $n_p$  principal efficiency components  $e_i$ , Eqs. (3.47) that is  $\{e_i\}_1 = \{e_1 > e_2 > \ldots e_{np}\}$ ,  $n_p < 2n$ . Rearranging the controller modal matrix T in accordance with the set  $\{e_i\}_1$  we write:

$$\mathbf{x} = \mathbf{T}_1 \boldsymbol{\epsilon}_1 + \mathbf{T}_2 \boldsymbol{\epsilon}_2 \tag{4.23}$$

where  $T_1$  are the controller modes corresponding to the characteristic efficiencies  $\lambda_i$  associated with the set  $\{e_i\}_1$  and subscript 2 denotes the truncated states. The decreasing order of principal efficiency components  $\{e_i\}$  does not necessarily correspond to highest  $n_p$  characteristic efficiencies  $\lambda_i$ . Similarly, the partitioned form of the closed-loop system in the efficiency space is:

where the matrices  $A^{\epsilon}$ ,  $B^{\epsilon}$  and  $G^{\epsilon}$  are given by Eqs. (3.28)-(3.31). A reduced order controlled system is obtained from the above by considering only the dynamics associated with  $\epsilon_1$ 

$$\stackrel{\bullet}{\overline{\epsilon}}_{1} = [A_{11}^{\bullet} + B_{1}^{\bullet}G_{1}^{\bullet}] \ \overline{\epsilon}_{1} = A_{CL11}^{\bullet}\overline{\epsilon}_{1} \qquad \overline{\epsilon}_{10} = \epsilon_{10}$$

$$(4.25)$$

for which the feedback control law is obtained by rendering  $G_2^e = 0$  in the full order control law Eq. (3.30).

$$\mathbf{F} = \mathbf{G}_1^{\mathbf{e}} \boldsymbol{\epsilon}_1 \tag{4.26}$$

However, considering the full-order 2n-th order design model Eq. (4.24) with  $G_2^{e}=0$ , the actual system dynamics with the reduced-order controller would be,

Thus, in the actual system, the reduced-order states  $\epsilon_1$  and the truncated states  $\epsilon_2$  retain their dynamic coupling. Because the simulations of (4.25) and (4.27) will yield different responses for  $\epsilon_1$ , an overbar is used for  $\epsilon_1$  in (4.25) in ignoring the dynamic coupling of  $\epsilon_1$  with the truncated efficiency states  $\epsilon_2$ . The implicit assumption in this approach is that the system (4.27) remain stable with the reduced-order controller. This assumption seems reasonable since the reduction criterion is based on retaining the states associated with the most efficient controller directions, thus, the anticipation is that significant control power still remains with the reduced-order controller. Put in other words, the truncated controller directions associated with the  $\epsilon_2$  states with low principal efficiency components e<sub>i</sub> will have high inefficiencies, thus, their elimination will save control authority. On the other hand, the stability of the reduced model Eq. (4.25) is guaranteed as we will note shortly. In addition, the efficiency of the reduced order model (4.25) with the reduced-order controller (4.26)would be of interest. Assuming that the  $A^{e}_{CL11}$  is stable, the Lyapunov Equations associated with the modal power of the design model  $S_C^M$  and the real power  $S^R$ become

$$A_{CL11}^{eT} P_{e}^{M} + P_{e}^{M} A_{CL11}^{e} + G_{1}^{eT} R_{C}^{M} G_{1}^{e} = 0$$

$$(4.28)$$

$$A_{CL11}^{eT} P_{e}^{R} + P_{e}^{R} A_{CL11}^{e} + G_{1}^{eT} R^{R} G_{1}^{e} = 0$$
(4.29)

Next, considering the  $n_Pxn$  upper left hand partitions of Equations (3.32) and (3.33), we obtain

$$A_{CL11}^{eT} \Lambda_{11} + \Lambda_{11} A_{CL11}^{e} + G_1^{eT} R_C^{m} G_1^{e} = 0$$
(4.30)

$$A_{CL11}^{eT} I_{11} + I_{11} A_{CL11}^{e} + G_1^{eT} R^R G_1^{eT} = 0$$
(4.31)

Comparing Eqs. (4.28)-(4.31) we conclude:

$$P_{e}^{M} = \Lambda_{11}$$
 ,  $P_{e}^{R} = I_{11}$  (4.32)

where

$$\Lambda_{11} = \operatorname{diag}[\lambda_1^{\bullet} \ \lambda_2^{\bullet} \dots \lambda_n^{\bullet}] \tag{4.33}$$

and  $I_{11}$  is the  $n_p$ -dimensional identity matrix. Conversely, associated with the closed-loop matrix  $A_{CL11}^{e}$ , since Eqs. (4.30), (4.31) guarantee  $\Lambda_{11}$  and  $I_{11}$  as positive-definite solutions, it follows that  $A_{CL11}^{e}$  is a stable matrix. Thus, we have shown that the reduced-order modal matrix  $T_1$  still simultaneously diagonalizes the modal and real power matrices  $S_C^{M}$  and  $S^{R}$  of the reduced-order model (4.25) without perturbing the original characteristic efficiencies associated with the  $\epsilon_1$  states. The  $n_p$ -th order efficiency of the reduced system becomes

$$\mathbf{e}_{nP} = \frac{\epsilon_{10}^{T} \Lambda_{11} \epsilon_{10}}{\epsilon_{10}^{T} \epsilon_{10}} = \sum_{i=1,2,}^{n_{P}} \epsilon_{i}^{2} \lambda_{i}^{e}$$
(4.33)

The change in the efficiency of (4.25) in comparison to the original system (4.24) is

$$\Delta e = e \cdot e_{nP} = \sum c_i^2 \lambda_i^e - \sum e_i$$

$$i = n_P + 1 \qquad i = n_P + 1 \qquad (4.34)$$

Hence, the efficiency of the reduced-system order in Eq. (4.25) will decrease by the amount of truncated efficiency components of the original system of Eq. (4.24).

This method of controller reduction in the  $\epsilon$ -space, because of the diagonal forms of the power matrices  $\Lambda_{11}$  and  $I_{11}$  is akin to the controller reduction via cost decoupled coordinates discussed in Ref. (7).

The controller reduction in the efficiency space is attractive from the point of view of that the reduced-order controller (4.26) requires no new computations for the control design since it is readily obtained by disregarding the control gains associated with the  $\epsilon_2$ -states in the original control law (3.30). Furthermore, the efficiency of the reduced system (4.25) requires no new computation either, since it is readily given by truncating the efficiency components  $e_i$  of the original system at  $i=n_p$  in (4.33). The method also has some disadvantages. The reduced order system (4.25) neglects dynamic coupling with the  $\epsilon_2$ -states which has to be assessed although one hopes that truncation of the least efficient states  $\epsilon_2$  results in truncation of least important dynamic coupling terms. Because of ignoring this dynamic coupling which is referred to as dynamic spillover between  $\epsilon_1$  and  $\epsilon_2$ , the sole assessment of the reduced-order controller (4.26) can only be obtained by simulation of the 2n-th order system (4.27) instead of the n<sub>p</sub>-th order system. However, in this case one notes that the reduced control to be simulated in (4.27) has the form  $F=G_1^e\epsilon_1$  instead of  $F=G_1^{e}\epsilon_1$ . If one were to evaluate (4.27) by using the latter form of reduced control using feedback of  $\bar{\epsilon}_1$ , this would constitute an open-loop control with respect to the  $\epsilon_1$  and  $\epsilon_2$  dynamics and the system (4.27) would still remain neu-The problem is that although the  $\epsilon$ -states (or any other trally stable. cost/power decoupling states as presented in Refs. 11, 12) diagonalize the power matrices  $P_N$  and  $P_D$  they remain dynamically coupled. The truncation of dynamically coupled states always leads to uncertainty of the behavior of the retained states in the actual system. In contrast, truncation of dynamically decoupled states still insures that the retained states in the actual system behave exactly as they do in the reduced-order model of the system. Certainly structural modal coordinates have precisely this feature and this should point

out why they are ever so desirable in the analysis and synthesis of structural dynamic systems. This being the case, it makes practical sense only to consider the reduced order controller form as  $F=G_1^{e}\epsilon_1$  instead of using feedback of  $\epsilon_1$  in (4.25) which leads one to consider Eq. (4.27) which in turn constitutes a closed-loop system. Finally, because the efficiency  $e_p$ , Eq. (4.33) of the reduced-order system given by (4.25) is obtained without regard to dynamic spillover between  $\epsilon_1$  and  $\epsilon_2$  states, the true efficiency of the dynamics of  $\epsilon_1$  states in (4.27) when the dynamic spillover is considered will be different from  $e_p$ , ta (4.33). This effect can indeed be significantly different in some cases so that the simplistic optimism provided by Eq. (4.34) to anticipate the change in the efficiency becomes unrealistic.

The evaluation of the stability, efficiency and the response of the system (4.27) will necessitate a recycle of the stability and efficiency analysis. This being the case, it proves more convenient to readdress the  $\epsilon$ -state truncation approach in terms of the original modal states x of which the structure-control designer has a better understanding. In this form, when studied in the modal space, the re-evaluation of the  $\epsilon$ -space controller reduction is tantamount to an efficiency filtering method. This we show in the next section.

## 4.4 Efficiency Filtering

Implementation of the reduced-controller in the form of Eq. (4.26) requires estimation of the  $\epsilon_1$  states which will theoretically be different than those states  $\epsilon_1$  in (4.27) with dynamic spillover considered. However, from an implementation point of view, there is no advantage gained in feeding back the  $\epsilon_1$  as discussed in the previous section, as opposed to feeding back the actual  $\epsilon_1$ 

states with which there is an advantage over  $\epsilon_1$  for the reasons just discussed. Thus, we consider the reduced feedback controller

$$\mathbf{F} = \mathbf{G}_1^{\mathbf{e}} \boldsymbol{\epsilon}_1 \tag{4.35}$$

Then, Eq. (4.27) represents the actual system in the  $\epsilon$ -space. However, the control law (4.35) and the system (4.27) can be transformed back to the original modal x-space. Write the control law (4.35) as

$$\mathbf{F} = \begin{bmatrix} G_1^{\bullet} & 0 \end{bmatrix} \begin{bmatrix} \epsilon_1 \\ \epsilon_2 \end{bmatrix}$$
(4.36)

an.1 note

$$\begin{bmatrix} \epsilon_1 \\ \epsilon_2 \end{bmatrix} = T^{-1} \mathbf{x} = \begin{bmatrix} L_1 \\ L_2 \end{bmatrix} \mathbf{x} \qquad , \quad [\mathbf{G}_1^e \ \mathbf{G}_2^e] = \mathbf{C}[\mathbf{T}_1 \ \mathbf{T}_2] \qquad (4.37)$$

Introducing the partions into (4.36), we obtain

$$\mathbf{F} = \mathbf{GT}_{1}\mathbf{L}_{1}\mathbf{x} = \mathbf{G}\mathbf{x} \tag{4.38}$$

which yields the new closed-loop dynamics

$$\mathbf{x} = [\mathbf{A} + \mathbf{B}\mathbf{G}]\mathbf{x} = [\mathbf{A} + \mathbf{B}\mathbf{G}\mathbf{T}_{1}\mathbf{L}_{1}]\mathbf{x}$$
(4.39)

Therefore, the  $\epsilon$ -space truncation of  $\epsilon_2$  states is equivalent to full x-state feedback through the new feedback gain matrix  $\overline{G}$  which simply represents a projection of the original G matrix through the matrix  $T_1L_1$  which is readily available as soon as the  $\epsilon_1$  partition is decided. The matrix  $T_1L_1$  essentially filters out the contribution of the least efficient  $\epsilon_2$  states to the x-states. Hence, we refer to this method as efficiency filtering.

Alternately, noting that

$$T_{1}L_{1} = I - T_{2}L_{2} \tag{4.40}$$

the closed-loop system becomes:

$$\dot{\mathbf{x}} = [\mathbf{A} + \mathbf{B}\mathbf{G}(\mathbf{I} - \mathbf{T}_2\mathbf{L}_2)]\mathbf{x} = [\mathbf{A}_{CL} - \mathbf{B}\mathbf{G}\mathbf{T}_2\mathbf{L}_2]\mathbf{x}$$
 (4.41)

or

$$\dot{\mathbf{x}} = [\mathbf{A}_{\mathrm{CL}} + \Delta \mathbf{G}]\mathbf{x}$$
,  $\Delta \mathbf{G} = \mathrm{B}\mathrm{GT}_{2}\mathrm{L}_{2}$  (4.42)

hence, the closed-loop systems of Eqs. (4.39) and (4.42) are equivalent and they can also be viewed as the perturbation of the original system (3.31) through the modal gain error matrix  $\Delta G$ . Because all matrices involved in  $\Delta G$  are known, the gain perturbation is a structured perturbation. The stability and the performance robustness (perturbation in  $P_G^M$  and  $P_R$ ) of the perturbed system (4.42) can now be studied by the methods of Refs. (12) and (13). From the performance bounds on  $P_G^M$  and  $P^R$ , the robustness of efficiency to filtering the  $\epsilon_2$  states can easily be inferred. We remind that, the study of stability and performance of efficiency filtered closed-loop system (4.42) in terms of the x-state is equivalent to the study of the  $\epsilon$ -state Equation (4.27) by using the reduced-controller (4.35).

The  $\epsilon$ -state truncation and the efficiency filtering technique as a way of obtaining and evaluating reduced order controllers do not seem attractive since the approach anticipates a lower efficiency in the reduced system. Furthermore, the truncation of efficiency states represents a further degree of abstraction in physical understanding of the controller-reduction in contrast to truncation of modal states. Also, there is the problem of dynamic spillover associated with the approach, therefore, it is plausible to search for controller reduction criteria which will culminate in an increase in the efficiency of the reduced-order system and the reduction is affected in terms of the original modal (x-space) states. At the very least, if no significant degradation of the efficiency is experienced in such a reduction process, the method would be preferable over the  $\epsilon$ -state reduction method. The underlying philosophy of such a method is then to identify and truncate modal coordinates x in the design model which have significant inefficiency associated with them so that their deletion will make more efficient use of control power available to the modes retained thereby resulting in an increase in the efficiency of the controller despite a reducedorder system.

# 4.5 Efficient Modal Controller Reduction

The premise of the controller reduction methodology we shall propose in this section is that the structural modal states  $x_j$  that contribute least to a given set of principal efficiency components  $e_i$  along the associated principal controller directions are to be deleted. Conversely, the modal coordinates which contribute most to a given set of spillover components  $sq_i$  along the associated principal controller directions are to be deleted. It turns out the two views do not necessarily yield the same set of truncated modal states.

The method involves computation of coupled or decoupled component efficiencies  $e_j$  as defined by Eqs. (4.16) and (4.17) for an initial, structuralcontrol design model and performing the 2n-th order efficiency modal analyses for one or more of the quotients e, e\* sq and sq\*. Upon studying the characteristic eigenvalues and the principal efficiency components of these quotients along the principal controller directions the significant components of relevant efficiency and/or spillover quotients are recognized to retain  $n_p \leq 2n$  principal directions in the computation of component efficiencies  $e_j$  according to Eqs. (4.20)-(4.22) where the summation over i involves  $n_p$  controller directions. Note that depending on which quotients are studied, the quotient  $e_j$  in Eqs. (4.20)-(4.22) could specifically represent any of quantities  $e_j$ ,  $e_j^*$ ,  $sq_j$  and  $sq_j^*$ .

After computing the component efficiencies, one then retains a desired number  $n_R$  of the structural components with the highest component efficiencies  $e_j$  and/or  $e_j^*$ . Conversely, if the above analyses are based on spillover quotients one retains  $n_R$  structural modal coordinates with the least component spillover

quotients  $sq_i$  and  $sq_i^*$ . As shown in Section (3.3) e and sq have the same eigenvectors, that is the same principal controller directions and their eigenvalues are related as  $(\lambda^e)$  and  $(\lambda^s) = (1-\lambda^e)$ . Therefore, if e and sq are considered, modal analysis of either one will be sufficient. If, however, e\* and  $sq^*$  are desired, no immediate relationship between their eigenvectors and eigenvalues are recognized, so modal analyses for both of them must be performed. Furthermore, recall that since e, e\* and sq cannot be greater than unity their characteristic eigenvalues cannot be greater than unity. The eigenvalues of the global spillover quotient sq\* can be greater than unity. After selecting the  $n_R$ structural modal coordinates as suggested, two avenues are available. Restating here for convenience the original control design model

$$\dot{\mathbf{x}} = \mathbf{A}\mathbf{x} + \mathbf{B}\mathbf{F}$$
,  $\mathbf{F} = \mathbf{G}\mathbf{x}$ 

on which the efficiency analyses are performed, the first avenue is to retain only the  $2n_R$  states  $x_R$  and truncate the remaining modal states and implement the reduced order model

$$\dot{x}_{R} = A_{R}x_{R} + B_{R}F$$
,  $F = G_{R}x_{R}$   $2n_{R} < 2n$  (4.43)

without redesigning the feedback control law where  $A_R$ ,  $B_R$  and  $G_R$  are the partitions of A, B and G corresponding to the  $2n_R$  states  $x_R$ . This is, in general, the most common spirit of controller reduction. To assess the effectiveness of the reduced order model the stability and new efficiency of the system (4.43) must be studied. In general, the characteristic efficiencies and the modes of the controller will change. But, if the reduced-order model identified is a good one its efficiency features will have changed over the original system in the direction that the analysis anticipates, that is truncation of least efficient or most inefficient modal states should leave behind at least a more efficient reduced model.

The second avenue available for controller reduction, rather than truncating the original control law, is to redesign the feedback control law for the set of modal states  $x_R$ 

$$\dot{\mathbf{x}}_{R} = \mathbf{A}_{R} + \mathbf{B}_{R}\mathbf{F}_{R} \qquad , \ \mathbf{F}_{R} = \mathbf{G}_{RR}\mathbf{x}_{R} \qquad (4.44)$$

where  $G_{RR}$  is the new redesigned control gain matrix for the reduced-order model for the reduced structural system. For example, if the initial control design is obtained via LQR it will be repeated for the system with the performance measure

$$2J = \int (x_R^T W_{xR} + F_R W_{CR} F_R) dt \qquad (4.45)$$

Of course, the control redesign is a conservative approach. It guarantees the stability of the reduced order model and its efficiency can be reanalyzed. Intuitively one would stand a better chance of improving it with the control redesign.

The redesigning of the controls is viewed here as a more realistic and, in fact, practical approach to the model/controller reduction. The reduced order model (4.44) is smaller than the original control design model for which algorithmic and theoretical capability exists to begin with. Consequently, application of the same capability to a reduced order system by using the same tools should not be a problem. Furthermore, the process of identifying a good reduced order model is an off-line procedure and performing control redesigns and efficiency modal reanalyses do not require more computational resources than available to perform the original 2n-th order design. Countless, aimless redesigns are not advocated here to identify a good reduced model that satisfies the system constraints. On the contrary, the efficiency modal analysis presented heretofore, does lay a foundation for the designer for a very much directed and purposeful avenue to do the redesigns efficiently so that at least one perhaps only a few redesigns will finish the job.

The final point that must be noted is that both types of reduction concepts discussed in this section and in Section 4.4, that is the efficient, modal control reduction and the efficiency-filtering can be used simultaneously. The truncation of modal structural states based on the criterion of this section to retain a  $2n_R$ -order structural dynamics model can be viewed as a closed-loop model reduction. On the other hand, the truncation of efficiency-states to obtain a control input as the feed back of a reduced number of  $n_p$ -efficiency states can be viewed as the closed-loop controller reduction. The control law design can thus, be designed for a  $2n_R$ -th reduced-order design model in the form of reduced  $n_p$ - $\epsilon_1$ -states' feedback, thus, affecting a simultaneous model/controller reduction technique.

In the following section, we discuss a number of additional aspects to guide the designer in using the efficiency approach. The process readily produces more efficient designs as we will present in the next section.

#### 4.6 Further Considerations for Efficient Model/Controller Reductions

Certain other features of the approaches discussed in the chapter, must be noted. Consider the pair e, sq for the design model. Their components  $e_i = c_i^2 \lambda_i^e$ and  $sq_i = c_i^2(1 - \lambda_i^e)$  have the property that the indices i in the set  $\{e_i\}$  with the highest  $n_p$  components will not necessarily correspond to the indices i in the set  $\{sq_i\}$  with the lowest  $n_p$  components. This anomaly depends on the relative values of the coefficients  $c_i^2$  which are normalized square initial disturbances along the principal controller directions. As this may seem disheartening, one cannot, however, change the fact that efficiency or the inefficiency of the structure control system can be a strong function of the initial disturbance  $x_0$ . Some disturbances are better handled by a given system design whereas others result in poor efficiency performance.

In choosing the  $n_p$  efficiency components  $e_i$ , we do not necessarily consider the highest  $n_p$  of them although this seems like a logical choice. Further care must be taken to identify the set  $\{e_i, i=1, \ldots, n_p\}$ . A principal efficiency component  $e_i$  may be high either because of a high  $c_i^2$  and low  $\lambda_i^e$ , or low  $c_i^2$  and high  $\lambda_i^e$  or both. Recalling that a spillover quotient  $sq_i$  is given by (3.45), in the first case, the corresponding spillover coefficient  $sq_i$  will also be high, in the second case,  $sq_i$  will be low and in the third case it will be relatively lower. All depends on relative separations of  $c_i^2$  and  $\lambda_i^e$ , so retaining a high  $e_i$  may also mean retaining a high spillover quotient  $sq_i$ . The word of caution is that before dominant  $e_i$ 's are taken into account corresponding  $sq_i$ 's must also be scrutinized. Similar caution must be taken in case of  $e_i^*$  and  $sq_i^*$ .

Along the same lines, if an initial control design has a low efficiency, it makes little sense to select modes based on their relative contributions to an already low efficiency. Instead, one should analyze their contributions to a high spillover quotient and retain or delete them based on that account. Conversely, one should bring in new modes into the control design model that were discarded in the original analysis.

The model reduction process that is being discussed here seems to assume that the controller configuration (number of inputs and their locations) is fixed. In reality, however, model reduction (or mode selection) process and the controller configuration are interdependent. For design purposes, one would normally try to achieve a desired level of efficiency for a reduced model order of  $2n_R$  and with an upper bound on the number of inputs  $m \leq m_{max}$ . Hence, because

the efficiencies are also a function of controller configuration, the controller reduction process should also utilize the controller configuration changes. The ultimate goal is, for the given initial disturbance, to identify the best reduced order model with the best input configuration to go with it to achieve a high efficiency for the structure-control system.

In the above discussion, it is the initial disturbance represented in the efficiency coefficients  $c_i^2$  responsible for not matching the set  $\{e_i\}$  vs.  $\{sq_i\}$  of  $\{e_i^*\}$  and  $\{sq_i^*\}$ . Hence, for a given control design, selection of the best principal controller directions will change if the initial disturbance changes. An alternate approach to selecting the set  $n_p$  of principal controller directions which avoids the initial disturbance is to consider the controller directions associated with the  $n_p$  highest characteristic efficiencies  $\{\lambda_i^e \ i=1,\ldots,n_p\}$ . Another feature of this choice is that the highest set  $\{\lambda_i^e=2,\ldots,n_p\}$ .

Recall from Section 3, that from the properties of a Rayleigh's quotient efficiency is bracketed by  $\lambda_{max} \ge e \ge \lambda_{min}$ . It follows that a control system whose efficiency will be least sensitive to the initial disturbance  $x_o$  must have a minimal separation between  $\lambda_{max}$  and  $\lambda_{min}$ , that is if the characteristic efficiencies are dense in a short interval. Ideally, if  $\lambda_{max} = \lambda_{min}$ , the control system will be uniformly efficient, better said equalized in all principal directions and will have the same efficiency regardless of  $x_o$ . This would be desirable feature. In addition, if  $\lambda_{min}$  can be made high then the system will also have a high efficiency regardless of  $x_o$ . The price that must be paid to make the structure-control system efficiency insensitive to  $x_o$  is that all principal controller directions will now become equally significant, hence, truncation of efficiency states will not be possible as each  $\epsilon$ -state and its principal direction will contribute equally to the efficiency. Therefore, one will not be able to discriminate among the principal controller directions to affect their truncation in the  $\epsilon$ -space.

Another feature related to this last observation is the sensitivity of the stability of the system to the principal controller directions. Since stability of a linear system does not depend on the initial conditions, the stability of a system truncated based on the efficiency will depend not on the magnitudes of the truncated efficiency components  $e_i(i-n_n+1,\ldots,2n)$  but on the modal properties  $\lambda_i$  and  $t_i$  associated with the truncated  $e_i$ . The more significant the modal properties of the truncated principal controller directions are, the easier it will be to destabilize the system via efficiency truncation. Again, because a poor  $x_o$  can associate poor component efficiencies  $e_i$  with significant controller modes, efficiency reduction based on truncation of controller modes of poor e, will conflict with maintaining closed-loop stability. Stability robustness, then, to efficiency-state truncation should be good if the truncated controller modes of poor e<sub>i</sub> are not significant. On the other hand, if the insensitivity of the efficiency of the structure-control system to initial disturbance is desired, since this will necessitate equalization of the characteristic efficiencies, each controller mode will become equally important. Consequently, insensitivity to  $x_o$  of the efficiency of the system, in the presence of a desire to controller mode truncation, will come at the expense of system with a sensitive stability.

In all of these endeavors, the efficiency and spillover quotients and the modal analyses serve as the design directors qualifying and quantifying the merits of design changes to achieve the best possible performance with high efficiency. In the next section, a number of illustrations are presented utilizing

all that is discussed in this chapter. Again, the particular structure is the ACOSS-4 tetrahedral structure used in the previous chapters.

### 4.7 <u>Illustrative Examples on ACOSS-4</u>

The same situation as in Section 3.6 is considered.

#### Decomposition of Efficiency Quotients

For the sake of brevity, the decomposition of the numerator and denominator control power matrices are presented only for a 4-th order control design model. The efficiency components  $e_i$  and coupled and decoupled component efficiencies of the structural modes are also shown in Appendix C.

- Example 4.1: This is the same case as in Examples 3.1. The numerator/denominator control power decomposition matrices [N] and [D] are given on in Appendix C for both global and model efficiencies. Efficiency components are  $e_i$ ,  $e_i^*$  and component efficiencies  $e_j$ ,  $e_j^*$  for coupled (Method 1) and decoupled (Method 2) cases are also shown.
- Example 4.2: The case of Example 3.4 is given with the same information listed under Example 4.1.

## Efficiency Decomposition Using np-principal Controller Modes

Example 4.3: The same system of Example 4.2 is considered. However, only the six  $(n_p=6)$  highest characteristic global and relative model efficiencies were used in the decomposition of numerator/denominator control power matrices and the component efficiencies of the

structural modes were based on the 6 retained dominant controller directions.

## In/Efficiency - filtering: Controller Reduction

Example 4.4: Example 3.5 with n=8 (natural order) modes and m=4 was considered. The original model had e\*=30.75%, e=80.65%, sq\*=0.628 and sq-19.35%. Although the model efficiency e is good, the global efficiency e<sup>\*</sup> seems relatively poor. Hence, an inefficiency of 70% is associated with the global efficiency which is reflected in the value of sq\*. (Note that by definition  $sq \neq 1-e^*$ ). Since there is more inefficiency associated with e\* than efficiency, the modal analysis was done on sq\*. The efficiency-filtering was performed on the modes of sq\* by truncating the highest  $n_p$  global spillover efficiency components  $sq_i^*$   $i=1,2,\ldots,n_p$ . Up to  $n_p$ spillover components could be filtered without destabilizing the original control design. The original gain matrix G was simply projected through  $G=GT_2L_2=G(1-T_1L_1)$ , as in Eq. (4.41) to obtain the filtered gain matrix G. Note that the  $n_p$  principal global spillover commonents were deleted, not retained, so it is necessary to retain the remaining components represented by subscript 2. Conversely, the projection  $(1-T_1L_1)$  eliminates the highest contributing n<sub>p</sub> states represented by subscript 1. Simulation results for  $n_p=4$  and  $n_p=8$  are shown by figures SIM32 (Figures 4.1, 4.2 and 4.3) and SIM34 (Figures 4.4, 4.5 and 4.6), respectively. The efficiencies of both of these sq\*-filtered cases were recomputed. For  $n_p=2$ , the new efficiency quotients

were  $e^{*}=26/1$ %,  $e^{-}\rightarrow .32$ %; for  $n_p=4$  the new values were  $e^{*}=40.9$ %,  $e^{+}73$  189 Hence, sq\*-filtering did improve the global efficiency over the original design by about 1/3. Other pertinent details are shown on the computer outputs listed in Appendix C. Note that the simulation results show hardly any difference from the original design SIM10 (Figures 3.10, 3.11, and 3.12). The consequence of this particular study is that the 16th order design model can be controlled by the feedback of only 8 efficiency states, hence, resulting in a 50% reduction in the controller size.

#### Efficient Model Selection (Reduction)

Example 4.5 The model efficiency of the 16-th order control design model of the first 8 natural modes with four inputs (n=8 natural, m=4)which was studied in Example 3.5 and Example 4.4 was e=80.25%. In an effort to improve the relative model efficiency e further by selecting a different set of 8 modes other than the lowest 8 natural modes, the component efficiencies of Example 3.5 were scrutinized. The component efficiencies for both e and e\* are listed in Appendix C. Modes 1, 2, 7 and 8 have the lowest coupled component efficiencies  $e_j^*$  and  $e_j$  according to Method 1. Hence, these modes were replaced with the modes that were not in the original set. A redesign was done for the new 16th order design model with the new selected set  $n=\{Modes: 3-6, 9-12\}$ . The model efficiency increased to e=97.21%, although the global efficiency was reduced to e\*=10.27%. Simulation results are shown in Figures SIM35 (Figures 4.7, 4.8 and 4.9).



Figure 4.1: Globally optimal system LOS for n=8 (modes: 1-8); example 4.4; controller reduction via efficiency filtering for  $n_p$  =4


Figure 4.2: Suboptimal system LOS for n=8 (modes: 1-8), m=4 (#1-4); example 4.4; controller reduction via efficiency filtering for  $n_p=4$ 



Figure 4.3: Evaluation model LOS for n=8 (modes: 1-8), m=4 (#1-4), example 4.4; controller reduction via efficiency filtering for  $n_p=4$ 



Figure 4.4: Globally optimal system LOS, n=8 (modes: 1-8), example 4.4; controller reduction via efficiency filtering for n<sub>P</sub>=8



Figure 4.5: Suboptimal system LOS, n=8 (modes: 1-8), m=4 (#1-4), example 4.4; controller reduction via efficiency filtering for  $n_p=8$ 



Figure 4.6: Evaluation model LOS, n=8 (modes: 1-8), m=4 (#1-4), example 4.4; controller reduction via efficiency filtering for  $n_P=8$ 



Figure 4.7: Globally optimal system LOS, n=8 (modes: 3-6, 9-12); example 4.5; efficient model selection reduction









Figure 4.9: Evaluation model LOS, n=8 (modes: 3-6, 9-1?) m-4 (#1 4), example 4.5; efficient model selection (reduction)

### 5.0 EFFICIENCY OF SYSTEMS WITH COMPENSATOR DYNAMICS

#### 5.1 Introduction

The developments presented in the previous chapters were based on a static state feedback control law F-Gx. In this chapter, we consider the case whereby the feedback control law is obtained via a dynamic compensator and extend the efficiency concepts studied in Chapters 2-4 to consider the effect of compensator dynamics. The extension of the previous results, in this case, is straightforward and the idea of compensator efficiency within the structure-control system arises naturally.

In Section 5.2, efficiency quotients are revisited including the compensator dynamics. Two definitions of compensator efficiency are given following the component efficiency definitions introduced in Section 4. A most common dynamic compensator is the state-estimator in the control loop. In Section 5.3, we, therefore, pay special attention to the design of a reduced-order state estimator relevant to the subject of this research. In particular, an  $n_p$ -th order efficiency state estimator design is given where  $n_p$  efficiency states  $\epsilon_1$ are estimated to implement the reduced-order control law  $F=G_1^e\epsilon_1$  discussed in Sections 4.3 and 4.4. Here  $n_p$  is the number of dominant principal controller directions which will have been identified from an efficiency modes analysis of the system, hence,  $n_p < 2n$ . Finally, in Section 5.4 we consider formally the quantitative and qualitative effect of control power spillover on the response of the residual dynamics not considered in the 2n-th order control design model. Although this topic may have been discussed in Chapter 2, we postponed its presentation until after the discussion of compensator dynamics, so that it will seal appropriately all the developments presented heretofore and conclude this

investigation by underlining the critical nature of the efficiency of structurecontrol systems.

No illustrative examples are presented in this chapter

# 5.2 <u>Efficiency of Structure - Control Systems with Compensator Dynamics -</u> <u>Compensator Efficiency</u>

In the previous chapters we assumed that the control law was a simple state-feedback. In this chapter, we generalize the concepts to the case where the control law is obtained via a dynamic compensator. Again, the system is described by the modal state-space equations

$$\dot{\mathbf{x}} = \mathbf{A}\mathbf{x} + \mathbf{B}\mathbf{F} \tag{5.1}$$

The controller is in this case defined by the following dynamic compensator equations with an  $n_c$ -dimensional state vector  $x_c$ .

$$\dot{\mathbf{x}}_{c} = \mathbf{A}_{c}\mathbf{x}_{c} + \mathbf{B}_{c}\mathbf{F} + \mathbf{K}_{c}\mathbf{y}$$
(5.2)

$$\mathbf{F} = \mathbf{G}_{c}\mathbf{x}_{c} + \mathbf{H}_{c}\mathbf{y} \tag{5.3}$$

where the  $m_v$ -dimensional system output y is given by

$$y = Cx \tag{5.4}$$

Then the closed-loop system is described by

$$\dot{\mathbf{x}} = \dot{\mathbf{A}}_{\mathrm{CL}} \ \mathbf{x}$$
,  $\dot{\mathbf{x}} = [\mathbf{x}^{\mathrm{T}} \ \mathbf{x}_{\mathrm{c}}^{\mathrm{T}}]^{\mathrm{T}}$  (5.5)

$$\bar{A}_{CL} = \begin{bmatrix} A+BH_{c}C & | & BG_{c} \\ - & - & - & - & - \\ B_{c}H_{c}C+KC & | & A_{c}+B_{c}G_{c} \end{bmatrix}$$
(5.6)

and the control law

$$\mathbf{F} = \mathbf{\tilde{G}}\mathbf{\tilde{x}} \quad , \quad \mathbf{\tilde{G}} = [\mathbf{H}_{c}\mathbf{C} \quad \mathbf{G}_{c}] \tag{5.7}$$

Since the definitions of the global efficiency and the model efficiency are defined in terms of only inputs F to the dynamic system not on how they are

computed or derived their definitions Eqs. (3.6-3.8), are still valid, hence, using the form of F as given by Eq. (5.3) in Eqs. (3.6), (3.7) we obtain

$$\bar{S}^{R} = \int F^{T} R^{R} F dt = \bar{x}_{0}^{T} \bar{P}^{R} \bar{x}_{0}$$
(5.8)

$$\bar{S}^{M} = \int F^{T} R_{C}^{M} F dt = \bar{x}_{O}^{T} \bar{P}_{C}^{M} \bar{x}_{O}$$
(5.9)

where  $\tilde{P}$  are the  $(n+n_c)$ -dimensional power matrices obtained from the Lyapunov Equations:

$$\bar{\mathbf{A}}_{\mathrm{CL}}^{\mathrm{T}}\bar{\mathbf{P}}^{\mathrm{R}} + \bar{\mathbf{P}}^{\mathrm{R}}\bar{\mathbf{A}}_{\mathrm{CL}} + \bar{\mathbf{G}}^{\mathrm{T}}\mathbf{R}^{\mathrm{R}}\bar{\mathbf{G}} = 0$$
(5.10)

$$\bar{A}_{CL}^{T}\bar{P}^{M} + \bar{P}^{M}\bar{A}_{CL} + \bar{G}^{T}R_{C}^{M}\bar{G} = 0$$
(5.11)

in which  $\mathbb{R}^{\mathbb{R}}$  and  $\mathbb{R}^{\mathbb{M}}_{\mathbb{C}}$  are still as given by Eqs. (3.6) and (3.7)

$$R^{R} = D^{T}M^{-1}D$$
 ,  $R^{M}_{C} = D^{T}E_{n}E_{n}^{T}D$ 

The model efficiency e and the spillover quotient sq are computed exactly as described in Chapter 3 by using

$$P_{N} = \bar{P}^{M} , P_{D} = \bar{P}^{R}$$

$$(5.12)$$

and all that is discussed in Chapters 3 and 4 will apply equally well to control with compensator dynamics. The global efficiency e\*, however, needs a little careful consideration. Recall that this definition involved control of the DPS with a continuously distributed input profile, where the optimal control was the independent feedback of the modal states of the DPS structure. An important feature of this optimal DPS control is that it is dynamically similar to the dynamics of the control design model of the structural modes in the closed-loop system. Hence, even in the case of a dynamic compensator, we shall insist on the similarity of the DPS distributed control design to the structural modes alone, regardless of the modes in the closed-loop system due to the additional compensator dynamics. Thus, if  $(\rho) = (\tilde{\rho}_{\rm S}, \tilde{\rho}_{\rm C})$  is the eigenvalue spectrum of the closed-loop system, Eq. (5.5), with the compensator, as a union of the structural eigenvalue spectrum  $(\tilde{\rho}_{\rm S})$  and the compensator eigenvalue spectrum  $(\tilde{\rho}_{\rm C})$ , the control

power S\* of the optimal distributed solution will be found by using  $(\rho_S)$  in the equations given in Section 2.4.

The global efficiency will then be

$$e^{\star} = -\frac{S^{\star}}{\bar{S}^{R}}$$
(5.13)

Although at this point, it is not clear whether S\* is greater or less than  $S^R$  with a dynamic compensator the quotient e\* will still preserve its function as a practical concept since it will relate the power performance of the actual design represented via  $S^R$  to an ideal baseline design S\* being a measuring stick. Specifically, the function of the compensator will be highlighted in such a setting. Thus writing the control power S\*

$$2S^{\star} = \mathbf{x}_0^{\mathrm{T}} \mathbf{P}^{\star} \mathbf{x}_0 \tag{5.14}$$

the global efficiency e\* will be given by

$$e^* = \frac{x_0^T P^* x_0}{\bar{x}_0^T \bar{p}^R \bar{x}_0}$$
(5.15)

where P\* is again computed as in Appendix B for the spectrum  $(\rho^*) = \{\bar{\rho}_s\}$ . Alternately, augmenting P\* with zeros

$$P * = \begin{bmatrix} P * & O_{n \times n_c} \\ O_{n_c \times n} & O_{n_c \times n_c} \end{bmatrix}$$

e\* can be written as

$$e^{\star} = \frac{\bar{x}_{0}^{T}\bar{p}^{\star}\bar{x}_{o}}{\bar{x}_{0}^{T}\bar{p}^{R}\bar{x}_{o}}$$
(5.17)

Next, partition the power matrices  $\bar{P}_C^M$  and  $\bar{P}^R$  corresponding to the structural modal states x and the compensator states  $x_c$ 

$$\tilde{P}^{R} = \begin{bmatrix} P_{S}^{R} & P_{SC}^{R} \\ P_{SC}^{RT} & P_{C}^{R} \end{bmatrix} , \quad \tilde{P}_{C}^{M} = \begin{bmatrix} P_{S}^{M} & P_{SC}^{M} \\ P_{SC}^{MT} & P_{C}^{M} \end{bmatrix}$$
(5.18)

where  $P_S$  and  $P_C$  are nxn and  $n_c x n_c$  dimensional, and  $P_{SC}$  are nxn<sub>c</sub> dimensional matrices. The subscripts S,C and SC correspond to the structural modal states, compensator states and the coupling between the structural and compensator states, respectively. The model efficiency e can now be written as in Section 4.2.

$$e = N/D$$

where

$$N = x_0^T P_S^M x_0 + 2x_0^T P_{SC}^M x_{c0} + x_{c0}^T P_C^M x_{c0}$$
  
$$D = x_0^T P_S^R x_0 + 2x_0^T P_{SC}^R x_{c0} + x_{c0}^T P_C^R x_{c0}$$
(5.19)

and  $x_o$  corresponds to the initial disturbance of structural states. In practice, the question of how to choose the compensator initial conditions  $x_{CO}$  arise. From the above decomposition it is clear what the function of  $x_{CO}$  can be in terms of efficiency of the controller. For  $x_{CO}=0$ , the efficiencies become

$$e = \frac{x_0^T P_S^M x_0}{x_0^T P_S^R x_0}, \quad e^* = \frac{x_0^T P_S^* x_0}{x_0^T P_S^R x_0}$$
(5.20)

where  $P_s^R$  and  $P_s^M$  are the nxn upper left partitions of  $\bar{P}^R$  add  $\bar{P}^M$  obtained from the  $(n+n_c)$ -dimensional Lyapunov Equations (5.10 and 5.11). In the case of e\* the care that had to be observed in defining it by Eq. (5.13) resolves itself automatically regardless of augmentation of P\* with zeros in Eq. (5.16) and it would

be plausible to require independent feedback of structural modal states to yield P\*.

On the other hand, if the compensator initial states are non-zero, the efficiencies e and e\* will be different than what will be obtained by Eqs. (5.20). Indeed, a careful choice of  $x_{CO}$  can enhance the efficiency of the control system with the participation of coupling terms. The case studies we presented in earlier in Chapter 4 demonstrated that coupling terms in the control powers could be beneficial.

This brings us to the point of defining compensator efficiency of the control system as a contribution to the overall efficiencies e and e<sup>\*</sup>. This can be done easily considering the developments presented in Section 4 on component efficiencies. The concepts presented therein can be generalized to this case by associating the subscript j in the equations with the structural modal states and compensator states, hence, we consider j=x,c and rewrite Eq. (4.1) as:

$$e = \Sigma e_{j} = e_{s} + e_{c}$$
(5.21)

where  $e_j = e_s$  and  $e_j = e_c$  follow from Eqs. (4.16) and (4.17)

$$e_{s} = \frac{ND_{s} + N_{s}D}{2D^{2}}$$
,  $e_{c} = \frac{ND_{c} + N_{c}D}{2D^{2}}$  (5.22)

$$D_{s} = Trace[\bar{P}^{R}\bar{X}_{o}]_{s} , D_{c} = Trace[\bar{P}^{R}\bar{X}_{o}]_{c}$$
(5.23)

$$N_{s} = \operatorname{Trace}[\bar{P}^{M}X_{o}]_{s} , N_{c} = \operatorname{Trace}[\bar{P}^{M}X_{o}]_{c}$$
(5.24)

$$\bar{X}_{o} = \bar{x}_{o} \bar{x}_{0}^{T}$$
(5.25)

Similarly for e\*

$$e^* = \Sigma = e_j^* = e_s^* + e_c^*$$
 (5.26)  
j

$$\frac{N*D_{s} + N_{s}^{*}D}{2D^{2}}, e_{c}^{*} - \frac{N*D_{c} + N_{c}^{*}D}{2D^{2}}$$
(5.27)

$$N_{s}^{*} = Trace[P*X_{o}] = S* = N*$$
 ,  $N_{c}^{*} = 0$ 

 $D_s$ ,  $D_c$  and D are as given by Eqs. (5.23), (5.24) and (5.19). We shall refer to  $e_c$  and  $e_c^*$  as the compensator efficiencies.

Alternately, decoupled compensator efficiencies can be defined as in Section 4.2. It is easy to see that, in this case

$$e_{j} = \frac{n_{j}}{d_{j}}$$
  $e_{j}^{*} = \frac{n_{j}^{*}}{d_{j}}$   $j=s,c$  (5.28)

$$n_{s} = x_{0}^{T} P_{S}^{M} x_{0}$$
 ,  $n_{c} = x_{C0}^{T} P_{C}^{M} x_{C0}$  (5.29)

$$d_{s} = x_{0}^{T} P_{S}^{R} x_{0}$$
,  $d_{c} = x_{c0}^{T} P_{C}^{R} x_{c0}$  (5.30)  
 $n_{s}^{\star} = x_{0}^{T} P^{\star} x_{0}$ ,  $n_{c}^{\star} = 0$ 

and the expected efficiencies are as in Eq. (4.19)

$$E(e) = \frac{n_s + n_c}{d_s + d_c}$$
,  $E(e^*) = \frac{n_s^*}{d_s + d_c}$  (5.31)

which rely on the expectation that the cross-coupling terms between the structural and compensator states will average out to zero.

We must emphasize that the developments resented here do not depend on the specific control theory used to design the controls. In fact, the efficiency of the system will serve to evaluate the control design in a most essential way. For example, the compensator can again be designed as optimal compensators in the sense of Linear Quadratic Gaussian (LQG) designs or as fixed order optimal compensators via Minimum Energy Optimal Projection (MEOP) technique (Ref. 14). The efficiency of all such design will still need to be considered as addressed in this investigation.

### 5.3 <u>Reduced-Order Efficiency-State Estimators</u>

The feedback control law

F = Gx

as given in the previous chapters requires the knowledge of structural modal states of the 2n-th order control design model. Typically, a full order state-estimator might be used to implement the feedback law, Eq. (3.10) by using an estimate  $\hat{x}$  of the structural states. The 2n-th order state-estimator constitutes a dynamic compensator discussed in Section 5.2, Eqs. (5.2 and 5.3) and has the form

$$\hat{\mathbf{x}} = \hat{\mathbf{A}}\hat{\mathbf{x}} + \hat{\mathbf{B}}\mathbf{F} + \mathbf{K}\mathbf{y}$$
(5.32)

$$\mathbf{F} = \mathbf{G}\hat{\mathbf{x}} \tag{5.33}$$

where ^ denotes estimator matrices to be found and G is the mx2n dimensional control gain matrix designed for the system by any suitable approach. The solution for the estimator in terms of the compensator notation of Section 5.2 is known to be

 $\hat{A} = A-KC = A_c$ ,  $\hat{B} = B = B_c$ ,  $G = G_c$ ,  $H_c = 0$  (5.34) where K is the estimator gain chosen to assign a desired set of estimator (compensator) eigenvalues ( $\rho_c$ ). In this case, the separation principle holds and the 4-nth order closed-loop eigenvalues will be  $(\bar{\rho}) = \{\bar{\rho}_S \ \bar{\rho}_C\}$  where  $\bar{\rho}_S$  are the eigenvalues of the closed-loop matrix  $A_{CL}=A+BG$  for the structural dynamics.

Hence, all of the developments in Section 5.2 apply to this full-state estimator and the impact of the estimator design on the overall system efficiency can be studied by following the procedure given therein. In this vein, the estimator efficiency concept surfaces as the compensator efficiency.

In practice, instead of full-order estimator, a reduced-order estimator of order  $n_c=2n-m_y$  can be designed where  $m_y$  is the number of measurements. This procedure is standard text book material and does not deserve any special attention here.

Another estimator which would be especially relevant to the theme of this investigation would be an efficiency-state estimator, that is an estimate  $\hat{\epsilon}$  of the efficiency state  $\epsilon$  introduced in Chapter 3. Since the efficiency state  $\epsilon$  is a 2n-dimensional vector, a full-order  $\epsilon$ -state estimator will not be of interest. Since it will merely be related to the full-order modal state estimator by the efficiency modal matrix

$$\hat{\mathbf{x}} = \mathbf{T}\hat{\boldsymbol{\epsilon}} \tag{5.35}$$

On the other hand, a reduced-order estimator in the  $\epsilon$ -space will be of interest if we recall the efficiency-filtering approach discussed in Section 4.4. We showed in Section 4.4 that a reduced-order control law, Eq. (4.35) F=G\_1^{e}\epsilon\_1 can be used where  $\epsilon_1$  are the dominant efficiency states identified along the n<sub>p</sub><2n principal controller directions. In Section 4.4, it was also shown that the control law was equivalent to a projection (filtering) of the 2n-dimensional structural states, implying essentially a full-order state feedback of the form

 $F = GT_1L_1x$ 

as given by Eq. (4.38).

However, the reduced-control law can be implemented in reality as the feedback of  $n_p$  retained efficiency states  $\epsilon_1$  if an estimate  $\epsilon_1$  of these states are available from an  $n_p$ -th order efficiency-state estimator. It turns out that design of such a reduced  $n_p$ -th order  $\epsilon_1$ -estimator is possible. Recalling that  $\epsilon_1$  and x are related via

$$\epsilon_1 = L_1 x$$

where  $L_1$  is the  $n_p x 2n$  upper partition of the T<sup>-1</sup> matrix, we recognize that  $\epsilon_1$  are  $n_p$  linear combinations of the structural modal states. An estimator, which can estimate these specific linear combinations of the x-states will be precisely what is needed to implement the reduced order efficiency state feedback control law (4.35). The design of estimators of linear combinations of states was presented in Ref. (15) in a different context and can be adapted in our case to design an  $n_p$ -th order  $\epsilon_1$ -estimator.

Assume the  $\epsilon_1$ -estimator in the form

$$\epsilon_1 = \hat{A} \hat{\epsilon}_1 + \hat{B}F + Ky \tag{5.36}$$

A, B and K are  $n_p x n_p$ ,  $n_p x m$  and  $n_p x m_y$  dimensional matrices to be found. Multiplying the modal structural dynamic equations on the left by  $L_1$ 

$$L_1 \dot{x} = L_1 A x + L_1 B F$$
 (5.37)

Subtract Eq. (4.37) from Eq. (4.36)

$$\hat{\epsilon} - L_1 \dot{x} = \hat{A}(\hat{\epsilon}_1 - L_1 x) + (KC - L_1 A + \hat{A}L)x + (\hat{B} - L_1)BF$$
 (5.38)

where the term  $AL_1x$  was added and subtracted on the right hand side. If K and B are chosen so as to satisfy

$$\dot{\mathbf{B}} = \mathbf{L}_1 \tag{5.39}$$

$$KC - L_1 A + AL = 0 (5.40)$$

and introducing the estimation error vector

$$\epsilon_{\mathbf{E}} = \hat{\epsilon}_1 - \mathbf{L}_1 \mathbf{x} \tag{5.41}$$

Eq. (5.38) reduces to

$$\dot{\epsilon}_{\rm E} = A\epsilon_{\rm E} \tag{5.42}$$

which has the solution

$$\epsilon_{\rm E} = e^{At} \epsilon_{\rm EO}$$
,  $\epsilon_{\rm EO} = \epsilon_{\rm E}(0)$  (5.43)

Hence, if the estimator matrix  $\hat{A}$  is chosen, such that it has eigenvalues with negative real parts, then  $\epsilon_{\rm E}(t)=0$  from which we obtain  $t\to\infty$ 

$$\lim_{t \to \infty} \hat{\epsilon}_1 = \lim_{t \to \infty} L_1 x$$

and the  $\epsilon_1$  will be estimated exactly as the desired combination of states x for large t.

If we make the simple choice

.

$$A = diag[\alpha_i] \quad \alpha_i < 0 \qquad i=1, \dots, n_p \tag{5.44}$$

equation (5.40) can be solved for the estimator gain matrix K by rewriting it as 2n-simultaneous equations

$$C^{T}K^{T} = A^{T}L_{1}^{T} - L_{1}^{T}A^{T}$$

$$(5.45)$$

where the  $2n \times m_y \ C^T$  matrix is assumed to have rank  $m_y$ ,  $m_y < 2n$ . Denoting the  $m_y$ dimensional row vectors of the estimator gain matrix K by  $k_i^T$  and the  $n_p$  column vectors of the right hand side of Eq. (5.45) by  $a_i$ ,  $i=1,2,\ldots,n_p$ , we write Eq. (5.45) in the form

The solutions to Eq.(5.46) can be obtained by singular value decomposition of the measurement matrix  $C^{T}$ 

$$C^{\mathrm{T}} = \mathrm{U}\Sigma\mathrm{V}\star\tag{5.47}$$

where U and V are 2nx2n and  $m_y x m_y$  unitary matrices; \* denotes complex conjugate transpose, and  $\Sigma$  is the 2nxm, matrix of singular values

$$\Sigma = \begin{bmatrix} \Sigma_0 \\ 0 \end{bmatrix} \qquad \Sigma_0 = \text{diag } \sigma_j \qquad j=1,2,\ldots,m_y \qquad (5.48)$$

The solution for the estimator gains  $\boldsymbol{k}_i$  can be shown to be

$$k_i = V \Sigma_0^{-1} U_1^* \alpha_i$$
  $i=1, ..., n_p$  (5.49)

where  $U_1^*$  is the myx2n upper partition of U\*. It follows that the estimator gain matrix K will be given by

$$K^{T} = V \Sigma_{0}^{-1} U_{1}^{*} (A^{T} L_{1}^{T} - L_{1}^{T} A^{T}).$$
(5.50)

This completes the design of  $\epsilon_1$ -estimator.

# 5.4 <u>Residual Dynamic Response to Control Power Spillover</u>

Consider the control-design model and the uncontrolled system dynamics

$$\dot{\mathbf{x}} = \mathbf{A}\mathbf{x} + \mathbf{B}\mathbf{F} \qquad \mathbf{F} = \mathbf{G}\mathbf{x} \tag{5.51}$$

$$\dot{\mathbf{x}}_{\mathrm{U}} = \mathbf{A}_{\mathrm{U}}\mathbf{x}_{\mathrm{U}} + \mathbf{B}_{\mathrm{U}}\mathbf{F} \tag{5.52}$$

The response of the closed loop control design model is

$$x = \phi_{CL}(t, t_0) x_0$$
,  $\phi(t) = e^{A_{CL}t}$  (5.53)

where  $\phi_{CL}$  is the closed-loop transition matrix for the control design model. Denoting the control period over which  $x_0$  is regulated by  $T=(t_f-t_0)$ , the spillover excitation of the uncontrolled modes can be measured by the total square response during T. The total squared modal displacement response of the uncontrolled dynamics will be given by

$$Y_{U}^{2} = \sum_{r=n+1}^{N} \int_{t_{o}}^{t} \xi_{ru}^{2} dt$$
 (5.54)

The uncontrolled modal displacements due to control spillover excitation are given by the convolution integrals

$$\xi_{ru} = \omega_{ru}^{-1} b_{ru}^{T} \int_{t_{o}}^{t} \sin \omega_{ru} (t-\tau) G x(\tau) d\tau$$
(5.55)

Substituting (5.55) and (5.53) into (5.54), we obtain

$$Y_{U}^{2} = x_{0}^{T} \left[ \int_{t_{0}}^{t} \phi_{CL}^{T}(\tau', t_{0}) B_{U}^{T} Q_{y}(t, \tau, \tau') B_{U} \phi_{CL}(\tau, t_{0}) d\tau d\tau' dt \right] x_{0}$$
(5.56)

where  $B_U$  is the input influence matrix of the truncated dynamics as given in Eq. (2.29). The time dependent weighting matrix  $Q_y$  is given by

$$Q_{y} = [\omega_{u}^{-2}] [S^{2}(t, \tau, \tau')]$$

$$[\omega_{u}^{-2}] = \text{diag} [\omega_{ru}^{-2}] \qquad r=n+1, \dots,$$

$$[S^{2}(t, \tau, \tau')] = \text{diag} [Sin\omega_{ru}(t-\tau')Sin\omega_{ru}(t-\tau)]$$
(5.57)

The weighting matrix  $Q_y$  obviously involves the cross correlations of the impulse response functions of the residual modes and is inversely proportional to the square of the uncontrolled natural frequencies. Equation (5.56) will then, noting that after integrations effectively an additional  $\omega_{ru}^{-2}$  will evolve so that the results will we inversely proportional to  $\omega_{ru}^{-4}$  yield

$$Y_{U}^{2} = x_{0}^{T} U_{y}(\omega_{u}^{-4}) x_{0}$$
 (5.58)

where  $U_y$  must be a positive definite resultant matrix of the weighted integration in Eq. (5.56) since  $Y_U^2$  is positive definite and the argument  $\omega_u^{-4}$  is shown to highlight the nature of dependency of the results on residual natural frequencies.

Next, considering the spillover control power  $S_U$  defined by Eq. (2.2g) and the Eqs. (5.51-5.53)

$$S_{U} = x_{O}^{T} \left[ \int_{t}^{t} \int_{t}^{t} \int_{t}^{t} \int_{t}^{t} \int_{t}^{t} \phi_{CL}^{T}(\tau', t_{o}) G^{T} B_{U}^{T} B_{U} G \phi_{CL}(\tau, t_{o}) d\tau d\tau' dt \right] x_{O}$$

$$(5.59)$$

which must yield

$$S_{U} = x_{0}^{T} P_{U}^{M} x_{o} \qquad P_{U}^{M} = P^{R} - P_{C}^{M}$$
 (5.60)

where the power matrices  $P^{R}$  and  $P_{C}^{M}$  are available as required for efficiency analysis. If the control law is derived via a dynamic compensator, that is if Eqs. (5.5) and (5.7) are used instead of (5.51), as described in Section 5.2, it is straight forward to see that the counterparts of Eqs. (5.58) and (5.59) are obtained by replacing  $x_{o}$  with  $\bar{x}_{o}$ ,  $\phi_{CL}$  with the  $\bar{\phi}_{CL}$ , the closed-loop transition matrix of Eq. (5.50) and G with  $\bar{G}$ . The weighting matrices  $B_{U}^{T}B_{U}$  and  $B_{U}^{T}Q_{y}B_{U}$ will remain the same. Consequently,  $\bar{P}_{U}-\bar{P}^{R}-\bar{P}_{C}^{M}$  will take the place of  $P_{U}^{M}$  in Eq. (5.60).

Hence, recognizing that the above derivation will remain the same quantitatively whether a compensator dynamics is included or not we return to Eq. (5.58), (5.59) for  $Y_U$  and  $S_U$ . Comparing  $Y_U$  with  $S_U$  we note that  $Y_U$  is a weighted version of the control power wasted in the uncontrolled modes where the weighting matrix is  $Q_y$  for  $Y_U^2$  and unity for  $S_U$ . Hence, the residual response and the wasted control power are correlated. The significance of this result is that a measure of residual dynamic response is available through the wasted power  $S_U$ , which is in turn available from the efficiency analysis directly without ever involving the residual dynamics in the computations. It follows that within the efficiency concepts presented in this investigation, evaluation model response simulations are not necessary to infer degradation of the control design model behavior.

Since  $Q_y$  is inversely proportional to the fourth power of uncontrolled natural frequencies, it follows that if natural frequencies larger than unity are in the uncontrolled dynamic model the net effect will be that the matrix  $U_y$  in Eq. (5.58) will be considerably smaller than the control power matrix  $P_y^{H}$ . Therefore, it may take considerable control power spillover or considerable inefficiency in the control design model before any significant degradation of response of control design model due residual excitation can be detected. This is certainly not surprising as it is well known that high frequency modes are difficult to excite, hence, considerable control power is bled by those modes even for minimal excitation of them if the structure control design is not efficient. To put it more bluntly, it is more critical to align the controller directions with the structural modes in the design model to obtain high efficiencies if higher modes are not in the control design model.

The above analysis proves the assertion made in Section 2.8 that performance of the structural-control system must not be judged on response studies alone. Although no residual dynamics response can be detected in the system outputs, the controller can still be wasting control power operating inefficiently. In space missions, ultimately, the power used inefficiently by the control system has to be generated on board the spacecraft and therefore, has the potential to impact the design of power subsystem and other subsystems and even the mission significantly. The efficiency results presented up to this point, will establish a most important interdisciplinary link of controller design to other important disciplines of space systems design. A good structural control design must therefore, strive for an efficient controller which can be accomplished only after an efficiency analysis is carried out.

Furthermore, if an inefficient controller is used to control a model with high frequencies in the design model which will require considerable control power and less than unity low frequency modes are in the uncontrolled set, it would take only a small amount of inefficiency to produce considerable residual mode excitation and degradation of the system response. In such cases design of efficient controllers will become even a more critical task.

### 6.0 OTHER APPLICATIONS OF CONTROLLER EFFICIENCY & APPLICATION TO CSDL-ACOSS-6

## 6.1 <u>Introduction</u>

In this chapter, we present some illustrations of the variety of structurecontrol system designs that can be addressed via the efficiency concepts. Specifically, in following the subject of Section 5.4 on "Residual Dynamic Response to Control Power Spillover," we give a vivid illustration of the performance of an inefficient controller in Section 6.2.

Another important topic in the design of structure-control system is to determine a good input configuration, that is how many actuators to use and where to put them, as well as quantifying the effectiveness of actuators. In Section 6.3, we give illustrations of how efficiencies can be used to select the input configuration.

Another application of the efficiency concept can be to choose the weighting matrices (or parameters) in the application of the LQR theory. Because the efficiencies are defined regardless of the particular control theory used in the design of the control law, the opportunity exists that the LQR weighting matrices can be chosen to improve the controller efficiency. This is done in Section 6.4.

As a final analysis and design study by using the efficiency technique, in Section 6.5 we present results of efficiency modes analysis of a representative large flexible spacecraft configuration, the Charles Stark Draper laboratory ACOSS-6 (Model 2) structure.

## 6.2 <u>Inefficient Control with Truncated Low Frequency Modes</u>

It was shown in Section 5.4, that the total square modal response of truncated dynamics of a system was a weighted version of the control power spillover and the weighting was inversely proportional to the fourth power of the natural frequencies of the truncated modes. Hence, the efficiency or inefficiency of a controller becomes critical with low frequency truncated modes, also from a response point of view.

As an illustration of this, we refer the reader to Examples 3.1 and 3.3 with respective simulations given in figures SIM39 (Figures 3.1, 3.2 and 3.3) and SIM41 (Figures 3.4, 3.5 and 3.6), in Chapter 3. Both of these controllers had low efficiencies, e\*=0.173%, e=4.31% and e\*=6.2%, e=28.59%. The control design model was 4-th order and consisted of high frequency modes (#11 and 12), the lowest frequencies were left in the truncated dynamics. Note that the degradation in the response will be as serious as the magnitude of the lowest frequency in the truncated dynamics. For the ACOSS-4 structure the lowest natural frequency is  $\omega_1$ =1.34 rad/sec., so it too has an attenuating effect on the control power spillover (proportional to  $1/\omega_1^4$ ), nevertheless, not as strong as the other truncated modes. Certainly if  $\omega_1$ <1, the control power spillover effect would be magnified in the residual response. However,  $\omega_1$ =1.34 was still low enough to accentuate the residual response degradation due to an inefficient controller. In both SIM39 (Figures 3.1-3.3) and SIM41 (Figures 3.4-3.6) evaluation model responses are significantly degraded.

Two other examples of evaluation model response degradation due to inefficient controllers are presented below:

Example 6.1: n=4 (Modes: 9-12), m=2(#1,#2). The system has e\*=8.39%, e=45.67%. SIM28 shows the responses (Figures 6.1 through 6.3). Example 6.2: n=4 {Modes: 9-12}, m=1 (#5). The system has e\*=0.202%, e=21.33%. SIM30 shows the responses (Figures 6.4 through 6.6). Clearly the more inefficient system of Example 6.2 has serious degradation of response.

### 6.3 Actuator Selection Via Efficiencies

The full-order 12 mode ACOSS-4 evaluation model was considered in conjunction with a full set of 12 actuators, n=12, m=12. Because there is no model truncation, the relative model efficiency of the designs are e=100%. However, e\* is not necessarily 100% since the 12 inputs do not represent a continuously distributed input field although they may come close to representing it. The number of inputs were decreased from m=12 to m=1, in each case e\* was computed, the changes in e\* with the deletion of each input being a reflection of the effectiveness of that input. The following results were obtained:

ACTUATOR EFFECTIVENESS ON ACOSS-4 EVALUATION MODEL

<u>m=Actuator Number</u>	<pre>%e*Global efficiencies</pre>	<u>∆e</u> *	<u>Actuator Rank</u>
12	70.88	-0.59	12
11	71.47	0.28	11
10	71.19	7.13	3
9	64.06	6.74	4
8	57.32	13.85	2
7	43.47	31.73	1
6	11.74	0.7	10
5	11.04	0.78	8
4	10.26	0.72	9
3	9.54	1.16	7
2	8.38	4.10	6
1	4.28	4.28	5

First, note that e\* is valid indicator of the quality of the input configuration and the effectiveness of an actuator can be measured by its contribution to e\*. Hence, each actuator was ranked according to its effectiveness. The most effective 5 actuators were (#7,8,10,9,1). For this model, the above ranking of the actuators is in agreement with the actuator ranking by computing the work done by each actuator during the control period (Ref. 16). A second study was done by using a reduced-order model n=1-8. In this case, since e is less than 100%, it too was considered in judging the actuator effectiveness along with the global efficiency\*.

_ <u>m_</u>	<u> </u>	<u> 8 e</u>	<u>Rank*</u>	_ <u>∆</u> e*%_	∆e%	Rank
1	5.69	56.92	8	5.69	56.92	1
2	8.41	55.26	12	2.72	-1.66	10
3	25.79	75.67	1	17.38	20.41	2
4	30,76	80.65	9	4.97	4.98	5
5	45.30	70.62	2	14.54	-10.03	12
6	53.09	60.84	5	7.79	-9.78	11
7	60.48	76.66	6	7.39	15.82	3
8	67.85	84.21	7	7.37	7.55	4
9	72.40	83.47	10	4.55	-0.74	9
10	81.19	87.29	3	8.79	3.82	6
11	89.36	87.58	4	8.17	0.29	7
12	93.08	87.28	11	3.72	-0.30	8

ACTUATOR EFFECTIVENESS ON ACOSS-4 WITH A REDUCED-ORDER MODEL

Again, note that e\* is a consistent indicator of actuator effectiveness whereas e is anomalous, both rankings are different and in this case because the control design model changed, ranking of actuators according to e\* has also changed from the previous example set when the evaluation model was considered. Simulation results of n=8, m=1; and n=8, m=12 are shown in figures labeled SIM8 and SIM18 (Figures 6.7 through 6.12).

### 6.4 Effect of LOR Weighting Matrices on Efficiencies

The global and model efficiencies were computed for different state and control weighting parameters q and r as discussed in Section 2.8, for the weighting matrices  $W_x$ =block-diag [q],  $W_c$ =r[1] for a given control design model and input configuration, n=8 (natural) and m=1. The results are shown below:

<u> </u>	<u> </u>	<u></u>	<u></u>
1.0	$\omega_{\rm r}^2$ , 1	5.69	56.93
0.1	$\omega_r^{\bar{2}}, 1$	12.12	56.93
<sup>^</sup> .05	$\omega_r^{\bar{2}}, 1$	16.89	56.93
J.01	$\omega_r^2$ , 1	27.29	56.93
1	ì	4.99	56.93
1	0.5	4.99	56.93
1	0.1	4.99	56.93
1	10	5.26	56.93
1	100	7.81	56.93
1	1000	31.86	56.93

#### EFFECT OF LOR WEIGHTING PARAMETERS

SIM20 and SIM26 (Figures 6.13 through 6.18) show the responses for r=0.05 and q=100, respectively. The above example shows that the model efficiency e is insensitive to control and state ghting, which might be expected since e is the ratio of two control powers  $S^M$ ,  $S^R$  for the same control design, hence, changing q and r changes both by the same proportion. On the other hand, e\* is based on two dynamically similar control designs but with different input profiles hence, e\* should be sensitive to changes in q and r.

These examples are based on a rather simple variation of the LQR weighting matrices. The effect of changes in the full weighting matrices, for example, in the case of different frequency shaped weighting matrices, may produce not so simple trends. In the above examples, the insensitivity of the model efficiency coefficient to weighting matrices can be used to advantage in the sense that it

will be possible to adjust the response time of the system by changing q and r without changing (a good) model efficiency.

# 6.5 <u>Application to a Large Space Structure: CSDL-ACOSS-6 (Model 2)</u>

Next, a realistic large space structure was considered. The structure shown in Fig. (6.19) is the Charles Stark Draper Laboratory model known as ACOSS-6 (Model 2) (ref. 17). The finite Element evaluation model that was considered was produced by NASTRAN at the Flight Dynamics Laboratory at WRDC/WPAFB. The model that was studied for efficiency had 294 degrees of freedom. Efficiency of two different control design models employing different sets of inputs were considered. There were available 21 actuators located at positions shown in reference 18. Efficiency of two designs were analyzed.

- Example 6.3: In this case, the control design model had 8 modes, n=(12, 13, 17, 21, 22, 24, 28, 30) with 4 inputs m=(18, 19, 20, 21). Actuator numbers are in consistence with the designation listed in Ref. (18). The modes chosen were designated significant for LOS studies (ref. 19). The simulation results of square x-y plane deflection at node 37 are shown in figures labeled SIMV1 (Figures 6.20 and 6.21). The global efficiency was e\*=0.176% and the relative model efficiency was e=0.657%.
- Example 6.4: The control design model had 10 modes  $n=\{7-16\}$  and 10 inputs  $m=\{1-10\}$ . The same deflection as in example 6.3 is shown in figures labeled SIMV2 (Figures 6.22 and 6.23). The efficiencies of this system were  $e^{*}=0.236$ % and e=0.401%.

Because the first six modes obtained from the structural eigenvalue problem were not reliable none of them were taken in the control design models in both of the examples. Such poorly known modes have no consequences for the reliability of efficiency analysis as long as they are in the truncated dynamics of the system as discussed in Section 2.6. The computer results of efficiency modes analysis are given in Appendix C.

In examples 6.3 and 6.4 for the ACOSS-6 structure the responses of the evaluation model were not shown since this would require simulation of a 588 th order state model, clearly beyond any reasonable expectation. However, in light of the illustrations and theoretical developments presented thus far, the controls designed for the ACOSS-6 structure are extremely inefficient. Almost 99.5% of control power will excite the truncated dynamics. To make this ineffieicney even more critical, there are some modes with natural frequencies less than unity left in the truncated dynamics. Hence, as demonstrated in Example 6.2 for the ACOSS-4 structure, in this case for ACOSS-6 too, serious degradation of the responses will occur in the actual (evaluation model) structure instead of the control design model responses shown in Figures (6.21) and 6.23). However, no simulation of the evaluation model is necessary based on the qualitative conclusion that the efficiency approach affords. (It is understood, however, that a simulatable model larger than the control design model and smaller than the 588-th order evaluation model can be used to underline these statements. The challenge here is to resist this temptation).

The efficiency obtained above for a realistic large space structure are extremely low indeed showing the need and demonstrating the merits of a fullscale assessment of the need for an efficiency analysis. As shown in Appendix A, the efficiencies defined in this research are literally the power efficiencies

of the control system. A 0.6% model efficiency simply means 6 energy units/sec. out of 1000 were consumed by the controller, the rest simply being wasted. The wasted power will have been derived on board the spacecraft to serve no useful control purpose. Such inefficiencies in the system cannot be tolerated. It should be recalled that although the control design model is a much smaller model, the information that is obtained from them in the form of power efficiency pertains to the behavior of the full-order evaluation model.



Figure 6.1: Globally optimal system LOS, n=4 (modes: 9-12), example 6.1; response of an inefficient system e\*=8.39%, e=45.67%



Figure 6.2: Suboptimal system LOS, n=4 (modes: 9-12) m=2 (#1,2), example 6.1; response of an inefficent system e\*=8.39%, e=45.67%



Figure 6.3: Evaluation for Model LOS, n=4 (modes: 9-12), m=2 (#1-2), example 6.1; response of an inefficient system e\*=8.39%, e=45.67%



Figure 6.4: Globally optimal system LOS, n=4 (modes: 9-12), example 6.2; response of an inefficient system e\*=0.2%, e=21.33%


Figure 6.5: Suboptimal system LOS, n=4 (modes: 9-12), m=1 (#5), example 6.2; response of an inefficient system, e\*=0.2%, e=21.33%



Figure 6.6: Evaluation model LOS, n=4 (modes: 9-12) m=1 (#5), example 6.2; response of an inefficient system, e\*=0.2%, e=21.33%



Figure 6.7: Globally optimal system LOS, n=8 (modes: 1-8) m=1 (#1) for actuator selection via efficiencies





Figure 6.8: Suboptimal system LOS, n=8 (modes: 1-8), m=1 ( 1) for actuator selection via efficiencies



Figure 6.9: Evaluation model LOS, n-8 (modes: 1-8), m-1 ( 1) for actuator selection via efficiencies



Figure 6.10: Globally optimal system LOS, n=8 (modes: 1-8), for actuator selection via efficiencies







Figure 6.12: Evaluation model LOS, n=8 (modes: 1-8), m=12; for actuator selection via efficiencies



Figure 6.13: Globally optimal system LOS, n=8 (modes: 1-8), effect of LQR weighting on efficiency, r=0.05, e\*=16.89%, e=56.93%



Figure 6.14: Suboptimal system LOS, n=8 (modes: 1-8), m=1; effect of LQR weighting on efficiency r=0.05, e\*=16.89%, e=56.93%



Figure 6.15: Evaluation model LOS, n=8 (modes: 1-8), m=1; effect of LQR weighting on efficiency r=0.05, e\*=16.89%, e=56.93%



Figure 6.16: Globally optimal system LOS, n=8 (modes: 1-8); effect of LQR weighting on efficiency, q=100, e\*=7.81%, e=56.93%



Figure 6.17: Suboptimal system LOS, n=8 (modes: 1-8); effect of LQR weighting on efficiency, q=100, e\*=7.81%, e=56.93%



Figure 6.18: Evaluation model LOS, n=8 (modes: 1-8); effect of LQR weighting on efficiency, q=100, e\*=7.81%, e=56.93%



Figure 6.19: CSDL - ACOSS - 6 (model 2) structure



Figure 6.20: Globally optimal system, square x-y plane deflection at node 37, n-8, (modes: 12,13,17,21,22,24,28,30), example 6.3



Figure 6.21: Suboptimal system, square x-y plane deflection at node 37, n=8, (modes: 12,13,17,21,22,24,28,30), m=4 (#18-21), example 6.3



Figure 6.22: Globally optimal system, square x-y plane deflection at node 37, n=10, (modes: 7-16), example 6.4



Figure 6.23: Suboptimal system, square x-y plane deflection at node 37, n=10, {modes: 7-16}, m=10 (#1-10), example 6.4

### 7.0 CONCLUSIONS

The concept of efficiency of an engineering system is a time proven design and analysis tool in many disciplines. Structure-Control Systems should not pose an exception to this practice. This research undertook the task of defining an efficiency concept for structure-control systems, and established physical grounds for the efficiencies which would have the potential to address the issues in the control of large structural dynamic systems. We have shown that the global and relative model efficiencies proposed and developed in this research are the power efficiencies of the control system and therefore, they must be considered as fundamental figures of merit of a control system design along with other more familiar performance measures. No less important a feature of the efficiencies is that the performance of the evaluation model can be judged on the basis of the control design model alone without explicitly involving the fullorder evaluation model, a feat made possible only by a unique judicious choice of the weighting matrices involved in the efficiency definitions.

The investigation developed the new concept of controller modes signifying characteristic avenues of efficient control power utilization. These efficiency modes are complementary to the familiar concept of structural modes and together they determine the effectiveness of any structural control design. Ultimately what counts is the degree of matching of these two sets of modes. The efficiency modes analysis leading to the recognition of principal controller directions led to a procedure to give structure to the internal workings of the structural control system through the control power matrices. We have attempted to exploit this internal characterization of the system by defining component efficiencies and efficiency components. Furthermore, we proposed and developed model/controller reduction approaches based on that structural information by correlating the structural modes and the controller (efficiency) modes. We have illustrated the significance of the efficiency concepts by a host of examples on realistic structures. The end point of this investigation offers many other avenues of research. To mention only two: optimization of structural control system for efficiency subjected to other design constraints, design of control gains by a direct approach utilizing the system efficiency as the target design criterion. We have demonstrated that the Linear Quadratic Regulator Theory can produce highly inefficient control designs although they are regarded as optimal. However, the optimality of an LQR design is subjective and theory merely serves as a means of obtaining control gains in some sense. On the other hand, the definitions of power efficiencies of the control system constitute absolute nondimensional unique performance measures. No subjectivity is involved in their definitions. The efficiency of all control systems regardless of the methodology by which they are designed can be analyzed by the approaches developed in this research and their merits can be established on a <u>common</u> ground.

### REFERENCES

- Oz, H., "Dynamically Similar Control Systems and a Globally Optimal Minimum Gain Control Technique: IMSC," <u>Journal of Optimization Theory and</u> <u>Applications</u>, Vol. 59, No. 2, 1988, pp. 183-207.
- Meirovitch, L., and Silverberg, L., "Globally Optimal Control of Self-Adjoint Distributed Systems," <u>Optimal Control Applications and Methods</u>, Vol. 4, pp. 365-386, 1983.
- 3. Oz, H., "Robust Approximations for Closed-loop Distributed Parameter Structures," The Ohio State University, Aeronautical and Astronautical Engineering Research Report on Dynamics and Control, No: AAE-RR-DC-102-1988.
- Meirovitch, L., Baruh, H., and Oz, H., "A Comparison of Control Techniques for Large Flexible Systems," <u>Journal of Guidance, Control and Dynamics</u>, Vol. 6, No. 4, 1983, pp. 302-310.
- 5. Oz, H., "A New Concept of Optimality for Control of Flexible Structures," The Ohio State University, Aeronautical and Astronautical Engineering Research Report on Dynamics and Control, No: AAE-RR-DC-101-1988.
- 6. Oz, H., and Adiguzel, E., "Generalized Natural Performance Charts for Control of Flexible Systems," AIAA-84-1951-CP AIAA Guidance and Control Conference, Aug. 1984, Seattle, WA.
- Yousuff, A., and Skelton, R., "Controller Reduction by Component Cost Analysis," <u>IEEE Trans. on Auto. Control</u>, Vol. AC-29, No. 6, 1984, pp. 520-530.
- Oz, H., "Efficient Controller Reduction for Structure Control," NASA/DOD Control/Structures Interaction Technology 1987, AFWAL-TR-88-3052, pp. 136-144.
- 9. Hill, P.G., and Peterson, C.R., <u>Mechanics and Thermodynamics of</u> <u>Propulsion</u>, Addison-Wesley Publishing Company, Reading, MA, 1965.
- RADC-TR 80-78, Vol I, ACOSS Four (Active Control of Space Structures) Theory, The Charles Stark Draper Laboratory, Inc., Cambridge, MA, April 1980.
- 11. Skelton, R.E., Dynamics Systems Control, John Wiley & Sons, Inc. 1988.
- 12. Yedavalli, R.K., "Time Domain Robust Control Design for Linear Quadratic Regulators by Perturbation Analysis," Proceedings IFAC Workshop on Model Error Concepts and Compensation, Boston, MA, June 1985, pp. 914-919.
- Yedavalli, R.K., "Perturbation Bounds for Robust Stability in Linear State Space Models," <u>International Journal of Control</u>, Vol. 42, 1985, pp. 1507-1517.

- Hyland, D.C., Bernstein, D.S., "The Optimal Projection Equations for Fixed-Order Dynamic Compensation," <u>IEEE Transactions on Automatic Control</u>, Vol. AC-29, No. 11, 1984, pp. 1034-1037.
- Oz, H., Meirovitch, L., "Modal-Space Control of Distributed Gyroscopic Systems," <u>Journal of Guidance and Control</u>, Vol. 3, No. 2, 1978, pp. 140-150.
- 16. Khot, N.S., "Personal Communications," 1988-89.
- 17. RADC-TR-80-377, ACOSS Six (Active Control of Space Structures), The Charles Stark Draper Laboratory, Inc., Cambridge, MA, Jan. 1981.
- Thyfault, D. "Decentralized Control of a Large Space Structure Using Direct Output Feedback", M.S. Thesis, Department of Aeronautical Engineering, Air Force Institute of Technology, Dec. 1983.
- 19. RADC-TR-82-21, ACOSS Five (Active Control of Space Structures) Phase IA., Lockheed Missiles and Space Company, Inc., March 1982.
- 20. Özgüner, Ü., Breitfeller, E., "Development of a Control Oriented Model of a Cantilevered Beam with End-Mass," Proceedings of the American Control Conference, Atlanta, GA, 1988.

# APPENDIX A: PROOF OF FUNCTIONAL SR AS CONTROL POWER

The control evaluation functionals  $S^R$ ,  $S^M$  and  $S^*$  which are uniquely defined for the efficiency analysis of structure-control systems represent power quantities, true rate of change of energy, a fundamental concept in the design of engineering systems. To see this, consider the starting point of this investigation in Chapter 2 which is the control functional  $S^R$  defined over the DPS:

$$S^{R} = \int f^{T}(p,t)m^{-1}(p)f(p,t)dD(p)dt \qquad (A-1)$$

corresponding to the partial differential equations of motion.

$$m(p)\ddot{u}(p,t) + L(up,t) = f(p,t)$$
 (A-2)

First, perform a dimensional analysis. In Eq. (A-2) m(p) has the units of mass/dimD where dimD is the dimension of Domain of the Structure, mass is the mass or inertia distribution over D. For one-dimensional beam elements dimD=length, for plate elements dimD is an area etc... Consistently, the input function f(p,t) represents a force density over the domain D, it has the dimension Load/dimD where load can be a force or a moment. Substituting the dimensional equivalents of the terms appearing in the definition (A-1)

$$f(p,t) = \frac{Load}{dimD}$$
,  $m(p) = \frac{mass}{dimD}$ 

$$S^{R} = \int \frac{Load}{dimD} \frac{dimD}{mass} \frac{Load}{dimD} dimD \star d(time)$$

$$S^{R} = \frac{\text{Load}^{2} * \text{Time}}{\text{mass}}$$
(A-3)

and noting

time<sup>2</sup>

Expression (A-3) yields after simplification

$$S^{R} = \frac{\text{Load } * \text{ displacement}}{\text{Time}} = \frac{\text{Energy}}{\text{Fime}} = \text{Power}$$
(A-4)

Since  $S^M$ ,  $S^*$  are also derived from  $S^R$  by an invariant frame transformation,  $S^M$  and  $S^*$  are also power quantities.

Next, we consider the mechanical aspects of this power. To make this easier to see, consider the one-dimensional mechanical analogy of the DPS equation of motion (A-2), the familiar spring mass system where the spatial dimension has been (integrated) out.

$$m\ddot{u} + ky = f(t)$$
,  $\frac{k}{m} = \omega^2 = rad/sec^2$  (A-5)

for which  $S^{R}$  is

$$S^{R} = \int m^{-1} f^{2}(t) dt \tag{A-6}$$

Multiply Eq. (A-5) by  $m^{-1}f(t)$  and integrate over time, and recognize that the right hand side becomes  $S^{R}$ :

$$\int f \ddot{u} dt + \int \omega^2 f u dt - S^R$$
(A-7)

substituting  $\ddot{u} = d\dot{u}/dt$ 

$$\int f d\dot{u} + \int \omega^2 f u dt = S^R \tag{A-8}$$

Again the left hand side has power units and specifically the first integral on the left is the area under the load versus velocity diagram. Alternately, write the equation of motion (A-5)

$$m(\ddot{u}+\omega^2 u) = f(t)$$
 (A-9)

and define the <u>effective rigid body acceleration</u> that the forcing function f(t)imparts to the mass m as  $\ddot{u}_R = \ddot{u} + \omega^2 u$  where  $u_R$  is the effective rigid body displacement due to f(t). Hence, (A-5) is equivalent to

$$mu_{R} = f(t)$$

Multiply this last expression by  $\dot{\textbf{u}}_{R}$ 

dt

$$mu_{R}\dot{u}_{R} = f(t)\dot{u}_{R}$$
(A-10)
$$\frac{d}{t_{2}} m\dot{u}_{R}^{2} = f\dot{u}_{R}$$
(A-11)

from which we recognize the effective rigid body kinetic energy

$$T^{R} = \frac{1}{2}m\dot{u}_{P}^{2}$$

and Eq. (A-10) is the more familiar power expression

$$\frac{d}{dt} = f\dot{u}_{R} = f\dot{u}_{R} = P$$
(A-12)

The incremental power change associate with Eq. (A-12) is

$$fdu_{R} = dP$$

The effective rigid body incremental velocity  $d\hat{u}_R$  can be written as

$$d\dot{u}_{R} = (u + \omega^{2}u)dt$$
 (A-13)

Multiplying on the left by f and recognizing (A-12)) we obtain

$$\int f(\ddot{u} + \omega^2 u) dt = \int f d\dot{u}_R = \int dP = P$$

where P is the total mechanical power. The integral on the left is  $S^R$  by virtue of Eq. (A-8), hence, we get

$$S^{R} = \int m^{-1} f^{2} dt = \int f d\ddot{u}_{R} = P$$

We have thus also shown mechanically that  $S^R$  functional is a total power expression associated with the control input f in trying to move the structure with an effective rigid body velocity  $U_R$ . It must be recognized that  $u_R$  is not a fictitious mathematical expression. Both  $\ddot{u}_R$  and  $\dot{u}_R$  as derivatives are well defined quantities at all times in terms of u as given by Eq. (A-13). The concepts of effective rigid body velocity/acceleration are also plausible because this is what the controller attempts to impart to the structural mass via the thrust of control input f. The controller actuators at any instant will not recognize that the mass has flexibility. The effect of flexibility is inherent in the structure defined by the domain of m, and it manifests itself to result in an actual acceleration  $\ddot{u}$  by extracting  $\omega^2 u$  from  $\ddot{u}_R$ 

$$\ddot{u}_{R} - \omega^{2} u = \ddot{u}$$
 (A-14)

which provides the mechanism to bleed off control power into the flexible dynamics to store it as elastic energy in the structure.

Next, we consider some specific control hardware.

# Thruster Jets

For a thruster using an exhaust jet, the thrust force f(t) is the control input and given by

$$f = \hat{m}_{p}(t) V_{ex}$$
 (A-15)

where  $V_{ex}$  is the exhaust jet velocity and  $\mathfrak{m}(t)$  is rate of flow of propellant mass  $m_{p}$ . Form the control functional  $S^R$  for such a thruster

$$S^{R} = \int m^{-1} f^{2}(t) dt \qquad (A-16)$$

and substituting (A-15) into (A-16) by assuming constant exhaust velocity

$$S^{R} = \frac{V_{ex}^{2}}{m} \int \tilde{m}_{p}^{2}(t) dt \qquad (A-17)$$

in which the propellant flow rate is the control variable. For simplicity, the change in the total mass m due to propellant flow was neglected. The kinetic energy rate of flow carried away by the exhaust jet is on the other hand

$$\mathbf{\hat{E}}_{p} = \frac{1}{2} - - - - - \left( \mathbf{m}_{p} \mathbf{V}_{ex}^{2} \right) = \frac{1}{2} \mathbf{V}_{ex}^{2} \mathbf{\hat{m}}_{p}(t) = \mathbf{P}(t)$$
(A-18)  
dt

where P is the power of the exhaust jet at time t. Substituting from (A-18) into (A-17) for  $\dot{m}_p$  we get

$$S^{R} = -\frac{4}{mV_{ex}^{2}} \int P^{2}(t) dt \qquad (A-19)$$

Denote the root mean square power in the exhaust beam by  $P_{\mbox{\tiny rms}}$  and write

$$P_{rme}^2 T = \int P^2(t) dt$$

where T is the control (regulation) time. Next, introduce an average characteristic power  $P_c$  for the combination of the structure to be controlled and the nature of the thruster propellant

$$P_{c} = \frac{mg_{o}^{2}I_{sp}^{2}}{2T}$$
 (A-20)

where  $I_{sp}$  is the specific impulse of the thruster propellant and  $g_o$  is the gravitational acceleration on earth. Now the S<sup>R</sup> can be rewritten as

$$S^{R} = ( \cdots ) P_{rms} \quad or \quad S^{R} = ( \cdots ) P_{rms} \quad (A-21)$$

$$P_{r} \qquad \qquad W_{0} g_{0} I_{cr}^{2}$$

in which  $W_o$  is the on earth weight of the structure.

Equations (A-21) clearly shows that  $S^R$  is proportional to the root mean square power of the exhaust beam rower with a nondimensional proportionality constant which provides an interesting combination of control, structure and thruster propellant parameters.

### Proof-Mass Actuators

As another controller example, we consider a proof-mass actuator of the type discussed in Ref. (20). The actuator dynamics given in Ref. (20) accounts for coupling with the structural dynamics and typically represents a compensator dynamics, a subject which was discussed in Section 5.2. For simplicity of this discussion, we shall disregard the interaction of the actuator with the structure and represent the force f(t) it exerts on the structure by the equation

$$f(t) = K_{amp}K_fV(t)$$
 (A-22)

where V(t) is the control voltage,  $K_f$  is a force coefficient (force/amper). The  $K_{amp}$  is given in terms of the armature current and voltage:

$$K_{amp}^{(s)} = I_{arm}^{(s)}/V(s) = \frac{1}{Ls + R}$$
 (A-23)

s is the LaPlace variable, L and R are the inductance and resistance of the actuator. If the time constant L/R is small, then  $K_{amp} = R^{-1}$ . In this case the actuator force output is simply given by

$$f(t) = K_f I_{arm}^{(t)} = K_f R^{-1} V(t)$$
 (A-24)

Forming  $S^{R}$ 

$$S^{R} = \int m^{-1}t^{2}(t) dt = \int m^{-1}K_{t}^{2}R^{-2}V^{2}dt = \int m^{-1}K_{t}^{2}l^{2}dt$$

Multiplying and dividing by R and recognizing that  $1^2$ R is the electrical power P consumed by the actuator to control the structure we write,

$$S^{R} = \frac{K_{f}^{2}}{mR} \int RI^{2}dt = \frac{K_{f}^{2}T}{mR}$$

for which we defined the mean power  $P_m = T^{-1} \int RI^2 dt$  representing the average electrical power of the proof-mass actuator over the control period. The coefficient of  $P_m$  can also be shown to be a nondimensional quantity relating the control, structure and actuator parameters as in the case of thrust jets discussed earlier.

The quantities  $S^M$  and  $S^*$  by definition have the same units as  $S^R$  and therefore, the efficiency quotients defined e, e\* in this report are physically the power efficiencies of the controller.

APPENDIX B: CLOSED FORM SOLUTION FOR S\*

The S\* is the control power for the continuously distributed input field over DPS where the modal inputs are based in the independent modal-space control approach (Regs. 1, 4). The modal state-space dynamics for the 2n-th order design model is:

$$\dot{\mathbf{x}}_{\mathbf{r}} = \begin{bmatrix} 0 & 1 \\ -\omega_{\mathbf{r}}^{2} & 0 \end{bmatrix} \mathbf{x}_{\mathbf{r}} + \begin{bmatrix} 0 \\ 1 \end{bmatrix} \mathbf{f}_{\mathbf{r}}^{*}(\mathbf{t}) \qquad \mathbf{r} = 1, 2, \dots, n \qquad (B-1)$$

with the modal states  $x_r = [\xi_i \xi_r]^T$ , and  $f_r^*(t)$  is the modal input given by

$$f_{r}^{*}(t) = g_{r1\xi r}^{*} + g_{r2}\xi_{r}$$

$$g_{r1}^{*} = \omega_{r}^{2} - \rho_{r1}\rho_{r2} , \quad g_{r2}^{*} = -(\rho_{r1}\rho_{r2})$$

 $\rho_{r1}$  and  $\rho_{r2}$  are the r-th closed-loop modal eigenvalues corresponding to the openloop structural eigenvalues  $\pm j\omega_r$ .

The control power S\* as defined in Chapter 2 can now be written as

$$S* = \int f_{r}^{*2}(t) dt = \sum_{r=1}^{n} x_{ro}^{T} P_{r}^{*} x_{ro}$$

 $P_r^*$  is the 2x2 optimal control power obtained as the solution of n 2x2 independent Lyapunov equations

$$A_{r}^{T}P_{r}^{*} + P_{r}^{*}A_{r} + G_{r}^{*T}G_{r}^{*} = 0 \qquad r=1,...,n \qquad (B-2)$$

$$A_{r} = \begin{bmatrix} 0 & 1 \\ -\omega_{r}^{2} & 0 \end{bmatrix} , \quad G_{r}^{*} = [g_{r1}^{*} g_{r2}^{*}]$$

The solution of these modal Lyapunov equations is easily obtained as

$$P_{r}^{\star} = \begin{bmatrix} P_{1}^{\star} & P_{0}^{\star} \\ P_{0}^{\star} & P_{2}^{\star} \end{bmatrix}$$
(R-3)

$$P_{1}^{*} = \frac{\omega g_{1}^{2} g_{2}^{2}}{2(g_{1} - \omega^{2})} + \frac{g_{2}(g_{1} - \omega^{2})}{2} - \frac{g_{1}^{2}}{2g_{2}} - g_{1}g_{2} \qquad (B-4)$$

$$P_0^* = -\frac{g_1^2}{2(g_1 - \omega^2)}$$
(B-5)

$$P_2^{\bullet} = \frac{g_1^2}{2g_2(g_1 - \omega^2)} - \frac{g_2}{2}$$
(B-6)

in which  $g_1 = g_{r1}^*$ ,  $g_2 = g_{r2}^*$ ,  $\omega = \omega_r$ , r = 1, 2, ..., n.

APPENDIX C: COMPUTER RESULTS OF EXAMPLES ILLUSTRATED IN SECTIONS 3, 4 & 6

# Nomenclature for the computer outputs

The following consists of description of terminology encountered in the computer program that may not be self explanatory. Some output statements will not be of any interest to the reader as they are only for program control and debugging purposes. Hence, no explanation is given for them.

- Reduced principal efficiency components NP is the number of efficiency states considered, np=1,2,...,NP.
- \* Structural natural frequencies for the evaluation model are ordered in increasing order. The first n are in the control design model in increasing order, the rest are in the truncated model in increasing order. NE is the evaluation model order. N is the controlled structural modes.
- \* Init. Gen. Displacement and Init. Gen. Velocity are  $q(t_0)$  and  $\dot{q}(t_0)$  for the evaluation model.
- \* S\* is S\* , SM is  $S_c^M$  and SR is  $S_c^R$ .
- \* Principal Components Analysis of \_\_\_\_\_ efficiency is \_\_\_\_\_ efficiency modes analysis.

166

- \* PO is the  $n_p$  dimensional efficiency states initial conditions  $\epsilon$ . CE is squared efficiency coefficients  $c_i^2$ . i=1,2,...,NP, Eq. (3.36).
- \* Characteristic efficiencies are  $\lambda_i^e$  in random order. Efficiency vectors are the efficiency components  $e_i$ , total efficiency is the efficiency (not in percent).
- \* Eff... Num... Control Cost is  $x_0^T P_N x_0$ Eff... Denom... Control Cost is  $x_0^T P_D x_0$ Numerator/Denominator Component cost is  $(X_0 P_N)_{jj}$  j=1,2,...,n for each structural mode.
- \* Unordered...Comp Eff (Num. comp. Costs/Denom) is  $(X_0P_N)_{ii}/x_0^TP_Dx_0$ .
- \* Unordered Comp. Efficiencies is Equation (4.16).
- \* Method 1 is the "coupled efficiencies" cf. Equation (4.16) of structural degrees of freedom.
- \* Method 2 is the "decoupled efficiencies" of Equation (4.17) of structural degrees of freedom.
- Mode (in modsel) gives in decreasing order coupled efficiencies e<sub>j</sub> for structural modes. In modsel implies that the modes are designated as labeled in selected modes for control design. For example if 14th

and 31st modes are selected for control in modsel they are designated as the first and 2nd selected modes, respectively.

- NPth order efficiency (ENP) indicates that efficiencies are computed
   via Eq. (4.20) (4.22) by considering NP efficiency components.
- \* Block-diag numerator and denominator Comp. costs are side by side  $n_j$ and  $d_j$  in Eq. (4.17). Each row corresponds to a j-th structural mode.
- \* Block-diag. eff...Num/Denom...Control cost is the numerator/denominator of Eq. (4.19). Unordered comp. efficiencies is Eq. (4.17) for each j.
- \* Block-diag..Comp..Eff.. is e<sub>j</sub> in Eq. (4.17) in decreasing order for the modes in selected modes.
- \* NPth order block-diag. efficiency ENP is Eq. (4.19).

# EXAMPLES 3.1 & 3.4

```
-1
                                                                                                                                                                     ARE DESIRED, DIMERWISE A DUWAY PARAMETER) =
                                                                                                                     4
                                                                                                                   = d N
                                                                                                                                                                                                                                                                                                            STRUCTURAL MATURAL FREQUENCIES FOR THE EVALUATION MODEL
                                                                                                                DRJER JE REDUCED PRINCIPAL EFFICIENCY COMPONENTS
                                                                                                                                                                 AJONUMGAPPLIES JALY IF AQACCLLCGATED DUTPUTS
                                         ~
ALONNA STEORING ERFICIENCY AVALYSIS
EVALUATINY MOULE GROEV AND IZ
                                         z
                                                                                                                                                                                                                                                                     NUMBER DF RESIDUAL MODES NR= 10
                                     THÉ CONTROL DÉSIGN MOCEL DROER
                                                                                                                                                                                                                                      ;
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 SELECTED ADDES: 11 12
                                                                                                                                             m
                                                                                                                                                                                                                                                                                                                                                                                          2.3936952432
2.9573953520
                                                                                                                                                                                                                                                                                                                                        10.2347723461
                                                                                                                                                                                                                                                                                                                                                    12.9051041170
                                                                                                                                                                                                                                                                                                                                                                1.3420152508
                                                                                                                                                                                                                                                                                                                                                                              1.6647171077
                                                                                                                                                                                                                                                                                                                                                                                                                   3.3932125322
                                                                                                                                                                                                                                                                                                                                                                                                                                4.2044559923
                                                                                                                                                                                                                                                                                                                                                                                                                                             4.5620581235
4.7552503174
                                                                                                                                                                                                                                                                                                                                                                                                                                                                        3.5334143164
                                                                                                                                            H T Z
                                                                 -
                                                                 H
                                                                                                                                          NUMBER DE CUTPUTS
                                                             NUMBER OF INPUTS
```

INIT. GEN. DISPLACEMENT
12	.000000		
11	000000.		
01	C:0CCC.		
¢	. 000000		
ъ	. 990300		
~	.000000		
ډر ا	.000000		
ŝ	.030300		
£	1.000000		
m			
7	.000000.		
**			
	-1		

## INIT. GEN. VELOCITY

12.00000 11. 1000000. 6 6 8 0000000-1000000. , 000000 .003039 , 0000300 

-----LIVEL? 20102111C REGULATOR CONTROL DESISY

CPTIMAL GAIN MATRIX

-1.37535482E-32 1.41421233E+00 -9.05713811E-01 -1.41283475E+00 804 I

CL3SED L32P EISENWALUES Real Part Imaginary Part

-.10254772E+02 --10244383E+00 102647725+02 -.13244383E+00

-.1293510+5+02 -.17511323E-01

.129051045+02

-.175110235-01

2.13674. .49999 2.035748 3.477907 2.499913 9.764363 -5.175956 .019071 3.2.336747 -5.475956444444444 .493051 4.3.17907 .019071 .493051 57.118582 ------1N RDUTIVE CLEIGF-----

CJMPLEX CLC5EU-LJJP EIGENVALUES --1024438E+0G --1028477E+02

.1025477E+02 -.12905105+32 •12905105+32

--10244335+30 --17511325-31 --17511025-01 I.

.0013509

.0399500.

RATICS :

DAMPING

DAMPING AATIC JF CONTROLLED MODES

-----AGNEE BELING NI------

RICCATI MATRIX

IN ROUTINE FOR EFFICIENCY COMPUTATIONS

INITIAL MODAL STATE VECTURE

-.30333755+00 .0000006+50 .81171592+00 .000000000+50

PER CENT EFFICIENCIES

. . . . . . . . . . . . . . . . . .

STATCIES CONFERENCE

.2350+536+32

GLUBAL EFFICIENCY

.1333403E+50

RELATIVE MODEL EFFICIENCY:

.43210925+01

---GLUBALLY MINIMU4 COST Sa+++: .2918747E+01

---MJDAL COST S4---: .68603785+32

---REAL CDIT SR---: .15875492+04

GLJ3AL SPILLOVER QUDTIENT: .5204443E+03

RELATIVE MODEL SPILLOVER QUDIIENT: .9567891E+00 -----IN ROUTINE SIMDAT------IN ROUTINE RESSIM ------IN ROUTINE MODEFF-------PRINCIPAL COMPONENTS ANALYSIS OF GLOBAL EFFICIENCY

90

1 33.509665 2 |-4.971874 3 1.69595 3 1.69595 4

•

1921526+51

•937230E+00

UMERATUR COMPONENT COSTS:

ENDRINATOR COMPLETE COSTS:

.2918756+01 .1567555+34 .183941E-02 -.000264 -.660254 .000001 1 --.000001 .062252 .000000 2 .000000 .002715 -.000015
3 .020563 .0000014 .000380
4 .-.011361 -.000007 .264648
------IN RCUTIME EMCD5EL-----EFF...NUM...CONTROL CCST : .15675.
EFF.LCIENCT CMSCK AFTER DECOMP.: .15675. m ~ --

GLDB.. EFF..M3DAL MATAIX T

CHARACTERISTIC EFFICIENCIES

1 51555 • 001555 • 001000

-~-.

L.,

1 •001229 •042001 •001229 

EFFICIENCY VECTOR

.001238 .000628 .000002 -nm+

172

.00154

TOTAL EFFICIENCY =

.151053E-02 •• .3278785-03 .2385422402 .1553795404 UNJRJERES. CC4P EFF(NU4.CG4P.CJSTS)/JEND4.) .5231176-J3 .1210295402 .327878240 JNGRJEREJ CG4P..EFFICLERCLES . .327878240

化物物物物物物物物物物物物的 计可存储存储 化合物合物 化合物物合物 化合物物物 化合物物 化合物

•1334415-32 2 .1510535-02 1 .3278786-03 MP TH ORDER EFFICIENCY END: C242..EFF.. MUDECIN MIDSELY

BLOCK-DIAS NUMERATOR & DENGMINATOR COMP. COSTS:

 237+316+52 .997230E+30 E

.1563565+04 .1921525+31 6

.2918756+01 BLGCK-DIAG.EFF...NUM...CONTROL COST :

.122896E-02 5T : .1587405404 .4203076-01 .1221 BLOCK-DIAG.EFF...JENDM...CDNTRCL CDST UNDRDERED COMP..EFFICIENCIES : .4

# 

BLJC<-DIAG..CJMF..EFF.. .420075-01 .1228866-02 (Jascon Nijacon 2

.183870E-02 

S

29.631343 -3.257025 2.120231 -1.374122

\*\*\*\*

**•5757246+02** .103137£+J1

UMÉRATOR COMPONÊNT COSTS:

.685038E+32 .158765E+04 .432109E-01 I 2 3 4 I --,009245 .015831 .058199 -.003099 2 .020784 .035547 .035322 .658922 3 .022194 -.0003811 .003478 -.000841 4 .025203 .250821 .081403 .001723 ------IN RCUTINE ENOSEL------6FF...NUM...CDMTRCL COST : .688038E+02 6FF...NUM...CDMTRCL COST : .1587656+04 6FFICIENCY CHECK AFTER DECOMP.: .4321095-01

REL.. EFF..MCDAL MATRIX T

.04321

TJTAL EFFICIENCY =

w) J

> 1 - 46 42 4 - 0066932 - 002831 ----

CHARACTERISTIC EFFICIENCIES

.043211 .043211 .043211 -

EFFICIENCY VECTOR

174

.042748 .003239 .003122

Denominator Component costs .1336815+32 .1503785+34 UN3735853. CUMP EFF(AUM-SCMP.C3575)/26434.) : .64343195-03 .4255136-01 .6430195-33 UMURDERED CUMP.EFFICIENSIES : .6430195-33

.4256135-01

MJJÉ(IN MJDSEL) CC4P.2F5. 2 .4256132-01 1 .5495192-03 NP TH CRUER SFFICIENCY ENP: .4321036-01

SLOCK-JIAG NUMERATOR & DENJMINATJR CJMP. COSTS:

+1325555+31 E .237431E+C2

•6755736+62 6 .155366E+94

BLJCK-DIAG.ÉFF...NUM...CUNTRCL CCST : ...6359305+02

\*

MCJE(IN YDJSEL) BLDCK-CIAG..COMP..EFF.. 1 .4321095-01 2 .4321055-01 N<sup>3</sup> TA CRUSA BLUCK-DIAG.EFFICIENCY ENP : .432109E-01 15.07.47.4CLP, C2, NITOIM3. 0.512KLMS.

### EXAMPLE 3.2

```
ACONS-4 STRUCTURE EFFICIENCY ANALYSIS
Evaluation Model Croer NET 12
```

THE CONTROL DESIGN MODEL DROER N= 2

NUMBER OF INPUTS ME 2

NUMBER OF RESIDUAL MODES NR= 10

JRDER JF REDUCED PRINCIPAL EFFICIENCY CUMPDNENTS NP#

NUMBER OF DUTPUTS NH= 3

ARE DESIRED, JTHERWISE A DUMMY PARAMETER)= NJGNUMCAPPLIES DALY IF NONCCLLECATED DUTPUTS

-

```
F T F
T F F
T F F
SELECTEJ MZDUTINE MANAL-------
SELECTEJ MZDUTINE MANAL-------
SELECTEJ MZDES: 11 12
SELECTED ACTUATORS: 11 12
SELECTED ACTUATORS: 3 4
-------14 ROUTINE REDRD-------
STAUCTURAL MATURAL FREQUENCIES FOR THE EVALUATION MDDEL
```

10.2847720461 12.9051041170 1.3420195599 1.6647171077 2.8905954202 2.9573805520 2.957380520 2.9520681235 4.5555555 8.5394143164 8.5334143164

INIT. GEN. DISPLACEMENT

:

12 11 000000 000000. .000009 .000000 2 3 4 .030003 .530993 1.330390 . 335330 --

INIT. GEN. VELOCITY

12 11.00000. 000000° 600000° 00000° 6 200000 . 0000000 . 000000 , 000000 

-----LINEAR QUADRATIC RESULATOR CONTROL DESIGN

CPTIMAL GAIN MATRIX

6.634981745-01 7.703223415-01 -2.913726575-02 -1.004404635+00 R04 1

-7.695641395-01 -1.245054215+00 9.95571342E-D1 -9.93476376E-03 RG4 2

IMAGIVARY PART CLOSED LOCP EIGFNVALUES Real Part Im

--1+5+9753E+00

-.102947726+02

102947725+02 --193373715-01 -.14549753E+00

-.12905104E+02

.12305104E+C2 --19837371E-01

RICCATI MATRIX

COMPLEX CLOSEU-LOOP EIGEVVALUES -.1454975E+00 -.1029477E+02 10234775+32 -.12905136+32 - • 1454375E+00 -.1;33737E-01

.1290510E+02

DAMPING ARTIC OF CONTROLLED MODES

-----IN ROUTINE EFFNCY-----

-.1933737č-01

\*\*\*\*

04MP143 RATISS : .0141455 .0015372

IN ROUTING FOR EFFICIENCY COMPUTATIONS

INITIAL MODAL STATE VECTOR:

-.30333756+00 .000000000000000 .81171535+00 .00000006+00

PER CONT SFFICIENCIES

EFFICIENCY COEFFICIENT:

.30553905+02

GLOBAL EFFICIENCY

.25753435+30

RELATIVE AGGEL EFFICIENCY:

.7894446E+31

--+6LEAALLY MI%IMUM COST Sa---: . .3593351E+01

---M304L C35T SH---: .1101562E+03

---REAL CIST SR---: .13952976+04

GLOBAL SPILLGVER QUOTIENT: .35764315+03

C a

1 37.060355 -4.067373 2.250434 013155

- ~ ~ \*

# ENDMINATOR COMPONENT COSTS:

### .2175795+01 .141556E+01

UMERATOR COMPONENT COSTS:

GLOB.. EFF..MODAL MATZIX T

179

1 • 001555 • 001015 • 00000•

CHARACTERISTIC EFFICIEVCIES

1 • 331579 • 034757 • 034757 • 084732

EFFICIENCY VECTER

ы с)

1 . 454364 . 311976 . 303695 

TOTAL EFFICIENCY =

-00258

.205215E-02 •• .5231965-33 .1647.526.402 .13794.124.04 UNDEDERED.. CLAP EFF(NUM.COMP.COSTS)/GENDM.) .101524E-02 .1560105-02 UNDRUERES . .523196E+0

ausasseesucceesusseesusseesuccesseesasseese Methjo 1: Interleting Contributions to the NP EFF...NJHER.. Compouents, IN DECTEASING ORDER:

化合物合物 计存储存储 经资源存储存储 化合物 化合物 化合物合物 化合物合物合物 化合物合物 化合物合物

.257535E-02 CDECT. MCDSELD COMP. EFF. 2 .2052156-02 1 .5231959-03 NP TH CRDER EFFICIENCY ENP: ACDECIN MCDSEL)

BLOCK-DIAG NUMERATOR & DENOMINATOR COMP. COSTS:

.157132E+02 .1+1556E+01 6

.

1373256+04 .2176795+01 £ .3593355+01 BLDCK-DIAS.EFF ... NUM ... CONTACL COST : BLJCK-DIAG.EFF...JEVD4...CDNTRCL CJST : .1394965+04 UNDRJEREJ CJMP..EFFICIENCIES : .847567E-01 .157939E-02

\*

BLOCK-DIAG..COMP..EFF.. .3475576-01 .1579336-02 ADJECIN MODSEL)

.2575956-02 EFFICIENCY NP TH JAJER BLOCK-JIAG.EFFICIENCY ENP : ------IN KJUTINE MJJEFF------Principal Cumpjnewts analysis of relative

1 37.043522 -4.103113 -2.244774 .33030 •

.

ці С CHRRACTERISTIC EFFICIES EFFICIENCY VECTOR REL.. EFF..MCDAL WATRIX T ÷6870. TOTAL EFFICIÈNCY = 1 - 363470 - 012421 - 003412 - 112421 .073350 .086133 .078356 .085133 1 • 0177547 • 001070 • 00002355 • 00000 -4 

.

.

UMERATOR COMPONENT COSTS:

•I+5+62E+01

1015955+C3

5T : .1394956+34 .3513105-01 .798541E-01 .779458E-01 .783413E-01 .110120E+03 BLOCK-SIAS NUMERATOR & DENOMINATOR COMP. COSTS: 特特特特特特特特的特别的特别的特别的特别的特别的特别的特别的特别的分子的特别的特别性性性性性性性 \*\*\*\*\*\*\*\*\*\*\* .789445E-01 NP TH CROER BLOCK-DIAG.EFFICIENCY ENP : 15.03.04.UCLP, CB, N1701H3, 0.512KLN5. BLJCK-DIAG..COMP..EFF.. .8613105-01 .7335415-01 BLOCK-DIAG.EFF...VUM...CONTROL COST : Denominator Component Costs .1571325+02 .1378255+04 4308(14 433581) COMP. 696. 2 . 7794586-01 1 . 9936956-03 NP TH 38366 EFFICIENCY ENP: J .143352E+J1 E .1085805+03 CISCOM NIDECK MODECIN MUDSEL)

#### 3.6 3 EXAMPLES 3.3

AGDIS-4 STRUCTURE BEFICTENCY ALALYSIS Evaluation Adoêl orde: Neb 12

ž THE CONTROL DESIGN MODEL DRDER

~

¥ NUMBER CF INPUTS

.

NUMBER OF RESIDUAL MODES NR= 10

# d N DRDER 3F REJUCEJ PRINCIPAL EFFICIENCY COMPGNENTS

m # X Z NUMBER OF OUTPUTS NJDNUMCAPPLIES ONLY IF NDNCGLLOCATED DUTPUTS

--

ARE DESIRED, DTHERWISE A DUMMY PARAMETER)=

+---- ROUTINE BANKE AND A RI-----SELECTED MODES: 

SELECTED ACTUATORS: 1 2 3 4 ------IN ROUTINE MATORO-------------IN ROUTINE REDRO-------Structural Matural Frequencies For the evaluation Model SELECTED ACTUATORS:

21

=

10.2947720461 12.9051041170 1.3420152509 1.6647171077 2.8906962402 2.9573860520 3.3932126302 4.2044659928

4.0620561235
4.052653174
4.752653174
8.5394143164
9.2305565555555555
-----1N ROUTINE JUTPUT-------1N ROUTINE JUTPT----

INIT. GEN. DISPLACEMENT

INIT. SEN. VELOCITY

.030000 11 -000000 9 10 •000000 •000000 8 900000 5 500000 . 000000. • 000000 • 000000 • 

-----LINEAR QUACRATIC REGULATER CONTROL DESIGN

OPTIMAL GAIN MATRIX

RC# 1 -2.47285917E-01 9.3226922E-01 -1.14545814E+00 -9.80532237E-)1

RQW 2 -2.472386086-01 8.323068476-01 -1.145507596+00 -9.805417896-01

ROM 3 -3.14605526E-02 -5.55576535E-01 7.29131982E-01 1.07580161E-01

CLDSED LODP EIGENVALUES Real Part Imaginary Part --26383437E+00 --10284814E+02

-.26333437E+30 .10234814E+02

--115113542+03 --129050525+02

-.11511354E+60 .12905052E+02

-1 N M +

-.10234915+02 .10224315+02 -.12905055+02 .12905055+022

CO47LEX CL3SE7-L33P EIGEWVALUES --26383446+00 --1028

-.26383446+00 -.11511352+00 -.11511352+00

RICCATI MATRIX

----ADNEE ERITOR VI-----

DAMP2'S AATIC OF CONTROLLED MOCES

02447[NG RITICS : .0236444 .0039197

IN AGUTINE FOR REFICIENCY COMPUTATIONS

INITIAL MODAL STATE VECTDR:

--33333736+93 .00093005+30 .8117159E+00 .00000006+00

PER CENT EFFLUIENCLES

**EFFICIENCY COEFFICIENT:** 

.4510910E+01 Glcaal effictency

**.**5201873E+01

RELATIVE 43DEL EFFICIENCY:

2353527E+02

---SLDBALLY MINIMUM CDST S4---: .1520400E+02

---MJDAL COST SM---: .7010426E+02

---REAL CJST SR---: .2451517E+03

GLOBAL SPILLCVER QUDTIENT: .1151325E+02

RELATIVE MODEL SPILLOVER JUDITENT: .7140373E400 -----IN AQUTIVE SIMDAT------IN ROUTINE RESSIM ------IN RCJTIVE MODEFF------PRINCIPAL COMPGNEVTS ANALYSIS JF GLOBAL EFFICIENCY

0 4

15.273354 -2.916153 1.542495 .707205

۰,

1253345+32 .2570515+31

UMERATOR COMPONENT COSTS:

.152040E+32 .245152E+03 .620137E-01 .000011 -.023426 -.002220 ..001862 .005663 -.000023 .673449 .007642 4 I 2 2 3 --000187 .097364 .000011 2 --000133 .241018 -.002220 -052617 .000599 .002663 --073154 .000699 .673443 ------Ei KGUTEAE EMODSEL------EFF...NUM...CONTROL COST : .15 EFF..0ENQ...CONTROL COST : .24

m

SLJ3.. EFF..MJDAL WATRIX T

186

1 • 352257 • 03-583 • 011035 • 002040

HNM+

... ບ

CHARACTERISTIC EFFICIEVCIES

1 • 053702 • 280174 • 053603 • 280139

EFFICIENCY VECTOR

1 • 051138 • 009719 • 000590 • 000572

TCTAL EFFICIENCY =

.06202

.5556945-01 BLDCK-DIAJ.EFT...DENDM...CJNTRDL CJST : 2444288403 UNDRDEREJ COMP..EFFICIENCIES : 2301548400 5370148-01 , •522023E-01 .152040E+32 EFFICIENCY •• .644923E-02 BLDCK-DIAS NUMERATOR & DENOMINATOR COMP. COSTS: \* .3537415+21 .2355145+03 JN.C.A.26482.. CC4P EFFRNUM.CC4P.CC3T5)/UEN3M.) .10+5536-01 .5153292-01 .NCA26467 CC4P..6FFIC16%CIES : .66469236-0 METHOD2: 3LUCK-DIAGCMAL CUNTRIGUTIONS TU NP EFFICIENCY COMPONENTS IN JECREASING DRDER: -520187E-01 NP TH CRDER BLJCK-DIAG.EFFICIENCY ENP : ------IN RJUTINE MJJEFF------Principal Components analysis of relative 3LJCK-DI4G..CDMP..EFF.. .2801545+00 .5370145-01 50 BLOCK-DIAG.EFF...NUY...CONTROL COST : Uenuinator Component costs 2352536+03 .9175725+01 2 .5556942-01 1 .5549295-02 NP TH JRJER EFFICIENCY END: C34P..EFF.. .2570615+31 5 .126334E+ù2 £ 1 15.153113 -3.234243 1.5588939 1.533032 MODECLM MODSEL) MODECIN MODSEL) ~ - N M +

.

#### .6700565+02 .3099706+01

-

## UMERATIR COMPONENT COSTS:

•	.007813	.009083	.006122	.675713		3435+02	1525+03	.2359636+00
m	002642	1.025133	.005164	006140		.1010	.2451	:*46
~	.099900	.027163	000330	053361	E EMCOSEL-	: אשר ככא	R2L C35T :	AFTER DEC
-	.000755	100515	.052313	078934	VITUDA NI	TNCD HUN	DENCH CONT	LENCY CHECK
	-1	~	'n	4			EFF.	ĒFI(

REL.. EFF..MJOAL MATRIX T

.28536 TOTAL EFFICIENCY =

.003205 .002715 266241 013801

ىدە ر.

1 • 337803 • 042663 • 009913 • 003549 

CHARACTERISTIC EFFICIEVCIES

7	.243879	.323446	.323460	.284267

-----

EFFICIENCY VECTOR

BLOCK-DIAS.EFF...JENCM...CONTROL COST : .244428E403 UNDRDERED COMP..EFFICIENCIES : .323205E+00 .284259E+00 -274030E+03 .235720E+00 .693381E+02 BLOCK-DIA; NUMERATCR & DENJMINATOR COMP. COSTS: \*2859636+00 NP TH DRDER BLÜCK-DIAG.EFFICIENCY ENP : 15.08.14.46LP, GE, MITJIM5, D.5124LM3. BLJCK-7145..CJMP..EFF.. .3232052+00 .2842536+00 BLJCK-CIA3.EFS...NUM...CONTRCL COST : Denominator Component Costs .9175726+01 .235253E+03 Z 27+980 - 00 Z - 27+980 - 00 I - 1136255-01 NP TH QZDER EFFICIENCY END: HODECIN MODSEL) COMP., EFF., .2955645+31 & .6587255+02 £ MODECIN NUDSEL)

٠ .

I'LIT. GEN. DISPLACEMENT

.....

----ISINI BNILDER NI-----

4.2 EXAMPLES 3.4 &

```
-
                                                                                                                                                                                                ARE DESIRED, DTHERWISE A DUMMT PARAMETER)=
                                                                                                                                         NP= 16
                                                                                                                                                                                                                                                                                                                              SELECTED ACTUATCAS: 1 2
------IN RDUTINE MATDRD-------
------IN RDUTINE RECRD-------
Structural Natural Frequencies for the evaluation Model
                                                                                                                                      DRJER DF REDUCED PRINCIPAL EFFICIENCY COMPONENTS
                                                                                                                                                                                                                                                                                                  ø
                                                                                                                                                                                                     NJDNU4(&PPLIES JNLY IF NENCOLLOCATED OUTPUTS
                                                                                                                                                                                                                                                                                                  ŝ
                                                æ
ACG55-4 STRUCTURE EFFICIENCY ANALYSIS
Evaluation Model Order Ne# 12
                                                                                                                                                                                                                                                                                                    ...
                                                ž
                                                THE CONTAUL DESIGN MODEL DRDER
                                                                                                                                                                                                                                                                                               m
                                                                                                                                                                                                                                                                                  ----IN ROUTINE MANAL------
                                                                                                            NUMBER OF RESIJUAL MODES NR#
                                                                                                                                                                           m
                                                                                                                                                                                                                                                                                                 ~
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      4.6620681235
4.7552603174
8.5334143164
9.2505565555
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     = 11
                                                                                                                                                                                                                                                                                                                                                                                                             1.3420152508
                                                                                                                                                                                                                                                                                                                                                                                                                            1.6647171077
                                                                                                                                                                                                                                                                                                                                                                                                                                          2.8306952402
                                                                                                                                                                                                                                                                                                                                                                                                                                                         2.9573960520
                                                                                                                                                                                                                                                                                                                                                                                                                                                                         3.3992125302
4.2044653329
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     10.2347720461
                                                                             NUMBER OF INPUTS M# 2
                                                                                                                                                                                                                                                                                                    -4
                                                                                                                                                                         NUMBER OF OUTPUTS
                                                                                                                                                                                                                                                                                                 SELECTED 400ES:
```

```
---
                                                                                                                                                                ARE DESIRED, OTHERWISE A DUMMY PARAMETER)=
                                                                                                               NP= 16
                                                                                                                                                                                                                                         æ
                                                                                                                                                                                                                                                                      DRJER DF REDUCEJ PRINCIPAL EFFICIENCY COMPUNENTS
                                                                                                                                                                                                                                          ÷
                                                                                                                                                            NJJNUMCAPPLIES JNLY IF NENCOLLICATED DUTPUTS
                                        nD
ACGSS-4 STRUCTURE EFFICIENCY AMALYSIS
Evaluation Model Groek Wee 12
                                                                                                                                                                                                                                         ۍ
                                        ž
                                                                                        4
                                     THE CONTRUL DESIGN MODEL DRCER
                                                                                                                                                                                                                                       m
                                                                                                                                                                                                                                                                                                                                                                                                                                                                  12-3051341176
----IN RGUTIVE JUTPUT------
                                                                                                                                                                                                                       ----IN RCUTIVE MANAL------
                                                                                   NUMBER DF RESIJUAL MCDES NR=
                                                                                                                                                                                                                                                                 ~
                                                                                                                                        m
                                                                                                                                                                                                                                      ~
                                                                                                                                                                                                                                                                                                                                                                           3.3992125302
4.2044553329
4.6620581235
4.7552603174
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         ----ISLAI BAILDER NI-----
                                                                                                                                                                                                                                                                 -
                                                                                                                                      ービズ
                                                                                                                                                                                                                                                                                                                            1.3420152502
                                                                                                                                                                                                                                                                                                                                                    2.8306952402
                                                                                                                                                                                                                                                                                                                                                               2.9573960520
                                                                                                                                                                                                                                                                                                                                                                                                                            8.5394143164
                                                                                                                                                                                                                                                                                                                                                                                                                                         3.2505565555
                                                                                                                                                                                                                                                                                                                                                                                                                                                     19-2347720451
                                                               ~4
                                                                                                                                                                                                                                                                                                                                     1.6647171077
                                                               H
7.
                                                                                                                                                                                                                                         --
                                                                                                                                      NUMBER OF GUTPUTS
                                                                                                                                                                                                                                                               SELECTED ACTUATORS:
                                                             NUMBER OF INPUTS
                                                                                                                                                                                                                                      SELECTED MODES:
```

INIT. GEN. DISPLACEMENT

· · · · · · · · · · · · · · · · · · ·		11 12 . 340060 . 000000																				
		100000.																				
. 00000		6 00000.			57306+00 94036-01 12056+00	3599E+60 11041E-01 12815E+00																
.0000.		0 n 0 0 0 0 .			1.325 -9.9002 1.6579	1.3254 -9.9023 1.6612																
, 00000.		7 • 00000			19927902E-01 16357344E-01 17390459E-01	192771962-01 16155062-01 164542372-01																
• • • • • • • •		001J00.	1		100 101 100 100	00 10 10 10 10 10 10 10																
500030		<b>2</b> 00000.			2.35914776E- 9.93682125E- 3.20262258E+	2.35996232E- -9.92929421E- .3.20044329E+																
1.333620		• 000000	191 DESIC		3515-01 322-01 043 <b>5-01</b>	857E-01 161E-01 182E-01																
3 • 00000	EN. VELDC		ואנס אכדען		9.919059 -3.42259 9.913926	- 9.915279 3.40983 - 9.313294	RY PART	35225+01	35225+01	59296+01	59296+01	19805+01	1980E+01	82026+01	82025+31	94542+01	04545+01	05165+01	05165+31	6000E+01	6006+01	03535+31
2 • 0 0 0 0 •	INIT. 5	-000000 46 contro: 146 users	RATIC REJU	×	53089E-01 43879E-01 53965E-01 83700E-01	30.357E-01 39358?-01 50049£-01 53818E+01	ALUES I4AGI4A	3397	7466.	2959	.2959	4203	.4203	1664	.1654	1342	.1342	4735	. 4755	2891	.2891	- 4661
1 • 900099			INEAS TUAD:	LEN MATAL	-1.72+ -8.884- 8.031 9.3661	1.722 -8.934 -8.076 9.3756	DOP EIGENVI Part	13270E+30	82705+30	15256E+00	\$256E+00	183956+30	80955+00	93236-01	93236-01	10846-01	10-34601	5350E-01	5030E-01	54056-01	5405E-01	6707E-01
-			]	0 147140	R0.4 1	2 108	CLJSED LC Real	3496	3495	2423	2423	2823	2829	6736	6736	4306	4 3 û 6	6536	- 6636	3781	3781	

+++231767012-21 +++54\*03632+C1

PICCATI MATRIX

51167       .000.0       1200.0       727743       .000.0       1200.0       1001.0       .0001.0       <		1	5	en.	4	5	¢	~	æ	6	10	11	12
04445       -010437       000352       2.149725       -112253       -000352       2.149725       -112253       -000352       2.149725       -112253       -000352       2.149725       -112253       -000352       2.149712       -122253       -000352       2.149712       -122253       -000352       2.149712       -10130       -000352       2.149712       -10130       -000352       -01112       -10130       -000512       -00159       -01139       -00150       -01139       -000517       -00150       -01139       -000517       -001507       -01139       -001507       <	4T.	13773-	14 14 15 15 15 15 15 15 15 15 15 15 15 15 15	15 • 0 0 C • 5 9	1¢ .30380a	.000613	000222	255526	527857	.000220	.000621	553494	203258
0100100101000101000101000101000100000010000000100000010000000100000001000000010000000100000001000000010000000100000001000000010000000100000001000000010000000010000000100000000100000000100000000100000000100000000100000000100000000100000000100000000100000000010000000000001000000000000000100000000000000000000100000000000000000000000000000000000			374397 23.473605	422000	.036039 .093155	. 200973	.00355	2.493013	.049250	000195	.000352	2.149722	122252
0.00523       .000137       .010512       .0101733       .010173       .010173		1.515700 .000453	331125	010549 41.365331	• • • • • • • • • • • • • • • • • • •	. 573044	965132	000450	.030413	912521	593365	030014	.000323
<ul> <li>Control 100000</li> <li>Control 10000</li> <licontrol 10000<="" li=""> <li>Control 10000</li> <li>Control</li></licontrol></ul>	•	.000523 .000803	.200127	-962322 -47531J	•219557 15•694535	2.745115	.354235	sr 100	191000	2.553033	160672	,610130	.00018
$\begin{array}{c} -002166 & -000365 & -003613 & .663703 & .292217 & 27.571135 & .005373 & -005303 & 5.973379 & -1.036744 & .001539 & -001377 & -000255 & -394208 & -394208 & -394208 & -394208 & -394208 & -394208 & -3091265 & -3591265 & -3591265 & -3591265 & -0001265 & -0001265 & -0001265 & -0001265 & -0001265 & -0001265 & -0001378 & -0001378 & -0001378 & -0001376 & -0000333 & -0000339 & -000339 & -000339 & -000339 & -000339 & -000339 & -000339 & -000339 & -000339 & -000339 & -000339 & -000339 & -000339 & -000339 & -000339 & -000339 & -000339 & -000339 & -000339 & -000331 & -1000315 & -1553655 & -11.702425 & 5.973379 & -000625 & -480358 & -000339 & -000339 & -000339 & -000333 & -100339 & -000333 & -100331 & -000333 & -000332 & -000332 & -000332 & -000332 & -000332 & -000332 & -000332 & -000332 & -000332 & -000332 & -000332 & -000333 & -0$		001033	.000130	-2.374332	.187535 2.7.511523	28.33342	.290217	058713	-*005054-	11.702825	-4.213543	005677	002619
$\begin{array}{c} -000653 & -000157 & -2 334400 & .381024 \\ -000177 & -000177 & -000177 & -001177 \\ -000131 & -000131 & -000197 & -001177 & -001177 & -001177 & -000553 & -196911 & -2547785 \\ -000131 & -000195 & -31200197 & -000033 & -005653 & -002007 & 35.985477 & .480358 & -002399 & -005203 \\ -000132 & -000195 & -3120212 & -555933-11.702825 & 5.973379 & -002007 & 35.985477 & .480358 & -0003399 & -005203 \\ -000151 & -0020195 & -3120512 & -555933-11.702825 & 5.973379 & -002007 & 35.985477 & .480358 & -000339 & -005203 \\ -000151 & -0020195 & -3125212 & -555933-11.702825 & 5.973379 & -002007 & 35.985477 & .480358 & -000339 & -005203 \\ -000151 & -002021 & -0003196 & -0003190 & -001579 & -000207 & 000338 & 5.034535 & -000203 \\ -000021 & -002021 & -0003190 & -001579 & -0016519 & -1000399 & -000338 & 5.046077 & .453758 \\ -000021 & -000202 & -0003190 & -000519 & -000351 & -1.847329 & -2564795 & -000203 & -0000313 & -0.5071158 \\ -000021 & -000202 & -0003190 & -0003190 & -000351 & -1.847329 & -2564795 & -000203 & -000015 & .463058 & 3.849973 \\ -112222 & -000022 & -000129 & -000139 & -000157 & -8804003 & -2527993 & -000203 & -0000203 & -000203 & -000203 & -000203 & -000203 & -000203 & -000203 & -0000130 & -0000518 & -0000203 & -0000203 & -0000130 & -0000157 & -6804003 & -527790 & -000203 & -000120 & -000130 & -0000130 & -0000157 & -6804003 & -5277903 & -000120 & -000130 & -000130 & -0001257 & -580433 & -2334400 & -000154 & -0001257 & -000130 & -000130 & -000130 & -0001257 & -000130 & -000130 & -0001257 & -000130 & -000130 & -0001457 & -8804003 & -5277903 & -0001257 & -2074992 & 546533 & -2334400 & -000154 & -000157 & -000130 & -000157 & -6804003 & -527790 & -0001253 & -000130 & -000130 & -0001254 & -0001257 & -2074992 & 546533 & -2334400 & -0000244 & -0001257 & -20149726 & -0001257 & -20149726 & -0001257 & -20149726 & -0001257 & -20149726 & -0001257 & -20149726 & -0001257 & -20149726 & -0001257 & -20149726 & -00001254 & -2000126 & -2000126 & -0000126 & -0000126 & -0000126 & -0000126 & -0000126 & -0000126 & -0000023 & -0000023 & -000$	•	002156	•0003541 •000355	5.565383 985132	.663303 .394238	.290217	27.571135	.005373	005303	5.973879	-1.036744	.001539	003351
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	1	000653	000157 2.453013	-2 334400	001205	058713	.005373	37.665093	.486657	.006653	.001462	-2.848985	-1.947329
$ \begin{array}{c} 2.248115 &527993 &000345 & .000345 & .000333 &002307 & 35.385477 & .480358 &300399 &002302 & .000105 & .150557 & 2.003325 & 2.163677 & .016672 & .000332 & .000332 & .000321 & .000321 & .000322 & .000321 & .000321 & .000321 & .000322 & .000322 & .000321 & .000321 & .000323 & .000679 & .000332 & .000323 & .000679 & .000332 & .000679 & .000332 & .000671 & .01670 & .000451 & .00015 & .068765 & 8.313125 & .028415 & .000125 & .000125 & .0000521 & .0000551 & .557793 & .0000551 & .0000521 & .0000521 & .0000521 & .0000521 & .0000521 & .0000521 & .0000521 & .0000521 & .0000521 & .0000521 & .0000521 & .0000541 & .0000541 & .0000541 & .0000551 & .0000551 & .0000551 & .0000551 & .0000551 & .0000551 & .0000551 & .0000551 & .0000551 & .0000521 & .0000551 & .0000551 & .0000551 & .0000551 & .0000551 & .0000551 & .0000551 & .0000551 & .0000551 & .0000551 & .0000552 & .0000551 & .0000551 & .0000551 & .0000551 & .0000551 & .0000552 & .0000551 & .0000552 & .0000551 & .0000551 & .0000551 & .0000551 & .0000552 & .0000551 & .0000551 & .0000551 & .0000552 & .0000551 & .0000552 & .0000551 & .0000551 & .0000551 & .0000552 & .0000551 & .0000552 & .0000551 & .0000551 & .0000551 & .0000552 & .0000551 & .0000551 & .0000551 & .0000552 & .0000551 & .0000552 & .0$	• •	-5-113316 -527857	8040C3 .045255	000154 .000413	000197	005054	006803	. 436657	4.344091	002001	.000525	3.796911	254735
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		2.248115	527903 000135	912345	.000033 2.553633-1	11.702825	5.973879	.006553	002001	35.985477	<b>.</b> 480358	200399	01010
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		.000451 .000321	-001050	1.505675 533365	2.073657	-4.213543	-1.086744	.001452	.000625	.460358	3.034535	.000333	100315
1.08432       -7.491363       .659325       .003321      000519      000519      000519      000519      000519      000519      000015       .463059       3.843973         2.031258       -11.259139       -504415       .000552       .000552      000523      0000199      125179         3.46952       11.259139       -504415       .000523      001089      072156      000523      001089      125179         3.46952       1.251670       .000523      001089      072156      000163      463054       8.913125         3.69552       .000523      001089      072156      000551      001893      7.91505       -1.253139         0.74897      331126       .00127       .000130      079541      001653      014925       -1.253139         0.74897      331126       .001267      001833      000093      001050      001657      001653      001932      0014925      0014126      0014925      0014325      0014126      0014925      0014325      0014126      0014925      0014126      0014925      0014925      0014925      0014425      0014126      0014126       .	•	001331	.020.93	-3.935435	- 000130	- 045477	.001539	-7.893.5-	116267.5	300399	.000333	69.644007	.453358
203258122552 .000020 .000019000519000351 -1.84732900020900020900020 .*00029 .*0002900183 8.918165 -1.259199004125 .000059001699002155000650 -5.113816 2.244116 .000451001831-21.084825 8.913135 5.939235 .028415 .001245 .000129001099002155000650 -5.113816 2.244116 .000451001831-21.084825 8.913135 5.939235 .028415 .001245 .000129001039000154000033527903 .001050109150 -1.259199 5.028415 25.410065001353 .00005300005380400000154000346 1.505675 -3.935406 .009325004125 5.00023 .0001245001853344.34932 .5565333 -2.334400000154000346 1.505675 -3.935406 .0093250014125 5.00023 .000128 .219567 .137835 .558303 .381024 .000177 .000083 2.079657 .216343 .003322 .000552 5.00029000129279201 15.090719	ĩ	1.034325	-7.491363	.003325	.003321							0 7 0 6 7 7	6706.40 <b>6</b>
.846952 1.516700 .000523001089022155000650 -5.113816 2.248116 .000451001831-21.094825 8.31315 5.33325 .028415 .001245 .602030 .00541000157804003527303 .001050 .000930 -7.491560 -1.253133 .028415 25.410065001353 .000093 .000093000157804005003346 1.505675 -3.935406 .009325004125 .000229000557 .952522 -2.074992 5.565333 -2.334400000154000346 1.505675 -3.935406 .009325004125 .001245001853344.349333 .374201 .568303 .381024 .0001540003346 1.505675 -3.935406 .00332' .000552 .002031 .000128 .219567 .137435 .558303 .381024 .000177 .000083 2.0779657 .216343 .00332' .000552	•	.203259 8.918105	122252 -1.253199	.000020 004125	.000219 .000552	000519	000351	-1.847329	254145	000203	c T A A A A * -	SCD104.	
5.333255 .028415 .001245 .602031 074837331125 .000127 .6020130 .000541000157804003527903 .001050 .000030 -7.~91360 -1.253139 .023415 2.5040650001353 .000053 .5565333 -2.334400000154000346 1.505675 -3.935406 .009325004125 .000229000549 .962522 -2.074992 5.565333 -2.334400000154000346 1.505675 -3.935406 .009325004125 .000229 .000153 .454933 .374201 .00333 -2.334400000154 .0003346 1.505675 -3.935406 .003325004125 .001245001853344.349333 .374201 .558303 .381024 .0000177 .000083 2.0779657 .216343 .003327 .000552 .002031 .000093 .2174257 .137435 .558303 .381024 .000177 .000083 2.0779657 .216343 .003327 .000552	•	. 346952	1.515700	.000523	001089	002155	000650	-5.113816	2.249116	.000451	001831-	21.054825	8.913135
-028415 25.410065001353 -000053 -000053 -000229 -000549 -962922 -2.074992 5.665333 -2.33440000015400346 1.605675 -3.935406 -009325004125 -00124500185344.34933 -373201 -002041 -000053 -219567 .137335 -558303 .381024 .0020177 .000083 2.079657 .216343 .00332' .000552 -002011 -000083 -219567 15.090719	5	6.333235 -074837	.028415 - 331126	.001245 003127	.602031 000130	000561	000157	100708	527903	.001050	060000.	-7.491360	661652.1-
.000229000549 .962522 -2.074992 5.56533 -2.33440000015400346 1.505675 -3.935406 .009325004125 .001245001853344.34933 .374201 .000031 .000128 .219557 .137435 .568303 .381024 .000177 .000083 2.079657 .216343 .00332 <sup>*</sup> .000552 .002031 .000053 .219557 15.090719		.023415	25.410065	001353	£50000°								
• • • • • • • • • • • • • • • • • • •		.000229	000543	- 362522 -	-2.074992	5.565333	-2.334400	000154	00346	1.505675	-3.935406	.009326	004125
-002011 .000053 .373201 15.090719 Th brittys fists		.000033	-000128		.137835	.558303	.381024	.000117	.00.00.83	2.079657	.216343	.00332'	.000552
	1	-14 8041	60000. 145 1 141	.373201	15.090719								

--63061046-01 .13429456+01 --57363335+01 --16649206+31 --57363335+01 --16649206+31 --57363335+01 -16648206+31 --37315416-01 -23316006+01 --24235265+00 -2353556+01

-

COMPLEX CLOSED-LOOP EIGENVALUES --4306103E-01 --1342045E+01

.0140605 .0000002400 .0087595 .0' 1731 -4082741646400 .1324450 .00000001E+30 .0000030E+00 .000300E+00 -0816140 .1573100E+01 .0133766 -25260355400 -3241424240 -94027895-01 •1053513E+02 .040-333 198993366+02 .53212195+91 .33973525+01 ---203134E+01 --+651035E+01 .42331386.431 • 4651335E+31 -.4755352E+01 -+755052E+01 IN RUJTINE FOR EFFICIENCY COMPJERIONS • 0300303€ +30 • 0030000€ +30 • 0033002€ +00 DAMPING RATIS OF CONTRCLLED MODES -0320697 イントーートロスルルロールアストラロメージオートートー ---NOAL COST SH--+: INITIAL MODAL STATE VECTOR: RELATIVE MODEL EFFICIENCY: GLJ3AL SPILLOVER QUDTIENT: EFFICIENCY COEFFICIENT: PER CENT EFFICIENCIES --34 34327E+00 --23234136+35 --20233106+00 -.4037571E-J1 -- \$=\$\$\$335E-C1 --+687571E-01 -.56865352-91 -.6214141E-01 -.35195926+00 -.6878961E-01 DAMPING RATICS : GLCBAL EFFICIENCY •6571713E+01 • 84033535+01 • 5 5 2 5 7 2 9 € • 3 2

-.3397352E+J1

-.3-55327E+30

,

	Ψ U
	1 03774 03774 03774 03774 032052 052052 052052 052557 052557 052557 052557 052557 052557 052557 052557 05250 05250 052000 052000 052000
今 9 4 6 10 10 10 10 10 10 10 10 10 10 10 10 10	

		2	E C	•	2	9	2	60	e	10	11	12
-4	E1 Fff200	14	15 002553	15 000003	665000°	600000.	.0000.	003001	010201	£00000°	100000.	100000
~	•003321 ••000433	.313735	035035 000501	000136 000032	005724	000016	.00000	200000.	J0391£	100000.	.00000	8000CG.
Ē	032357 .303000	007775	.000001	000897	• • • • • •	.000531	.296454	.001036	00005	.1132.4	78+65C.	029763
	.00000	000502 001737	.000000	000434	.000002	006966	003113	013096	.00000	.01510.	.356131	.399775

JLJ8.. EFF..4J0AL 4274IX T

\_\_\_\_

TOTAL EFFICIENCY = .08403

1	.032833	.02123-	.020353	.00++35	C~110C.	.201.52	.000413	.000425	.000249	.000145	111000.	.000038	.000333	.000023	000000.	-000000	
	-1	~	~	4	Ś	ç	~	60	œ	10	11	12	13	14	15	16	

Ω.
O.
-
1.
Ċ,
ш,
>
U.
~
- T.
5
٠,
ы
14.
м.

4	.37.252	.537539	.25552.	. 103615	.367155	. 168713	7-0616.	.525763	CCE600.	.019545	.019051	.019527	.303655	.067733	.037753	.263353	
	-1	~	Ĩ	•	'n	•	1	Ŧ	<b>~</b>	1)	11	12	13		15	15	

1:0000-1	3 .000013 015475	000072 .000072	.000001	00004		061245	.009150	.300016	PIFSOO.	. 900592	003613
00000	2 .030005	000119 000119	.231735	200043	.419184	003877	052032	300055	571806.	005032	.J21198
.00031	• -• 010005 •	.000031 .342453	-000003	. 0 3 3 2 7 7	• 00000 •	00000.	000006	• 000 2 6 4	000021		100000-
.30013	C . 00000. 0	.074748	025423 000015	221982	00000	000000.	.000367	.000353	000002	.0000.2	00005
. 3923ŭ 3000ĉ	5 .005037 581185	.000022 00009	1.01-575 .000773	000002	.000013	.00034	.052323	000000	0000	.000145	.00000
+00000	0 .0000000	110000.	000004 000293	.00003	.002470	000526	1.211178	.00000	.000025	.901723	
• 00000 265130	000000° 0	.0000000	000038 .000000	.052440	.00000	00000.	000002	.000307	00000	000000.	.00000.
•03013 223335	2 .000011 .000011	-033835	00001	1.037501	.000000	00000.	010000-	000382	000001	.000003	000003
.05401	5000073 .000000	6340553	.169200	117800	.0000	.00000.	000031	.389852	< 00000 · -	.000023	110000
- 56607 055343	+16100° 0	.020267	000040	342642	000031	000051	100000.	311350	.000117	.0000.	003025
.37349 600003	2009243	.022312	027627	.00006	000565	030452	.001314	.000015	516860.	036381	.023531
-000034	3 .000000 002693	000012	•000035 •003036	.000019	009376	.136766	.016666	. 300010	185574	353309	.332633
.00007 RGUTI	1 .000001		.000010								
	TRCL COST : Tral Cost : K After dec	.167] .1539 3 <b>M</b> P.:	310E+01 981E+02 • 640836E-7	1							

UMERATJR COMPONENT COSTS:

.1236336+00	.833562E+00	
.1190156-01	.423741E+00 .133573E-01	
.2392675-33	.262362E+00 .421412E-02	

ENDMINATCR COMPONENT COSTS:

.325331E-01 .251583±-01		. 804612Ë-72 . 365441E+30
.156424E-02 .115218E-01 .136957E-02		3E+01 853E+32 •191107E-31 •527735E+33
.1397476+32 .2467056+31 .152706434.) : .5213326-02 .4149166-01 .1300526-03 .8379566-02 .1255926-02	ссеческиенте 1 16 JRJE2: 51 NG JRJE2: 6408366-01 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	1 0 1 :16731. .254875-02 .256875-02 .97+0455-62
.5937346+CC .4135536+20 .72556+20 eff(01%47569+20 .5991226-03 .2129569-01 .2129569-01 .2129569-01 .2129569-03 .2129569569-03 .21200569569-03 .21205690-03 .21205690-03 .21205690-03 .21205	<pre>XBUUGU WING CONT I CUT I</pre>	
4 2 3 9 4 6 6 - 0 1 4 1 1 2 4 7 4 6 1 4 1 1 2 4 7 4 6 1 4 1 2 4 2 4 7 2 6 4 0 6 4 1 2 1 2 6 9 6 1 0 4 4 1 3 1 3 6 9 6 - 0 4 4 2 1 1 7 6 6 1 - 0 4 4 2 1 1 7 6 6 1 - 0 1 0 1 2 2 1 2 6 7 1 - 0 1 0 1 2 2 0 2 2 7 2 4 0 1 2 2 0 2 2 7 2 4 0 1 2 2 2 2 2 2 2 2 2 2 1 0 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	#5444444444444444444444444444444444444	.8335626+00 6 .4214126-02 6 .4214126-02 6 .1335736-01 5 81200-0145.65550 81200-0145.65550 040705620 50495 UNJRJ6850 50495 UNJRJ6850 50495

aunaseounasuuruunouenanasuusesaaneaaanusaa Méthijoi Jujok-Diagomal Contaibutions to NP éfsicieuri Components in deckeasing ekderi •

3L3CK-DIA3CJ#P€FF. .527785€+00 .36541€+00 .255357€+00 .194412€-01 .19107€-01 .974512€-02 .804612€-02 .804612€-02 .73227€-02	
MCDECUN MCDSEL MCDSEL MCDECUN MCDSEL MCDECUN MCDECUN	

5

~

	0
	0 11 h 10 11 m in in in in
	MONMANJANU
~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	4044400
	4 N H C O O O O O O
~~~~	
4NM4N4N40000000400400	

e C

	ENCIES		VECTOR
	EFFICI		LENCY
	RISTIC		) I
ιΟ <b>Δ</b> ιΟ <sup>-</sup> Ο	ARACTE	000	
9 + 1 46 60 NN 46 60 000000 	Ľ		
1 2 4 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7		- いちからの - ちの - ちの - ち しょう しょう しょう しょう - しょ - しょ	

1 .239795	.111515 .063305	.056174	.036873	.016+53	.003032	.006443	.004196	.004054	£1710C.	.001732	.000333	.000207	.000137	-0000-	
-4	~ ~	• •	s	Ŷ	~	œ	Ţ	10	11	12	13	14	15	15	

.55257

TJTAL EFFICIENCY =

REL.. EFF..433AL HAIRIA T

	1	1.4	e <b>1</b>	÷	'n	\$	2	80	6	10	11	12
	13 200001	14 100000	1 2000.	-1 -101	384375	233457	116522.	100000.	00000	ES2E00	.001923	-119401-
• •	.00005	.00000500	000002 000067	.056687 .037374	050272	.003352	.155032	100000	000002	005469	013572	.356435
	.000006	-193925 -193925	003031	150347 000001	100000.	000002	*00000*	005721	.041549	E00000*-	200000.	.000001
	. 640964 151127	• <b>61</b> 0745 - • 067045	.019523 .058472	000007 .000013	· 000314	000302	00000	034380	332390	00001	60000C*-	:00000°
	373-61	012612	15523	.000001	610000	000004	.000221	.035383	.003953	000020	2000.7	.000222
	006532	- 013517 - 013517		-0000-3 -00000-3	- 1001-5	090021	000051	045786	.349102	.000035	000030	.000CTZ
	232151	.036755 .036755		.000211	.331123	034356	.085129	100000	220027	079694	012711	.051131
	.000023	-000033	.000023	.112733 .333544	356373	.053552	111935	000011	000143	.050873	132394	. 34 34 73
	.000101	000015	•00013e	-933652 -000011	- 000322	000019	000005	.335152	020222	-*00001	000001	÷0000°*
	.018633	039505	-017105	.000011	00035	000001	000013	196345	211591	.0000.	000017	000339
	050155	1.130897	.35+134	00007+	.035559	.024175	£73110.	.000012	.00002	106841.	-,399198	052799
	000014	000021		067349	. 228638	- 394194	.101623	.000070	000005	.798551	.063168	.123333
	000043	.000053	.000041	.162357	.001034	080000	000573	000001	200000.	034152	.053910	154500*-
	.000005	100000.	370C00.									
•	000005	1.00004	000031	.146071	.020354	243564	.010933	000002	*10C00*	123651.	610115.	0.0700*-
- 0	109434	010000	.072059	000002	100000	000034	000000.	012767	006535	-000013	£0000C°	.00000
1	.016556	.010653 -110573	02570-	200056	000012	000042	00000	035914	.104750	000012	-,000161	000052
	+2+692.	080405	.413545	00000								
zi	AGUTINE CJNTR	E EMDOSEL	.1099	351E+02								
ຄູ່ວົ	1CJATR - CHECK	AFTER JECS	.1924 1 <b>MP.:</b>	031E+02 .552573E+0	00							

UMERATOR COMPONENT COSTS:

.7553112+01 .1+7526E+01	
.323546E+00 .223571E+60 .392317E+00	
<ul> <li>2618915-31</li> <li>6752736+00</li> <li>325759€+00</li> </ul>	

ENDMINATOR COMPONENT COSTS:

.343834E+00 .713257E-01					.597331E+00
.1544705-01 .113615-01 .1933705-01					55+02 5-35+02 55404905+00 •5404905+00 •5404905+00
.1397475+02 .246705:401 .150705404.) : .3795906400 .7414106401 .1266196402 .3255125401 .1574995401	44444444444444444444444444444444444444	552573E+00	сочР. ссรтร: 1 3 2	- 0 - 0	0 1 :
.531995400 .4136535400 .413654500 .13656500 .162652600 .12359600 .112359600 .112359600 .112359600 .112359600 .112359600 .131546000 .131546000 .131546000000000000000000000000000000000000	saaaaaaaaaaaaaaaaaaaaaaa [N] Cuntrleution5 [N] Cuntrleution5 [ENTS] IN DECREAS Jemesaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa	1049547 	ACTÁNIMINATOR <b>3870316</b> <b>5227645</b> <b>5227645</b> <b>5227645</b>	<ul> <li>1021515+0</li> <li>8023675+0</li> <li>2230936+0</li> <li>4326416+0</li> </ul>	
10-19-46-4 10:-3-4-6-111:- 10:-3-7-6-111:- 10:-1-7-1-1-1- 10:-1-7-1-1-1- 10:-1-7-1-1-1- 10:-1-7-1-1-1- 10:-1-7-1-1-1- 10:-1-7-1-1-1- 10:-1-7-1-1-1- 10:-1-7-1-1-1- 10:-1-7-1-1-1- 10:-1-1-1-1-1- 10:-1-1-1-1-1- 10:-1-1-1-1-1- 10:-1-1-1-1-1- 10:-1-1-1-1-1- 10:-1-1-1-1-1- 10:-1-1-1-1-1- 10:-1-1-1-1-1- 10:-1-1-1-1-1- 10:-1-1-1-1-1-1- 10:-1-1-1-1-1-1- 10:-1-1-1-1-1-1-1- 10:-1-1-1-1-1-1-1- 10:-1-1-1-1-1-1-1-1- 10:-1-1-1-1-1-1-1-1-1- 10:-1-1-1-1-1-1-1-1-1- 10:-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-	40400000000000000000000000000000000000	MODECIA MODSEL) C 3 4 4 6 8 8 2 7 7 7 7 1 1 1 1 1 1 1 1 1 1 1 1 1	BLOCK-DIAG NUMERA1 -2314455-01 5 -336538E400 5 -772855E401 6	.610353E+00 E .433941E+00 E .136400E+01 E .258715E+00 E	<ul> <li>3715296+000 6</li> <li>8LGCK-DIAG.6FF</li> <li>8LDCK-DIAG.6FF</li> <li>8LDCK-DIAG.6FF</li> <li>8LDCK-DIAG.6FF</li> <li>8LDCK-DIAG.6FF</li> <li>8LDCK-DIAG.6FF</li> <li>9LDCK-06860 CO49</li> <li>9LDKD6860 CO49</li> </ul>

.

49988844894489448894949499999994494649946949484848949 Mettidi: Plock-Diagoral Contaigutions to Ng Enticiency Components in Decrepsing Orders:

\*

1036(IN MCJSEL) aL3CK-2145..CCMF..EF=.. 1 .557591E+30 6 .597591E+00 7 .597991E+00 5 .597991E+00 5 .592650E+03 6 .540450E+03 7 .540450E+03 8 .54050E+03 8 .54050E+03

• •

INIT. GEN. CISPLACEMENT

EXAMPLE 4.4

```
-
                                                                                                                                                                     ARE DESIRED, DIMERNISE A DUMMY PARAMETER)=
                                                                                                                    -
                                                                                                                   # d X
                                                                                                                                                                                                                                                                                                           STRUCTURAL MATURAL FREQUENCIES FOR THE EVALUATION MODEL
                                                                                                                   REDUCED PRINCIPAL EFFICIENCY COMPONENTS
                                                                                                                                                                                                                                                   <u>م</u>
                                                                                                                                                                    NDDNUMCAPPLIES DNLY IF NDNCCLLCCATED DUTPUTS
                                                                                                                                                                                                                                                 ŝ
                                        σ
ACOSS-4 STRUCTURE EFFICIENCY ANALYS'S
EVALUATION MODEL GROER NEE 12
                                         ۲Z
                                                                                                                                                                                                                                                                          •
                                                                                                                                                                                                                                                                     JULICUTEU ACTUATORS: 1 2
                                                                                            4
                                      THE CONTROL DESIGN MODEL DROER
                                                                                                                                                                                                                                                                                                                                                                                                                                                                              m
                                                                                                                                                                                                                                 ----IN ROUTINE MANAL------
                                                                                       NUMBER CF RESIDUAL MGDES NR=
                                                                                                                                            m
                                                                                                                                                                                                                                               ~
                                                                                                                                                                                                                                                                                                                                       1.3420152508
                                                                                                                                                                                                                                                                                                                                                                             2.9573860520
                                                                                                                                                                                                                                                                                                                                                                                                                               4.7552603174
                                                                                                                                                                                                                                                                                                                                                                                                                                                       9.2505565555
                                                                                                                                                                                                                                                                                                                                                               2.8906962402
                                                                                                                                                                                                                                                                                                                                                                                          3.3982126302
                                                                                                                                                                                                                                                                                                                                                                                                        4.2044659928
                                                                                                                                                                                                                                                                                                                                                                                                                    4.6620681236
                                                                                                                                                                                                                                                                                                                                                                                                                                            8.5394143164
                                                                                                                                                                                                                                                                                                                                                                                                                                                                     10.2347720461
                                                                                                                                            = WN
                                                                  *
                                                                NUMBER OF INPUTS M=
                                                                                                                                                                                                                                                 --
                                                                                                                                          NUMBER OF DUTPUTS
                                                                                                                                                                                                                                              SELECTED MODES:
                                                                                                                                                                                           4
4
4
1
1
1
1
1
1
                                                                                                                  CROER CF
                                                                                                                                                                                             ...
```

. 200200 10.000. 600000 1 2 2 3 4 5 6 7 8 •000099 •000090 •000090 •000000 •000000 •000000 -

INIT. GEN. VELOCITY

-----LINEAR QUADRATIC REGULATOR CONTRCL DESIGN

CPTIMAL GAIN MATRIX

176-01	105-01	316E-01
1556-01	1715-01	822E-01
1376-01	1595-01	149E+00
2.753966	-2.725964	3.942383
-9.497369	-9.677049	1.306773
4.884757	7.322153	-1.484538
-6.62599131E-01	-6.24042715E-01	-1.467747042-01
-5.13950713E-01	7.20930538E-02	-1.500219225-01
3.56038895E-02	-6.37707224E-01	-1.19721225E+00
4.15960513E-02	1.41864677E-01	1.16612022E-01
8.49677661E-01	-9.14830736E-01	-5.04848164E-02
-5.77145560E-61	-1.08671550E+03	2.72726146E-01
6.32484885E-01	-6.88705489E-01	-1.00471596E+00
-2.10472434E-01	-1.33053134E-01	-4.962823326-01
7.30524475E-01	-6.60005489E-01	7.06328753E-01
-7.032434975-02	5.78972084E-02	1.050308355-01
-3.425297265-01	-8.54849464E-02	-1.324412885+00
4.356099345-01	-5.99372083E-01	7.973414+95-01
2.221044815-01	7.43317951E-01	-3.407965595-01
-	~	m
3 U U	1 Cl Cl	

205

R04	4	2.56983595E-02 2.90177173E-61	-3.24182592E-01 1.77899962E-01	-1.39253492E-01 6.32231335E-01	1.054372735+00 8.87340164E-02	6.61325334E-02 -3.44260384E-01
		-2.10961246E-J1 -9.98572560E-01	-6.79917941E-01	1.35313030E+00	-4.56368529E-01	-1.93950941E+00

CLUSED LDCP EIGENVALUES Real Part Imaginary Part

.333799755+01 -.33979975E+01 -.367374745+00 -.36737474E+00

--47067222E+01 --47256953E+00

--47356953E+00

.47067222E+01

--29523828E+01 -.Z85E5433E+00

• 29523828E+01 -.24845493E+00

-.420234385+31 -.4043939191E+00

--28934289E+31

-.19793909E+00 -.13793909E+00

.28934288E+01

• + 202343EE + 01

-\*+0+39991E+30
-.103755036+00 -.156447536+01 -.102755036+30 ..165647536+31 -.455246386-31 ..134207936+01 -.655246386+31 ..134207366+01 -.85522196-01 -.473970466+01 PICCATI MATRIX

12		243235		.00126		.163940		.023452		1.125577		.135296		653392		143557		444381		152121		.465231	•	2.530260		049193		.213516		.365524		108939		
11		331555		.475649	,	.158401		-1-063471		2.254827		-2.453150		-1.793081		1.452371		-2.345001		.716136		46.1491.84		.465231		201022		.024738		-2.319652		241502		
10		011268		.027537		263264		051067		286-25		113842		.388166		.228979		.497023		2.766136		.715136		152121		1.300555		108793		-1.297594		.054023		
ø		.065234		796730.		293170		1.110346		920305		.421436		2.205239		534099		32.311124		.487023		-2.346001		444391		-1.155045		704729		.522808		. 569190		
80		256923		072172		301655		109474		276079		736010		.480566		3.738799		534099		.228979		1.452371		143567		.665535		061587		-1.790497		.089568		
•		231757		1.270144		472787		.939836		-5.644858		.313565		32.780452		.430556		2.205439		.388156		-1.793081		693392		.006908		234902		.619276		.694065		
ç		297617		009676		390874		038093		.481453		5.411908		.313565		736010		.421436		113842		-2.453150		.195296		2.260618		045232		061785		105739		
Ś		128134		1.376434		318090		1.174428		4.648627		.491453		-5.644858		276079		920305		286425		2.254827		1.126577		661074		884495		-1-217538		.051270		
e	16	.331714	014261	-18091.	054030	.491450	.219006	9.805290	.141798	1.174428 4	.057270	038093	105739	.953636 -	.634066	109474	.089568	1.110846	.669190	061067	.054023	1.063471	241502	.023452	108939	.042355	-3.874402	138532	-5-076340	1.8-5209 -	.350961	.1.1798	7.326759	
°	15	.348236	201287	518601	.283343	6.923734	.579347	.491450	-1.845809	31809J	-1.217598	390374	061783	472787	.619270	301659	-1.790-97	293170	.522405	263264	-1.297594	- 10+851.	-2.319662	.163940	.365524	552361	*****	.006136	4.219799	- 579347 -	oë.742100	.214006	.350961	
r d	14	.494450	157668	15.434752	- 0C8118	518601 i	.003136	.180917	138632	1.376434	834495	003576	045232	1.270144	294902	072172	061587	.057347	704729	.027537	108793	.476649	.024733	.000726	.010615	1.879954	• 4 2 I 3 3 9 0	.003119	7.159651	.298843	1667912.4	050460	-5.076340	NE CLEIGP-
	13	27.599853	152940	. 494450 1	1.879954	.348236	552351	.331714	.042355	128134	661074	297617	2.260519	231757	.006,009	256923	. 565535	.06523.	-1.195046	011268	1.300556	031565	201022	0+9296	049193	152840	155.366303	157666	.421339	201247	*****	014251	-3.674402	ILDÜM NI
		-1		~		~		4		Ś		÷		1		60		ው		10		11		12		13		*1		15		10		

13+2079E+01

-.5552464E-J1 -.1037552E+00

--1664875E+01

CUMPLEX CLOSED-LJOP EIGÉNVALUES -.6552464E-C1 -.1342079E+01

.156~E75E+01	Z358423E+01	10+30298652*	235383E+01	•2952383E+01	3397998E+C1	.33379985+01	4202344E+01	.42023445+01	4706722E+01	.4705722E+01	4709709E+01	-4709709E+01	
1037551€+00	18793315+30	1%79391E+00	28385-65+03	283 <b>8548</b> 5400	3673747E+00	3673747E+00	- <b>.</b> 40439395+00	0C+25868704	4735695E+00	4735695E+00	8669222E-01	3669222E-01	-TH PANTTUR CECU

	.0447058 .097
	400%CAQ.
RGLLED MODES	-0487451
ATIO DE CONTI	- SUITE
DAMPING R	DAMP TNG

.0134040 .1011001. .0957992 .1074837 .0973729 .0647058 •0071334 IN ROUTINE FOR EFFICIENCY COMPUTATIONS •0487651 RATIOS : DAMPING

INITIAL MODAL STATE VECTOR:

11 .0000006+00 .25260365+00 .00 .00000005+0032414245+00 .0000 .0000006+0094027895-01 .0000
11 .000000E+00 .25260 .000000E+003241424 .000000E+009402789
11 .0000000 .0000000 .0000006+

PÉR CENT EFFICIENCIES

EFFICIENCY COERFICIENT:

• 2622305E+01

GLJBAL EFFICIENCY

.3075699E+02

RELATIVE MODEL EFFICIENCY:

• 8065421E+02

---GLEBALLY MINIMUM COST SE++-: 26753996+01

---- 4004L CCST SM---: .70157148+01

---REAL CUST SR---: .30985096+01

GLCBAL SPILLOVER QUOTIENT: .52898855+00

RELATIVE MODEL SPILLOVER QUDTIENT: .1934575E+00 -----In Routine Simcat-----in Routine Ressim ------In Routine Modeff-------Paincipal Components Amalysis of Glopal Spillover Quotient

6

1 761161 7613884 560308 922959 0239592 023815 033815 0359123	- 068785 - 019224 - 019224 - 025889 - 292535 - 292535 - 094115	1 216553 201036 117629 20139402 015930 0159219 20030125 20030125 2003025 2003025 2003025 2003025 2003025 200325 20055 20055 20055 20055 20055 20055 20055 20055 20055 20055 20055 200555 20055 20055 200555 200555 200555 200555 200555 200555 200555 200555 200555 200555 200555 200555 200555 200555 200555 200555 200555 2005555 2005555 2005555 2005555 2005555 2005555 2005555 20055555 20055555 20055555 20055555 20055555 20055555 20055555 20055555 20055555 200555555 2005555555 20055555555
HNM45557000		まってんちょうできょうでしょうできょう

ш U

#### . 301107 . 331987 . 002599

#### 1 N N

CHARACTERISTIC EFFILIENCIES

1	.678933	.703031	.903094	.231669	2.218093	7.309411	.294804	7.611583	4.668067	18.718626	2.258960	18.247127	7.798900	.915139	.003503	•005061
	-1	۰.	m	<b>.</b> #	Ś	Ģ	۰.	თი	6	10	11	12	13	14	15	16

### EFFICIENCY VECTUR

-

.147025	.142334	.106050	.039684	.035111	.030836	.023995	.019987	.01+173	.009530	• 00 5 9 9 5	.002547	-002524	-001068	.000112	.000018	
-1	~	~	•	ŝ	Ŷ	7	80	<b>~</b>	10	11	12	13	14	15	16	

## GLOB..SPG MOCAL MATRIX T

• 65839 TUTAL EFFICIENCY =

•

91       3.243383      072032       1.456         74      720323      035752       1.         51      053333      035752       1.         137      003035       .051496          137      003035       .051496          137      005232       1736413          137      005232       1736413          137      005232       1736413          137      005232       1131550          159      01117      937285          171       .003549       1.104941          171       .003583       .104941          171       .000399       .007791          147       .000339       .007793          147       .003553      033665          148       .007553      033665          148       .003152      013123          157      0364577       .089359	91 3.24398307524 747203230357459 510533330351456 24 .003036 .051496 28012665094103 28012665094103 79 .001117037285 156 .002683 .104941 71 .000867 .010885 167000399007535 147000411 .007535 147009152013123 97086477 .089359	91 3.243383075 74720329077 51053333005 24 .003036 .051 137 .005232 1.736 28012665004 79 .001117037 79 .001117037 14700389007 14700389033 147003152013 14 .009152013 197036477 .089
	2.46.010    01217.4    720329       520403    102351    053035       004.035    0230024     .0030365       14351     .0056037     .005235       002159     .0554350     .005453       012150     .0554350     .0054635       021305     .0554350     .0054635       021305     .0554350     .005463       021305     .0554350     .005463       021305     .0554350     .005463       021305     .056379     .0012663       031117     .005355     .001365       032159     .002335     .000399       01120     .0020335     .000399       01122     .1151347     .000399       01122     .1151347     .000399       01122     .1151347     .000399       01129     .125645     .017352       01129     .125646     .0009152       01129     .125646     .0009152       01129     .019097     .019057       01129     .019097     .019057       01129     .019097     .019057	66.01.0      03277.4      720929         204.04      102351      0533333         04.035      030024       .0050315         04.035      030024       .0053036         04.035      030024       .0053036         04.035      030024       .005232         02237       .0023728      0128665         022159       .054350       .0054593         02159       .054350       .005467         02159       .054350       .005467         02159       .054350       .005467         02159       .054379       .001117         03171       .0023871       .000867         03171       .0023871       .000867         03171       .002320      000339         01129       .125647       .017353         18940       .405684       .009152         04271       .079097      0864777
<ul> <li>.520484102351553383</li> <li>.004035030024 .003035</li> <li>.143514 .055037 .005237</li> <li>.022139 .0027290125653</li> <li>.022159 .055355 .005379 .001117</li> <li>.022159 .005356 .002683</li> <li>.003121 .0028356 .002683</li> <li>.003121 .0028356 .002683</li> <li>.003121 .0028356 .002683</li> <li>.003121 .0028356 .0028356</li> <li>.003122 .125642 .017353</li> <li>.018940 .079097086477</li> <li>.804271 .079097086477</li> </ul>	520484      102351      053333         504035      030024       .005435         143514       .056037       .0055232         143514       .055037       .0055232         143514       .0554350       .0012659         125535       .0554350       .00126583         1170       .0554350       .00126583         102159      0563376       .0026833         102159       .0068356       .0026833         102151       .002831       .0003679         103171       .002835       .0028399         103171       .0028356       .0003399         109122      151347       .001353         118940       .405684       .001355         118940       .405684       .001355	20484      102351      050024       .003036         34035      050027       .005232         43514       .055037       .005232         21305       .005273       .005453         21305       .055337       .005483         21305       .005379       .001117         21305       .055379       .001117         21305       .056379       .001117         21509       .005379       .0012683         2111       .0028376       .0026839         22151       .0028379       .0028399         20112       .002811       .0003867         20112       .01281347       .0003399         20112       .0129320       .017353         20112       .012947       .017353         20112       .0156484       .009152         201220       .0173953       .017353         204201       .079097       .086477
$\begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$	004035      030024       .003036       .0         143514       .056037       .005232       1.1         002237       .002728      012865      0         013059       .002728       .012865      0         02159       .055379       .001117      0         02159       .055379       .001117      0         02159       .055379       .001117      0         03171       .002871       .000867       .0         03171      020320      000399      0         099122      151347       .001353      0         01129       .125642       .017353      0         01129       .125645       .009152      0         018940       .079097      0       .0	0.4035    030024     .003036     .0       4.9514     .056037     .005232     1.1       2.237     .056037     .0012665    0       2.1305     .054350     .0012665    0       2.1305     .054350     .001117    0       2.1305     .056379     .001117    0       2.1305     .058356     .001117    0       2.131     .0026871     .003867     .0       3.2535     .0028379     .0     .0       3.171     .002817     .0     .0       3.2535     .0028379     .0     .0       3.171     .002817     .0     .0       3.171     .0028379     .0     .0       3.171     .0028379     .0     .0       3.171     .0028370     .0     .0       0.125     .151347     .0     .0       0.125     .125642     .017352     .0       1.94097     .0969152     .0     .0       0.4211     .079097     .086477     .0
.143514       .056037       .005222       1.736413         .022237       .002728      012665      004103         .021305       .054350       .012665      014550         .022159       .054350       .005459       1.132560         .022159       .056379       .001117      937285         .022159       .006379       .001117      937285         .032171       .0058356       .0026833       .104941         .032171       .002871       .000399       .007791         .099122      151347       .000399       .007791         .091129       .125642       .017353      033565         .01129       .125642       .017353       .033565         .018940       .079057       .0313123         .018940       .079097       .089359	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	<pre>43514 .056037 .005232 1.736 22237 .002728012865004 21365 .054350 .005459 1.131 22535 .008356 .002683 .104 32535 .008356 .002683 .104 32535 .008356 .002683 .004 03171 .002871 .000867 .010 03171 .002871 .000867 .010 03122 -151347000411 .007 09122 -151347009151 .033 18940 .405684 .009152013 18940 .079097086477 .089</pre>
02237       .002728      012665      004103         21305       .054350       .005457       .131550         02159      006379       .001117      937285         02159      006379       .001117      937285         02159      006379       .001117      937285         02159      006379       .001494       .010867         03171       .002871       .000399       .010865         03171      020320      0003399       .007791         03122      151347      000349       .007535         01129       .125642       .017353      033665         18940       .405684       .009152      013123         18940       .079097      086477       .089369	02237       .002728      012665      004103         21305       .054350       .0026579       .001117      037285         02159      006379       .001117      037285         02159      006379       .001117      037285         02159      006379       .001117      037285         02151      000367       .010865       .017991         03171       .002871       .000339       .0077535         03171      020320      000339       .0077535         03122      151347      000349       .0077535         03122      151347      0017353       .033665         01129       .125642       .017353       .0333659         18940       .405684       .009152      013123         04271       .079097      086477       .089369	02237       .002728      012865      004         21305       .054350       .005453       1.131         02159      006379       .001117      937         02159      006379       .001117      937         02159      006379       .001117      931         02159      008356       .002683       .104         03171       .002871       .002867       .010         03171      020320      000399      010         03172      151347      000411       .007         09122      151347      0004153      033         01129       .125642       .017353      033         18940       .405684       .009152      013         04271       .079097      086477       .989
1305       .054350       .0054350       .001117      037285         2159      006379       .001117      937285         2535       .008356       .001117      937285         3171       .002871       .002683       .104941         3171       .002871       .002867       .010885         3211      002320      000339      007791         9211      0220320      000339      0333665         9212      151347      017353      0333665         1129       .125642       .017353      0333665         8940       .405684       .009152      013123         4271       .079097      086477       .089359	1305       .054350       .0054350       .00117       .111050         2159       -0068356       .001107       .104941         2171       .0028351       .002683       .104945         3171       .0028379       .0010857       .104945         3171       .0028379       .0010865       .010885         3171       .002330       .0010865       .007791         9212      151347       .0003399      0077535         9122      151347       .0003153      033653         9122       .125642       .017353      0313123         9240       .079097      086477       .089359	1305       .054350       .0054359       .001117       .131         2159       -006379       .001117       .037         2151       -006376       .002683       .010         3171       .002871       .000399       .001         3171       -020320       -000399       .0010         9211       -020320       -000399       .007         9122       -151347       .000399       .003         9123       -125642       .017353       .033         9440       .405684       .009152      013         8940       .405684       .009152       .013         4271       .079097      086477       .089
<ul> <li>2159006379 .001117037285</li> <li>22535 .008356 .002683 .104941</li> <li>2171 .002871 .000867 .010885</li> <li>19121020320007535</li> <li>19122151347000411 .007535</li> <li>1129 .125642 .017353033665</li> <li>1129 .125642 .017353013123</li> <li>18940 .079097086477 .089359</li> </ul>	212159      006379       .001111       .037285         124535       .008356       .0028356       .0028356       .0028356         149401       .002811       .000867       .010885         19171       .002817       .000867       .010885         19171       .002810       .000856       .010865         19122       .151347       .0009152       .0336535         21123       .125642       .017353       .013123         18940       .079097      089152       .013123	12159    006379     .001117    937       12535     .008356     .002683     .104       12171     .002871     .001867     .010       19211    002320    000399    007       19222    151347    0003411     .007       19129     .125642     .017353    033       18940     .405684     .009152    013       18940     .079097    086477     .989
32535       .008356       .002683       .104941       .         03171       .002817       .000867       .104945       .         09211       .0028320       .0003399       .01785       .         09212      151347       .0003399       .007793       .         09122       .125642       .017353       .037535       .         18400       .125642       .017353      033665       .         18940       .405684       .009152      013123       .         18940       .079097      086477       .089359       .	32535       .008356       .002683       .104941         03171       .002811       .000867       .010885         03211       .002320      000399       .007917         09212      151347       .000347       .007791         09122      151347       .007953       .017353         01129       .125642       .017353      0333665         18940       .405684       .009152      013123         18940       .405684       .009152       .013123         04271       .079097      086477       .0893569	32535 .008356 .002683 .104 03171 .002871 .000867 .010 09211020320000399007 09122151347000411 .007 01129 .125642 .017353033 18940 .405684 .009152013 18940 .079097086477 .089
003171 .002871 .000867 .010885 - 003121020320000339007791 009122151347000411 .007535 . 601129 .125642 .017353033665 . 018940 .405684 .009152013123	<pre>171 .002811 .000867 .010885 19120 -020320000863 191200203200007535 209122151340009153033663 2009152013123 208940 .009152013123 204271 .0790970386477 .089369</pre>	03171     .002871     .000867     .010       09211    020320    000399    007       09122    151347    000411     .077       01129     .125642     .017353    033       18940     .405684     .009152    013       18940     .079097    086477     .089
.009211020320000399007791 .0091221513470073553037535 . .601129 .125642 .017353033665 . .018940 .405684 .009152013123	19711020320000399007791 20912215134700051 -007535 601129 -125642 -017353033665 218940 -405684 -009152013123 218940 -079097036477 -089359	<b>19211020320000399007</b> <b>09122151347000411 .037</b> <b>01129 .125642 .017353033</b> <b>18940 .405684 .009152013</b> <b>18940 .079097086477 .089</b>
.009122151347000411 .007535 . .401129 .125642 .017353033665 . .018940 .405684 .009152013123 .804271 .079097086477 .089359	009122151347000411 .007535 601129 .125642 .017353033665 518940 .405684 .009152013123 804271 .079097036477 .089359	09122151347000411 .007 01129 .125642 .017353033 18940 .405684 .009152013 04271 .079097086477 .989
.401129 .125642 .017353033665 . .018940 .405684 .009152013123 .804271 .079097086477 .089359	601129 .125642 .017353033665 318940 .405684 .009152313123 904271 .079097086477 .989369	01129 .125642 .017353033 18940 .405684 .009152013 04271 .079097086477 .089
.018940 .405684 .009152013123 .804271 .079097086477 .089359	318940 .405684 .009152313123 804271 .079097036477 .989359	18940 .405684 .009152013 04271 .079097086477 .989
clicit lettou. *80eur. Ustalu. . 804271 . 079097 - 086477 . 989369 -	calciu- action. *80cu+. 0.84216 804271 .079097086477 .989369	
<b>.804271 .079097086477 .089369</b>	804271 <b>.</b> 079097086477 .989369	94271 •079097 -•086477 •989.

UMERATOR COMPONENT COSTS:

•681859E+00	.3924176+00	
•123610E +00	.232456E+90	104273E-01
.351563E-02	.169545E+00	.8944976-02

ENDMINATCR COMPONENT CESTS:

.199259E+00 .210419E+00			.154247E+01 .297593E+00
.249939E-01 .359370E-01 .237609E-02			7
.6110436401 .1142045401 .1142045401 .5548626405 .2548626405 .12429386403 .1498076-03 .1938676-01 .1938776-01 .1938777777777777777777777777777777777777	);;;; ;; ;; ;; ;; ; ; ; ; ; ; ; ; ; ; ;	598630E+00 598630E+00 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1	1 :
.15100.5-01 .4455635+00 -4455635+00 -76535915-02 .452045501 .452045501 .452045501 .3337495501 -3337495501 -333749501 -333974501 -333974501 -333974501 -33397501 -333975050 -333975050 -333975050 -333975050 -333975050 -333975050 -333975050 -333975050 -333975050 -333975050 -333975050 -333975050 -333975050 -333975050 -333975050 -333975050 -333975050 -333975050 -333975050 -3339750 -3339750 -3339750 -3339750 -3339750 -3339750 -3339750 -3359750 -3359750 -3359750 -3359750 -3359750 -3359750 -3359750 -3359750 -3359750 -3359750 -3359750 -3359750 -3359750 -3359750 -3359750 -3359750 -3359750 -35597500 -35597500 -35597500 -3559750000000000000000000000000000000000	<pre>\$</pre>	<pre>2356096-02 280576096-02 1498676-02 1546966-0 608842660 608842660 609842660 6098328660 117409660 117409660 11936660</pre>	UMCONTROL CUS ErdmControl C Ffictencis : Ffictencies : Ffictencies :
.5339796-03 .5339796-03 .2112086+00 .22506860.00029 .13146986-02 .13146086-02 .13146086-02 .1314608602 .3343426-02 .0008668602049.65 .0008668602049.65 .0008668602049.55 .0008668602049.55 .0008668602049.55 .00086686020049.55 .00086686020049.55 .00086686020049.55 .00086686020049.55 .00086686020049.55 .00086686020049.55 .00086686020049.55 .00086686020049.55 .00086686020049.55 .00086686020049.55 .00086686000000000000000000000000000000	A C C C C C C C C C C C C C C C C C C C	NP TH JRDER EFFICI BLOCK-DIAG NUMERAT .11.0076-03 6 .11.00006+01 6 .1000006+01 6 .4369946+00 6 .2934176+00 6 .3474036+00 6 .1332036-02 6	ULOCK-CLIGGEFFN BLOCK-CLIGGEFFD UPOROERCD COMPE UNDROERCD COMPE UNDROERCO COMPE

BLGCK-DIAG..CCMP..EFF.. .7473736+01 \$11281E+01
.396906=+01
.164247E+01
.164247E+01
.141501ê+01 •746562E+01 .716285E+CO .297593E+00 MODECIN MODSEL) 00 m + m 00 **.8728782+00** 

-----IN ROUTINE CLEIGP-----

COMPLEX CLOSED-LOOP EIGENVALUES -.5721622E-01 -.1345591E+01

.1345581E+01	15620015+01	.1662001E+01	2911917E+01	.2911917E+01	2919084E+01	*2919084E+01	3307799E+01	.3307799E+01	4164775E+01	.4164775E+01	4707505E+01
- <b>-</b> 5721622E-01	8396655E-01	3396655E-01	17238316+J0	1723801E+33	2777074E+30	2777074E+00	6455285E-01	6455285E-01	6137389E-02	6137389£-02	-,4697455E+00

.0.35555 **•**6392934 .001-735 .0135116 .0947075 .0590947 .0504570 UAMPING RITIC OF CONTROLLED MCDES .0+2+832 CAMPING RATIOS :

-| |

1

----IN ROUTINE EFFNCY-----

++707505E+01

-.4697455E+00 -.17-1002E-01 -.17+1002E-01

-.4762697E+01 +762697E+01

IN ROUTINE FOR EFFICIENCY COMPUTATIONS

INITIAL MODAL STATE VECTCR:

6214141E-01	- 0033000E+00	<ul> <li>2526036E+90</li> </ul>	. 000000E+00
35195326+30	.000000000.	32414246+30	•0000000E+30
<b>5878361</b> ë-U1	• COODODOE + DO	94327896-01	• 000000E+00

.00030036+30 .00000366+00

-\$243160E+00 -\$4082741E+00

PER CENT EFFICIENCIES

ÉFFICIENCY COEFFICIENT:

.1787869E+01 GLOBAL EFFICIENCY

•4088485E+02

RÊLATIVE MODEL EFFICIENCY:

----

7317851E+02

---GLOBALLY MINIMUM COST S#---: .47292936+01

---MODAL COST SM---: .8464813E+01

---REAL COST S9----: .1156735E+02

GLOBAL SPILLOVER QUOTIENT: .6560253E+00

RELATIVE MODEL SPILLDVER QUGTIENT: .26821496+00 -----in Routime Simdat-----in Routime Ressim -----im Routime Modeff-----principal Compoments analysis of Global Spillover Gugtient

04

.52763 .52763 .52763 .6633653 .593971 .593971 .593971 .5572426 .154163 .154163

~~~~

### EFFICIENCY VECTOR

ł

|  | CHARACTERISTIC EFFIC:<br>2.371737<br>2.371737<br>2.371737<br>2.205943<br>1.293920<br>1.3933048<br>1.293920<br>1.3933048<br>1.293928<br>1.293928<br>2.173209<br>2.094150<br>2.094150<br>2.15506<br>2.12556<br>2.0024908 |
|--|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|--|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|

#### 214

с С

-.010771 -.036119 -.036576 -.069109 -.069109 -.069109 -.005463 -.188869 -.201901

4014444 4014444 4014

| - | -139544 | .1296-1 | .1278)3 | .105259 | .064253 | .031655 | .027013 | .010545    | .013041 | .005922 | .005586 | .002478 | .000100. | .0000. | .000035 | .000032 |  |
|---|---------|---------|---------|---------|---------|---------|---------|------------|---------|---------|---------|---------|----------|--------|---------|---------|--|
|   | -1      | ~       | ~       | Ŧ       | \$      | Ŷ       | ~       | <b>6</b> 0 | 6       | 13      | 11      | 12      | 13       | 4      | 15      | 15      |  |

TOTAL EFFICIENCY = .65603

GLOB..SPQ MCDAL MATRIX T

| 11 12    |    | .916270 .005120     | .85630502580* |                 | .222010019346 |         | .89030910532. |               | •00442414398J |         | .003321 1.927765 |         | .000525197335 |         | .018647 .329274 |         | .001180 001295 |          | .012993 -1.133792 |         | .000040062991 |         | -000304041231 |        | .000928 .002131 |         | .008552022351 |          | .232374 .025942 |         |
|----------|----|---------------------|---------------|-----------------|---------------|---------|---------------|---------------|---------------|---------|------------------|---------|---------------|---------|-----------------|---------|----------------|----------|-------------------|---------|---------------|---------|---------------|--------|-----------------|---------|---------------|----------|-----------------|---------|
| 10       |    | 626899 2            | 3.781289      | •               | 675310 -      |         | 433346        |               | 001958 -      |         | 003798           |         | 013343 -      |         | 003967          |         | .002308        |          | -009818 -         |         | - 36L000.     |         | - 076000.     |        | .005206         |         | .925269 -     |          | 017523 -        |         |
| 6        |    | 642463              | -018749       |                 | 371180        |         | 2.543671      |               | 049499        |         | 036331           |         | .000546       |         | 010311          |         | .003408        |          | 052923            |         | 001776        |         | 008235        |        | 045540          |         | .051224       |          | . 692276        |         |
| 80       |    | .013869             | 020733        |                 | .005378       |         | 045592        |               | 300535        |         | .738126          |         | 081626        |         | 1599951         |         | 002651         |          | 1.824139          |         | 008852        |         | .147181       |        | .006322         |         | . 302003      |          | .033090         |         |
| 2        |    | 094594              | 1-447740      |                 | 1.889303      |         | .733758       |               | .009233       |         | .071967          |         | .008304       |         | .001045         |         | 001071         |          | 004379            |         | 00000         |         | .009854       |        | .001324         |         | .012059       |          | £ 5 1 C O O .   |         |
| Ş        |    | 007433              | 034253        | )<br> <br> <br> | .012336       |         | .043930       |               | .143685       |         | .011196          |         | 131170        |         | -048494         |         | .322509        |          | .202497           |         | .015736       |         | -1.113358     |        | 611076          |         | 041948        |          | 015029          |         |
| ŝ        |    | 003271              | 017641        |                 | .013042       |         | 028304        |               | 029322        |         | 338085           |         | 029039        |         | .461558         |         | .240797        |          | 314799            |         | -017874       |         | 1.457609      |        | 002382          |         | 020415        |          | 600357          |         |
| Ł        | 16 | .011549<br>- 000400 | -019467       | 035800          | 043311        | .000375 | 026705        | • 0 4 3 3 I 3 | 034240        | .004237 | 1.011451         | 120436  | .398654       | 003699  | -1-0-0548       | .091793 | .119961        | • 003362 | .212002           | 059480  | 038679        | 002444  | .194582       | 062231 | .017162         | .270809 | 004119        | 535373   | .015088         | .269927 |
| £        | 15 | 011861              | 044945        | 007527          | .007132       | 010736  | 097420        | .012117       | 308020        | .027423 | 314572           | .000887 | .37-997       | 020647  | 1.103213        | 000651  | .022672        | .012626  | 523343            | 003167  | .003712       | .015737 | 539119        | 01+791 | 007919          | .111228 | .061171       | 1.303483 | 027589          | .043755 |
| 7        | 14 | 001222              |               | .457151         | 004793        | 033442  | .015368       | .002071       | 018573        | .014118 | .124524          | 075741  | -016947       | .004374 | 065201          | e11300. | 038199         | -003592  | .03-431           | .021445 | .227194       | .000818 | 013575        | 015143 | 003161          | 010905  | 341703        | 279700   | .010391         | 027533  |
| <b>-</b> | 13 | 029874              | -002239       | .117295         | 067244        | -270160 | .107252       | -1.969749     | .650148       | .008155 | .705651          | 088961  | .176699       | .002030 | .748233         | 039153  | 037112         | .002870  | .308926           | 011047  | 001112        | 000239  | .141231       | 024325 | 004503          | 053346  | 012738        | E3EE50.  | .051845         | .891142 |
|          |    |                     | ~             |                 | m             |         | 4             |               | \$            |         | Ŷ                |         | ~             |         | <b>00</b>       |         | ¢              |          | 10                |         | 11            |         | 4             |        | 13              |         | 14            |          | 15              |         |

04

| 1<br>-1.995067 | 1.780715 | -1.499371 | 1.164361 | .374718 | .307945 | .428973 | 270057 | .288056 | 274486 | .168407 | .180523 | .222543 | 097206 | 236711 | 169031 |
|----------------|----------|-----------|----------|---------|---------|---------|--------|---------|--------|---------|---------|---------|--------|--------|--------|
| 1              | 7        | m         | 4        | ŝ       | Ŷ       | ~       | 80     | •       | 10     | 11      | 12      | 13      | 14     | 15     | 10     |

•

| 1<br>34065<br>274129<br>194091<br>17204 | .012139<br>.018198<br>.015908<br>.006309<br>.1790.<br>.01119                                | .002452<br>.002817<br>.002817<br>.004281<br>.004949<br>.002470 |
|-----------------------------------------|---------------------------------------------------------------------------------------------|----------------------------------------------------------------|
|                                         | 1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1 | - ~                                                            |

## 6/4700.

.... ..... ..... .....

CHAPACTERISTIC EFFICIENCIES

1 •265735 •243493

-- ~

ພ ບ

| 12          | .915242            | 001153            | •003534           | .01-265           | • 000 + 55         |
|-------------|--------------------|-------------------|-------------------|-------------------|--------------------|
| 11          | 003824             | 347959            | .005327           | 333745            | .135776            |
| 10          | .039665            | 049948            | 069050            | 013696            | 016300             |
| 6           | .283230            | .1771.96          | .070708           | 339542            | . Co9259           |
| <b>6</b> 7) | .237124            | .107186           | .003356           | .502351           | 012670             |
| 2           | 142500             | • 399825          | .214075           | .135650           | .023920            |
| \$          | •017714            | 010350            | <103ED.           | 002874            | .036700            |
| s           | .07¢975            | 303056            | 2-1005-           | 025690            | 039069             |
| 4<br>16     | .020799<br>.005521 | 029525            |                   | 016505<br>.014338 | .001195<br>.023048 |
| 3<br>15     | -019983<br>001200  | 036463<br>025322  | 02753C            | 013527<br>-030314 | 004291<br>.000195  |
| 2<br>1<br>1 | 050042<br>015893   | 105507<br>.057205 | 048371            | .176387           | .403930            |
| 1<br>13     | 021665<br>.017312  | 040852<br>.028581 | JISIL9<br>.010983 | .053331<br>035956 | 054858<br>.004421  |
|             | -4                 | 7                 | <b>~</b> 1        | t                 | ŝ                  |

REL.SPG MCDAL MATRIX T

.26821 TÜTAL EFFICIENCY =

217

CT VECTOR

|                               | EFFICIENC |
|-------------------------------|-----------|
|                               |           |
| 1450 0 P. 17 0 P. N. 9 4 10 9 |           |

.090524 .065748 .065748 .053634 .031249 .00153536 .001535 .0015395 .0012195 .001182 .001182 .0001182 

.

| 501144. 040316 04024 -014918 023933 535719 113445055328016997 737353103573 -002388<br>7 0005441 -0014914 -019555 -018954 -013914 -019575 -011275<br>8 -011221149502 -012194 02354 -01165<br>9 -0045456 003303012194 023555 -01165<br>9 -011321 -010323 -001315 -01165<br>9 -011321 -010323 -01165<br>9 -011321 -010323 -012109 -101615<br>9 -011321 -010323 -003938 -01165<br>9 -011321 -010323 -003938 -01165<br>9 -011321 -010323 -003938 -01165<br>9 -011321 -0104205 -003932 -01165<br>9 -010312 -014319 -012205 -009933<br>10 -011321 -0104205 -004933 034713 -090044 115239 -026712 054724 -031441 -021549 -0113473<br>9 -004545 -004403 -011025 -009324 050154 -010104 03272 -011239 -026712 054724 -031441 -021549 -0113473<br>10 -004549 -001035 -009324 050156 000753 -0085175 036655 -0384023 -0117409 0119196 -002481 -011323<br>11 -000494 -001035 -001935 0001255 000753 -0081151 007667 -008114 001769 -017637 -008105<br>11 -000494 -001035 -001935 000753 -008125 00753 -085175 0316070 -1126059 -017632 -008105<br>10 -00332 -011952 -003265 -0019354 000753 -0081121 007667 -008577 -001145 017637 -008105<br>10 -00332 -011952 -001269 -0019073 -012649 -0071014 001557 -0061121 007657 -008577 -001145 007639 -018409 -011952<br>10 -003345 -014903 -012265 -000753 -000753 -0081121 007657 -008577 -001145 007639 -007539 -007539<br>10 -003345 -004190 -012265 -000752 -0001421 007555 -000453 -013171 003354 -001354 -001754<br>10 -00548 -003345 -004392 -004972 -001421 001557 -000453 -013279 -003347 -003345 -003939 -017649<br>10 -00548 -003345 -004392 -004972 -004423 -012794 003345 -003793 -009549<br>10 -00548 -003345 -004392 -004972 -000453 -012799 -003497 -003349 -003939 -017649<br>10 -003345 -004392 -004972 -00443 -013872 -000453 -003497 -003346 -003939 -003939 -003939<br>10 -00548 -003548 -012549<br>10 -005648 -003548 -012390 -012390 -012390 -01249 -003345 -003345 -003939 -017649<br>10 -003882 -003345 -003345 -004972 -000453 -012178 -003497 -003346 -0033945 -003939 -017649<br>10 -005648 -003548 -003948 -010395 -00449 -01249 -0034914 -003346 -003939 -017649 -003939 -003939 -017649<br>10 -005648 -00348 -004929 -004929 -00044 |         |         |         |         |         |         |         | 0       | .268215E+0 | 0MP.:   | AFTER DEC  | CIENCY CHECK |             |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------|---------|---------|---------|---------|---------|---------|---------|------------|---------|------------|--------------|-------------|
| 5       -001545       .000525       -001801       .000525       -101817       .002308       .012175       .0023085       -101817       .0023085       .011275         7       .001525       -101817       -001801       -011814       .011377       .0023085       .011275       .011275       .011275       .011877       .0023085       .011275       .011275       .011275       .011275       .011275       .011275       .011275       .011275       .011275       .011275       .011275       .011275       .011275       .011275       .011275       .011275       .011275       .011271       .012100       .011275       .011373       .001377       .001377       .001372       .011373       .001377       .001372       .011371       .001377       .001372       .011373       .001377       .001371       .011373       .001377       .001371       .001371       .001371       .001371       .001371       .001371       .001371       .001474       .001371       .001474       .001371       .001877       .001877       .001877       .001877       .001877       .001871       .001877       .001877       .001877       .001877       .001877       .001877       .001877       .001877       .001875       .0018775       .001877       .0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |         |         |         |         |         |         |         |         | 573E+02    | .115    | ROL CCST : | -DENCHCONT   | •<br>•<br>• |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |         |         |         |         |         |         |         |         | 254E+01    | .310.   | RCL CGST : | NUMCONT      |             |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |         |         |         |         |         |         |         |         |            |         | E ENCOSEL- | IN ROUTIN    | -           |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |         |         |         |         |         |         |         |         | . 318396   | 322993  | .423927    | .072112      |             |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | .987692 | .003865 | .005704 | 057386  | 037112  | .002873 | 021708  | -01030- | +00600.    | .004332 | 115498     | 058812       | 16          |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |         |         |         |         |         |         |         |         | .067074    | .191350 | .202908    | .000458      |             |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 676060  | 003933  | .035584 | .003217 | .003497 | .019279 | 000453  | .038822 | .020774    | .041410 | 002057     | .000617      | ξŢ          |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |         |         |         |         |         |         |         |         | .848721    | 526387  | 334349     | • 34 C 5 7 4 |             |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 017243  | .018403 | 073476  | 033339  | .092191 | 035559  | .016936 | 047355  | 040392     | 053974  | 063250     | 03385.       | :           |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |         |         |         |         |         |         |         |         | 114190     | .154893 | 003-95     | . 310520     |             |
| <ul> <li>011414 06.0316 06.0524078318 .028933 .535713 .133452065328016977 .737353103273 .002386</li> <li>0052410148390188035011253</li> <li>.003042012194012194013641 .013377 .009929 .245467050851 .0564314 .315040011275</li> <li>005245012194 .001355015124013611 .013377 .009929 .245467050851 .056434 .315040011275</li> <li>005242012194 .001355015129 .013545013164 .0130292 .013412043219024647051649 .013745</li> <li>01137 .071321149502012194 .001354 .013104 .030292013239026712 .054284034611021549 .013743</li> <li>011437 .07333 .039317 .009205 .001205 .001204 .030292013239026712 .054284034614021549 .013473</li> <li>018549074634022104140137013044 .152330041293 .098121 .1685660888706013916</li> <li>018549074634041205 .0029235 .0320144 .152330041293 .098121 .1685660888706018731</li> <li>006496011092 .260304010325 .0056126 .0022605 .028695 .308623317409 .019196002461018231</li> <li>0064960110366012305001025 .0021590085175 .035705126053018549003584066105</li> <li>110064960110366013269000753085175 .035705126059076324003584066105</li> <li>120044960120860132690090753085175 .035705126059003584066105</li> <li>13004496012086013269003753085175 .035705126059076324005584066105</li> </ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | .070498 | .003150 | .020945 | 901145  | 009657  | .037667 | 006121  | 007357  | .012245    | .012837 | 01357L     | 006177       | <b>ب</b>    |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |         |         |         |         |         |         |         |         | 019044     | 043504  | .084315    | 1.072572     |             |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 066105  | 003584  | 076323  | 126053  | .036705 | 085175  | 000753  | .021594 | 049073     | 032459  | 100366     | 004798       | 12          |
| $ \begin{array}{rcrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |         |         |         |         |         |         |         |         | .010566    | 003505  | 018064     | .003342      |             |
| <ul> <li>5011404 004010 004054078418 028933 .535719 .134452065328016947 .7373531032713 .002388</li> <li>7 .0026491014839048095015253</li> <li>7 .0504910137770016914019841 .013377 .009929 .045467050851 .056414 .315040011275</li> <li>7 .050491012054012194019841 .013377 .009929 .045467050851 .056414 .315040011275</li> <li>8 .071221149502012194 .083541079575 .9437891061700134120432199436299043693 .003745</li> <li>9 .011221149502012194 .0116197010104 .0330292013239026712 .054284024361021548 .013473</li> <li>9 .1665.E .0545180221001401977010104 .0330292013239026712 .054284034961021548 .013473</li> <li>9 .1665.E .0545180221001401977010104 .0330292013239026712 .054284024361021548 .013473</li> <li>1 .1655.E .0545180221001401977010104 .0330292013239026712 .054284024361021548 .013473</li> <li>1 .1655.E .0545180221001401977010104 .0330292013239026712 .054284024361021548 .013473</li> <li>1 .1655.E .0545450078635 .009823 .0347130990044 .152930041293 .098121 .1685660888956013916</li> <li>1 .1685450746340742650098824</li> </ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 019231  | 002481  | .019196 | 317409  | .308623 | .028695 | .002505 | .006126 | 010325     | .260304 | 011032     | 006-96       | 11          |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |         |         |         |         |         |         |         |         | 009824     | 043265  | 07+63+     | 018549       |             |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 019916  | 088836  | .168556 | .098121 | 041293  | .152930 | 0060    | .034713 | .09833     | .033817 | .163033    | 210644.      | 10          |
| <ul> <li>501141 .04.0316 .04.0524078d18 .028933 .535719 .133452065328016977 .737353103273 .002886</li> <li>7 .00164101893088035015124</li> <li>7 .05049113774030691015414013861 .013377 .009929 .145467050851 .056414 .315040011275</li> <li>7 .05049113774000691015414013861 .013377 .009929 .145467050851 .056414 .315040011275</li> <li>7 .050491137740200535015124 .013981 .013377 .009929 .145467050851 .056434 .315040011275</li> <li>7 .051210012194 .073551073575 .943789110170013412043219436299942693 .003745</li> <li>7 .013213 .005210140197010104 .030292013299022109022109012174</li> <li>7 .1665.E .054214022100140197010104 .030292013299026712 .054244022104</li> </ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |         |         |         |         |         |         |         |         | .009436    | .001205 | . 301-10   | 038275       |             |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | .013473 | 021548  | 034961  | .054234 | 026712  | 013239  | .030292 | 010104  | 140197     | 022100  | . J6:518   | .1605.5      | ÷           |
| <ul> <li>5011414 .04.316 .040524078318 .028933 .535713 .133452065328016977 .737353103273 .002386</li> <li>0.012641016335018253</li> <li>0.029549 .156467050851 .015253</li> <li>0.09929 .164667050851 .0156467011275</li> <li>0.09224013844013841 .013377015124 .0139431 .013347 .0039439 .013412043219042219042219 .003745</li> <li>0.071221143502012194073551079575 .943789110412013412043219042619 .003745</li> </ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |         |         |         |         |         |         |         |         | -0116-     | .006938 | 626670.    | 011437       |             |
| 5011414 .04.0316 .04.0524078418 .028933 .535718 .133452065328016977 .737353103273 .002888<br>.001641014833088045015253<br>7 .050493137745030091014914013871 .013377 .009929 .245467050851 .056414 .315040011275<br>005236 .033077300535015124                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | .003745 | 742693  | 436209  | 043219  | 013412  | 100170  | .943789 | 079575  | .083541    | 012194  | 143502     | .071221      | ••          |
| 5011414. 04.0316 .040524078418 .028933 .535718 .133452065328016947 .737353103273 .002888<br>.001641014833088045015253<br>7 .050493137743000091014914013871 .013377 .009929 .245467050851 .056414 .315040011275                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |         |         |         |         |         |         |         |         | 015124     | 300535  | 770£C0.    | 005236       |             |
| 50114j4 .04.316 .04.0524078418 .028933 .535718 .133452965328016947 .737353103273 .002888<br>.001641014833088045015253                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 011275  | .315040 | .056314 | 050851  | .245467 | •299CO. | .013377 | 019841  | 014914     | 030091  | c+7761     | .950493      | ۴-          |
| 50114i+ .04.316 .04.052407841E .028933 .535718 .134452065328016947 .737353103273 .002888                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |         |         |         |         |         |         |         |         | 015253     | 088045  | ee8-10     | -001041      |             |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | .002388 | 103273  | .737353 | 016947  | 065328  | .133452 | .53571ª | .029933 | 073318     | .0+0524 | .045316    |              | r           |

.

UMĒRATOR COMPONENT COSTS:

|             |                            | .3172076-02                 |
|-------------|----------------------------|-----------------------------|
| .102929E+01 | .149263E+JO<br>.72230F5+DO | •176321E-01<br>- 3878445400 |
| 10.00105.01 | 1 4 0 1 4 2 5 4 3 0        | 1742216-01                  |

ENCMINATOR COMPONENT COSTS:

|                                                                                                                                                           | .911355-01<br>.6001065-01                                                                                                                                               |                                                                                                                                                                                                               |                                                                                                                                                                                 | .257655E+00<br>.276435E+00                                                                 |
|-----------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|
|                                                                                                                                                           | .109012E-01<br>.622201E-01<br>153390E-03                                                                                                                                |                                                                                                                                                                                                               | + E + 0 1                                                                                                                                                                       | 1905+01<br>.4008936+00<br>.2660542+00<br>.4427326+00                                       |
| .4023295+01<br>.2554495+U1<br>.)/DENCM.) :<br>.8899320E-01<br>.607836E-01                                                                                 | -143243E-02<br>-342372E-01<br>-177610E-63<br>-177610E-63<br>-177610E-63<br>-6176ER:                                                                                     |                                                                                                                                                                                                               | : .216084                                                                                                                                                                       | :T : ~ 7911<br>.3833946+00<br>.2624792+00<br>.4379366+00                                   |
| <ul> <li>3937685+30</li> <li>2674155+01</li> <li>2991065-03</li> <li>5741450495-01</li> <li>1290395-01</li> <li>5243435-03</li> <li>3160155-03</li> </ul> | FICIENCIES :<br>FICIENCIES :<br>FICIENCIES :<br>FICIENCIES :<br>6 CONTRIGUTION :<br>NT 5, IN DECREASIA                                                                  | ######################################                                                                                                                                                                        | R & DENDMINATD? (<br>• 122923E-01<br>• 387384E+00<br>• 286307E+01<br>• 352527E+01<br>• 181239E+01<br>• 246836E+01<br>• 390180E-02<br>• 547422E-02                               | HJMCONTRCL CJS<br>Ficiencies :<br>Ficiencies :<br>Ficiencies :                             |
| • 5781435-91<br>• 1567096+91<br>• 1567096+91<br>• 3493026-02<br>• 3493026-02<br>• 31524306-02<br>• 31524306-02<br>• 31524306-03                           | UNDADERED CGMP., EFI<br>UNDRDERED CUMP., EFI<br>UNDRDERED CUMP., EFI<br>UNDRDERED CCMP., EFI<br>UNDRDERED CCMP., EFI<br>METHOD 1: INTERACOMO:<br>FFF., NUMER., COMPONDI | MDDECIN MODSEL) CO<br>MDDECIN MODSEL) CO<br>3<br>6<br>6<br>1<br>1<br>7<br>1<br>7<br>1<br>7<br>1<br>7<br>1<br>7<br>1<br>7<br>1<br>7<br>1<br>7<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1 | BLDCK-FLAG NUMERATG<br>.477424E-02 E<br>.155300E+30 E<br>.737684E+00 E<br>.925309E+00 E<br>.483790E+00 E<br>.483790E+00 E<br>.483790E+00 E<br>.2423625-62 E<br>BLUCK-FLAG.EFFNU | BLGCK-DIJG.EFFJC<br>Unopdered Compe<br>Undpcred Compe<br>Undpcred Compe<br>Undrgered Compe |

#### • EXAMPLE 4.5

```
---
                                                                                                                                                                    ARE DESIRED, DTHERWISE A DUMMY PARAMETER) =
                                                                                                                    NP= 16
                                                                                                                                                                                                                                                     12
                                                                                                                                                                                                                                                                              11
                                                                                                                   DROER JF REJUCED PRINCIPAL EFFICIENCY COMPUNENTS
                                                                                                                                                                                                                                                     2
                                                                                                                                                                        NJONUM(APPLIES ONLY IF NJNCCLLOCATED JUTPUTS
                                                                                                                                                                                                                                                      •
                                      œ
ACONS-4 STRUCTURE SEFICIENCY ANALYSIS
Evaluation Mouse Crosp No= 12
                                                                                                                                                                                                                                                       ÷
                                        H
Z
                                                                                           4
                                        THE CONTROL DESIGN MOVEL ORDER
                                                                                           NUMBER OF RESIDUAL MEDES NR=
                                                                                                                                                                                                                                       SELECTED MODES: 3 4 5
                                                                                                                                               m
                                                                                                                                                                                                                                                                                                                                                                                                               9.2535565555
10.2947720451
12.9051041170
                                                                                                                                              11 2 2
                                                                                                                                                                                                                                                                                                                                             2.8905362402
                                                                                                                                                                                                                                                                                                                                                             2.9573860520
                                                                                                                                                                                                                                                                                                                                                                         3.3982125302
                                                                                                                                                                                                                                                                                                                                                                                       4.2044653926
                                                                                                                                                                                                                                                                                                                                                                                                                                                     1.3420152508
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                4.5520581235
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             4.7552503174
                                                                                                                                                                                                                                                                                                                                                                                                  8.5394143164
                                                                                                                                                                                                                                                                                                                                                                                                                                                                   1.5647171077
                                                                  4
                                                                NUMBER OF INPUTS MH
                                                                                                                                               NUMBER CE DUTPUTS
                                                                                                                                                                                                                           . . . . .
                                                                                                                                                                                                  4L 14L
14- 14L
14- 14-
```

INIT. CEN. DISFLACEMENT

-----INAINC BAILICE WI-----

u.

| <pre>* * * * * * * * * * * * * * * * * * *</pre> | <ul> <li>0000000</li> <li>0000000000</li> <li>00000000</li> <li>00000000</li> <li>000000000</li></ul> | <pre></pre>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | το, υσούος, στορόος, στορόσος, το το το το το 2<br>το, υσούος, στορόος, στοτάσος, το | rr<br>.030360 .936350 .000600 .000003 .000000 .000003 .000050 .000<br>- | N9IS3C 1C2   |            | 52E-01 -2.80569598E-01 9.25633383E-01 -3.25321251E-01<br>125E-01 7.36502745E-01 -1.254235106+00 -7.55558854E-01<br>17.5-31 1.227525776+03 7.90924189E-31 -2.41622456E+00 | 05E-02 4.1517848E-03 -9.3038T209E-01 1.68746798E-01<br>136E-01 -5.63847524E-01 6.39466185E-01 -6.44731471E-01<br>135E-01 3.06922359E-01 8.74983137E-01 -2.31270652E+00 | 57E+OD -2.34~63DD7E-D1 -6.02992932E-D2 -1.82739575E-01<br>4.77596579E-D1 -7.38572668E-O1 -7.21963751E-01<br>198E-D1 -5.38230565E-O1 -5.46382547E-01 3.72538001E-01 | 952-01 1.951153275-01 6.540140735-01 -1.614914615-02<br>1332-02 -7.13310963E-01 3.900511365-01 6.62702595E-01<br>1142-01 7.040312553-01 5.364216342+01 -6.83599431E-01 |                    |           |           |           |           |           |           |           |            |  |
|--------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------|--------------|------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------|-----------|-----------|-----------|-----------|-----------|-----------|-----------|------------|--|
|                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 2000000<br>INNIT. 1<br>INNIT. | , 100000.                                                                                                                |                                                                         | LATOR CONTRO |            | -3.17423452<br>2.28662525<br>5.47322574                                                                                                                                  | -6.98635505<br>-6.03411036<br>-8.74953135                                                                                                                              | -1.34192157<br>6.45067482<br>-3.54990298                                                                                                                           | 2.37371495<br>-9.32705733<br>4.08567314                                                                                                                                | ARY PART           | 1998CE+01 | 10+308661 | 217315+31 | 10+315+01 | 25853E+01 | 25863E+C1 | 30075E+01 | 300796+01  |  |
|                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 1<br>                                                                                                                    | 1<br>- 303050<br>- 1 N 2JUTI                                            | CALE SEAN    | GAIN MATRI | 3.532<br>-5.594<br>-9.588<br>-9.588                                                                                                                                      | -1.278<br>-9.513<br>-1.808<br>-9.775                                                                                                                                   | 2.413<br>1.128<br>-3.525<br>1.095                                                                                                                                  | -3.342<br>-3.573<br>5.823<br>-2.053                                                                                                                                    | JCP EIGENV<br>Part | 10170E+00 | 101706+00 | 235345+30 | 235945+00 | 64217E+03 | 642175+30 | 733586+00 | 1733545+00 |  |

| 324374255401 | .924074255+31 | 1029+3726+22 | +13284372E+32 | 1230505~E+02            | .123050546+02 |
|--------------|---------------|--------------|---------------|-------------------------|---------------|
| 3512=3326+33 | 3>1253025+00  | 25233345600  | 2623334432+30 | - <b>-11</b> 483373E+C3 | 11+3337GE+00  |

#### RICCATI PATRIX

| 12         | 035779          | 005974             | 216935                | 023518           | .043302             | 024276            | 206337                  | 001514             | -2.050287          | 243329              | .477960            | 2.853674            | 2-035249           | 115632               | 319382               | .027119                                |                       |
|------------|-----------------|--------------------|-----------------------|------------------|---------------------|-------------------|-------------------------|--------------------|--------------------|---------------------|--------------------|---------------------|--------------------|----------------------|----------------------|----------------------------------------|-----------------------|
| 11         | .165752         | .369560            | 364232                | 2.102630         | 577888              | 297305            | . 455893                | .952705            | 19-056383          | 2.415128            | 50.032803          | 77360               | 10.369152          | -1.633440            | 3.839482             | .143558                                |                       |
| 10         | 159378          | 032310             | 010062                | .028740          | 205544              | 001553            | .373502                 | .033333            | -481156-           | 3.087895            | 2.4161282          | 243329              | -1.371201-         | .014757              | 2.055914             | 023564                                 |                       |
| 6          | 854223          | 1.584187           | .163093               | .513947          | .242315             | 1.241331          | .280437                 | -1.471684          | 26.543433          | • + 81156           | 19.056383          | -2.050287           | 1.278353           | .351454              | -1.305269            | 905736                                 |                       |
| 30         | 1.037914        | .178658            | 677376                | 130370           | 4 4 3 5 3 9         | 152793            | .476031                 | 2.533578           | -1.4716342         | .033333             | .952706-1          | 001518              | 566506*-           | 005672               | .190885 -            | 000202                                 |                       |
| 1          | 2.240407        | -2.364056          | -1.648511             | 1.396139         | -2.228359           | ** 1212.          | 45.970159               | .476001            | .280+37            | .373502             | <b>455834</b>      | 206337              | 872802             | .154423              | .462730              | 027232                                 |                       |
| Ŷ          | 239360          | 125959             | .388464               | .212020          | .492727             | 2.112522          | .712144                 | 152733             | 1.241331           | 001563              | 291305             | 024276              | -2.093529          | .018752              | 2.092738             | 030310                                 |                       |
| Ś          | -1.121638       | • 434535           | 2.022917              | 527956           | 12.151713           | . 492727          | -2.228359               | 448599             | .242315            | 2055-4              | 677838             | .043302             | .198077            | 726622.              | 219740               | 147140                                 |                       |
| Ł          | 261735 -        | 007333<br>697901   | 002434<br>.490157     | 001500           | .003258<br>527956 3 | 147143            | 030310<br>1.396109 -    | 027232             | 090202             | 905796<br>.023780   | 023564             | .143569<br>029518   | .027119<br>.242838 | 3.176752 312440      | .046478<br>022543    | .464595<br>.003258                     | a.733923              |
| m          | 15<br>5.653935  | :34972<br>.300509  | .184759<br>2.453365   | .227502          | 022543              | 213740<br>.388454 | Z.0927J9<br>1.643511    | 677375             | .190845<br>.753099 | -1.305259<br>370052 | 2.055014           | 3.839492216935      | 319382<br>093773   | 3.235686<br>019334   | -4.360436<br>.227502 | 44444444444444444444444444444444444444 |                       |
| ~ <b>4</b> | 14<br>•431369 - | 047339<br>5.301563 | 006387<br>.300508 3   | 019334<br>597901 | 012440<br>.434585   | 125959            | - 019752<br>2.364055 -  | .154423<br>.17365à | 006572<br>1.534137 | .951454<br>032310   | .014757<br>.369560 | -1.689440<br>306574 | 116582<br>.595571  | .498719<br>005387    | 3.847935<br>.184759  | -4.960495*<br>002434                   | -046474<br>25 CLEIGO- |
|            | 193237          | 197281<br>.481359  | .5353736<br>-5.053936 | 261735           | -1.121683           | .198377<br>239350 | -2.083529<br>2.240467 - | 872302<br>1.057914 | 905955<br>854223   | 1.27835d<br>169976  | -1.371201          | -10.369152          | 2.0962+9           | 408.937721<br>0+7330 | 138572               | 3.235636<br>007633                     | 3.176752<br>IN RGUTI  |
|            | 1               | 2                  | ~                     | ÷                | 5                   | ÷                 | *                       | ro                 | 6                  | 10                  | 1:                 | 12                  | 13                 | 4                    | 15                   | 15                                     |                       |

COMPLEX CLSSED-LDDP EIGE4VALUES -.13973366+30 -.28980096+01 .2698038E+31 -.29525866+31

-.1397336E+CO -.2386422E+0J

:

| 2330+22E+JJ                                | .29525965.                                                                                                      | 104                                       |           |                           |              |                    |                        |          |
|--------------------------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------------|-----------|---------------------------|--------------|--------------------|------------------------|----------|
| 3571017E+90                                |                                                                                                                 | 101                                       |           |                           |              |                    |                        |          |
| 3571317E+00                                | .3347936.                                                                                                       | 104                                       |           |                           |              |                    |                        |          |
|                                            |                                                                                                                 | 10,                                       |           |                           |              |                    |                        |          |
|                                            | . 4202173≣.                                                                                                     | 10.                                       |           |                           |              |                    |                        |          |
| 32957765+30                                |                                                                                                                 | -01                                       |           |                           |              |                    |                        |          |
| 32357765+00                                | . 954145SE                                                                                                      | -01                                       |           |                           |              |                    |                        |          |
| 3512330E+90                                | 92437436                                                                                                        | -01                                       |           |                           |              |                    |                        |          |
| 351233GE+00                                | .92487435                                                                                                       | •01                                       |           |                           |              |                    |                        |          |
| - * 2 5 2 3 9 4 E + 0 C                    | 10254375                                                                                                        | • 0 2                                     |           |                           |              |                    |                        |          |
| Z623334E+00                                | 1 .1026437E                                                                                                     | .02                                       |           |                           |              |                    |                        |          |
| 11483376+00                                | 129056                                                                                                          | • 5 2                                     |           |                           |              |                    |                        |          |
| 114337E+00                                 | .1270505E                                                                                                       | • 0 2                                     |           |                           |              |                    |                        |          |
| RCUTINE EF                                 | = F NC Y                                                                                                        |                                           |           |                           |              |                    |                        |          |
| DAMPING RATIC OF CONT                      | IRJLLEO MGDES                                                                                                   |                                           |           |                           |              |                    |                        |          |
| DAMPING RATICS :                           | •0+23305 •0                                                                                                     | 972953 .10740                             | 97 .09575 | 49 .03                    | 85570        | .0379543           | .0255061               | •00800*  |
| IN ROUTINE FOR EFFIC                       | LIENCY COMPUTATION                                                                                              | 2                                         |           |                           |              |                    |                        |          |
| INITIAL MOGAL STATE                        | VECT32:                                                                                                         |                                           |           |                           |              |                    |                        |          |
| .62491675+00<br>40827415+00<br>30333755+00 | <ul> <li>• 0 0 0 0 0 0 0 2 + 3 0</li> <li>• 0 0 0 0 0 0 0 0 2 + 0 0</li> <li>• 0 0 0 0 0 0 0 2 + 0 0</li> </ul> | 3513532E+0<br>4250273E+00<br>.4117159E+00 | 000000    | 000E+00<br>0E+00<br>05+00 | 324<br>44833 | 1424E+00<br>81E+00 | .0000000<br>.0000005+0 | 0<br>• 0 |
| PER CÊNT EFFICIÊNCIË                       | S                                                                                                               |                                           |           |                           |              |                    |                        |          |
|                                            |                                                                                                                 |                                           |           |                           |              |                    |                        |          |
| EFFICIÊNCY CJEFFICI                        | 1 E N 1 :                                                                                                       |                                           |           |                           |              |                    |                        |          |
| .3465399E+31                               |                                                                                                                 |                                           |           |                           |              |                    |                        |          |
| GLJBAL EFFICIENCY                          |                                                                                                                 |                                           |           |                           |              |                    |                        |          |
| .13273196+32                               |                                                                                                                 |                                           |           |                           |              |                    |                        |          |

RÉLATIVE ADDEL EFFICIENCY:

•9721561E+J2

---51244LFY MINIMUM COST 54---1 ...28133576+02

GLG3AL SPILLOVER QUDTIENT: .27101665+00

RELATIVE 40DEL SPILLEVER QUOTIENT: .27833926-01 -----IN 2JUTIVE SIMDAT------IN RGJTINE RESSIM ------IM RQUTINE MGDEFF------PRINCIPAL COMPONENTS AMALYSIS OF GLOBAL EFFICIENCY

0.4

|                                         | ພ<br>ບ                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|-----------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| • 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 | 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~  | までで、10 mm - 10 mm - 1 |

#### . 393055 . 903074 . 563346

\* 5 6

| FEICLENCIES |             |          |         |         |         |         |         |         |         |         |         |         |               |         |        |
|-------------|-------------|----------|---------|---------|---------|---------|---------|---------|---------|---------|---------|---------|---------------|---------|--------|
| 3 011010101 |             |          |         |         |         |         |         |         |         |         |         |         |               |         |        |
| 5 T T S     | <br>.635079 | . 527250 | .231343 | -16524. | .522693 | .701955 | .155542 | .551227 | .053631 | .335633 | .589351 | .156319 | • 5 4 9 5 5 4 | .230556 | .33525 |

**のちゃをとすの**をらんでちゃをとす

### SFFICIENCY VECTOR

| -4 | .044535 | .014335 | .014620 | -009479 | .004545 | .003716 | .003493 | .002693 | .061770 | .001521 | - 300974 | .060191 | .000134 | -000036 | .00001 | .000015 |  |
|----|---------|---------|---------|---------|---------|---------|---------|---------|---------|---------|----------|---------|---------|---------|--------|---------|--|
|    | 1       | ~       | m       | 4       | ~       | 9       | -       | 8       | •       | 10      | 11       | 12      | 13      | 14      | 15     | 16      |  |

## GLOB.. EFF..430AL MATRIX T

l i i

ī

.13279 TJTAL EFFICIENCY =

| 12 |    | 201256           |               |          | 126500.                 |           | 0+3355   |                 | .056335 |         | 12021-  |          | ******       |         | 1.275339 |         | 002959    |               | .924652  |         | .002579  |         | 025434   |         |         |         | 000251  |                 | 000064  |         | • 000139 |          |            |            |                   |             |
|----|----|------------------|---------------|----------|-------------------------|-----------|----------|-----------------|---------|---------|---------|----------|--------------|---------|----------|---------|-----------|---------------|----------|---------|----------|---------|----------|---------|---------|---------|---------|-----------------|---------|---------|----------|----------|------------|------------|-------------------|-------------|
| 11 |    | .053370          | 00.0000       | 601010*I | 629225.                 |           | 103257   |                 | .019024 |         | .07510. |          | 166100.      |         |          |         | 000848    |               | .001154  |         | .003605  |         | 001695   |         | 141106. |         | .013364 |                 | 000043  |         | 000599   |          |            |            |                   |             |
| 10 |    | +<0000           |               | CCT005+  | .000050                 |           | 000042   |                 | .00000  |         | 678000. |          | 200000.      |         |          |         | 210000.   |               | .000735  |         | -000012  |         | 000400   |         | 000013  |         | ESE100  |                 | .009642 |         | .571804  |          |            |            |                   |             |
| Ŧ  |    | 003128           |               | 00.00    | 014410                  |           | .151557  |                 | .3533-9 |         | .072709 |          | . 320793     |         | 155527   |         | .033148   |               | 008579   |         | 003807   |         | .024433  |         | .000156 |         | 002173  |                 | 000046  |         | 000149   |          |            |            |                   |             |
| æ  |    | .3335 <b>6</b> 5 |               | .11911.  | 012516                  |           | 050343   |                 | .032244 |         | 007059  |          | 004510       |         | .90.1.2  |         | .000587   |               | 001717   |         | 000230   |         | 000791   |         | .000128 |         | .001075 |                 | 000036  |         | .000443  |          |            |            |                   |             |
| 1  |    | .01120.          |               | .10520.  | 193510                  |           | 033745   |                 | 031648  |         | 22+524  |          | .306327      |         | 135427   |         | 000582    |               | .013352  |         | .000353  |         | 015944   |         | 000632  |         | .000050 |                 | .000053 |         | .000632  |          |            |            |                   |             |
| ÷  |    | 901232           |               | 001767   | 226000-                 |           | 029815   |                 | 007677  |         | .025590 |          | • 00 • • 5 3 |         | .031142  |         | .061389   |               | 120580   |         | .009523  |         | 1.095259 |         | 003955  |         | 034071  |                 | 000032  |         | .001837  |          |            |            |                   |             |
| ŝ  |    | .02230           |               | 1E1410   | 76100                   |           | £62000.  |                 | .034279 |         | 001331  |          | 002904       |         | 022153   |         | .009953   |               | 1.035957 |         | .043830. |         | .034039  |         | 902709  |         | -050319 |                 | .00000  |         | 003425   |          |            |            |                   | 00          |
|    | 15 | 000463           | C 7 6 7 2 0 . | .010015  | - 001014<br>- 001014    | 510520°   | .031224  | 1.055563        | 201009  | 025216  | 000336  | •05264   | .000899      | .004937 | .000543  | .034246 | 002712    | .000375       | .012535  | 030135  | .003512  | .000505 | .035203  | -016245 | 151650. | 003685  | 125214  | - • 0 3 4 9 2 7 | .000293 | -000103 | 001026   | 000331   |            | 1375+02    | 134E+03           | 1027026+(   |
| ñ  | 51 | 142100-          | 731137        | 00000-1  | - 60341413<br>- 6034443 |           | E46100 - | .005235         | .003536 | 000215  | .010358 | 037331   | 000321       | 000035  | 015+32   | .001213 | 013922    | 031254        | 507225   | 32231+  | .123130  | 033311  | 087254   | .032511 | 003313  | .0110.  | .035542 | 4.024571        | .000153 | .000124 | .000874  | .0059500 |            | .2813      | .2739             | 3MP.:       |
| ~1 | 1. | 00277,           | .02051        | 007915   |                         |           | .000320  | 0 <b>.</b> 63ð1 | 002548  | 021642  | 023541  | 1.193247 | 000945       | .945152 | .509164  | .093399 | .120562   | <b>716100</b> | 093163   | .010605 | .003511  | 000715  | 473456   | 041525  | .0.5254 | .000341 | .026279 | .012345         | 000142  | 000053  | 000423   | 004564   | EMCDSEL    | 101 COST : | 10L COST :        | AFTER DECC  |
| -1 | 13 | .300042          | 035333        | .000507  | • 445,994<br>• 100,000  | - CC 2222 | 000 - 51 | 091513          | ****    | .602132 | 003016  | .007300  | 000013       | 001534  | 10100.   | 019915  | 6 70000 * | .6003.        | 000127   | 106500. | 00000-   | .000279 | 000055   | 001393  | 000102  | 000273  | .000027 | .001032         | .052146 | 000100  | 124654   | 030273   | IN ROUTINE | NUM CONTR  | <b>JENDHCONTR</b> | CENCY CHECK |
|    |    | -1               |               | 7        | ~                       | •         | Ł        |                 | 5       |         | ç       |          | ~            |         | Ŧ        |         | ~         |               | 1,       |         | 11       |         | 12       |         | 13      |         | ÷       |                 | 15      |         | 16       |          |            | EFF        | EFF               | EFFICI      |

UMERATOR COMPENENT COSTS:

| .444647E+33 | .634329E+01 |                                |
|-------------|-------------|--------------------------------|
| .311729E+00 | .433715E+01 | <ul> <li>126J815+J2</li> </ul> |
| .622520E+00 | .1190506+01 | .2555756+01                    |

ENDRINATOR COMPONENT COSTS:

| .95572dE-03                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                        |                                                                              | • 5505295+00<br>• 506333E+00                                                                  |
|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------|------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|
| .734449E-03<br>.974094E-03<br>.974094E-02                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                        |                                                                              | 7±+02<br>4245+03<br>•328550±+03<br>•448622€+03<br>•538126E-01                                 |
| .7511536+00<br>.1242396+02<br>.152706%J%.) :<br>.1622906-02<br>.2207536-01<br>.1932986-02<br>.2531796-02<br>.5415296-02                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 0000000000000000000000000000000000000                                      | 102702E+00                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | CO49. COSTS:                                                           |                                                                              | T : .28136<br>.257 : .272<br>.157621E+00<br>.691295E+00<br>.282296E+00                        |
| <pre>.120351E+01<br/>.35471E+01<br/>.35471E+01<br/>.2044.004<br/>.2044.004<br/>.113776E-02<br/>.153030E-01<br/>.153030E-01<br/>.153030E-01<br/>.153030E-01<br/>.153030E-01<br/>.1510ECTES<br/>.1510ECTES<br/>.1510ECTES<br/>.1510ECTES</pre>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | LUGOGACCECCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCC                                 | MPEFE<br>669594E-01<br>1336735-01<br>974094E-02<br>6415295-02<br>19321955-02<br>1932195-02<br>19537285-03<br>1944495-03<br>3697286-03<br>1944495-03                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 7214MIMUT812<br>3949475+01<br>39488025+0<br>69075735401<br>10+351012   | <pre>.1.1.1.2.1.2.4.0.<br/>.119335240.<br/>.905377240.<br/>.234236540.</pre> | H+CONTROL COS<br>NomControl C<br>Ffictencies :<br>Ffictencies :                               |
| <pre>4.252025401<br/>4.15555401<br/>4.1515555401<br/>4.252420.4<br/>4.245156402<br/>4.245155402<br/>4.345155402<br/>4.345155402<br/>4.345155402<br/>4.345155402<br/>4.345155402<br/>4.345155402<br/>4.345155402<br/>4.345155402<br/>4.345155402<br/>4.345155402<br/>4.345155402<br/>4.345155402<br/>4.34515540<br/>4.34515540<br/>4.34515540<br/>4.34515540<br/>4.34515540<br/>4.34515540<br/>4.34515540<br/>4.34515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515540<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.3515560<br/>4.35155600<br/>4.35155600<br/>4.3515560000000000000000000000000000000000</pre> | ¢\$J\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$ | MODECIN MODSEL) CO<br>8 8<br>5 5<br>7 7<br>1 5<br>5 5<br>5 5<br>5 5<br>5 5<br>5 5<br>5 5<br>5 5<br>7 7<br>7 7<br>8 2<br>8 0<br>8 1 0<br>7 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 | BLCCK-DIAG NUMERATC<br>.622520E+00 5<br>.311729E+03 6<br>.444647E+00 6 | - 435715E+01 E<br>.435715E+01 E<br>.604829E+J1 E<br>.255575E+01 E            | 9LGCK-DIAS.EFFN.<br>8Ljčk-DIAS.EFF3E<br>UNGRJERED CQMP5<br>UNGRJERED CQMP5<br>UNGRJERED CQMP5 |

\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*

MQDECIN MJDJELJ BLJCK-DIAJ..CCMP..EFF.. 5912958+00 5505298+00 5 5505298+00 5 55059336+00 5 55059336+00 5 2828650 1 28228650 1 282285650 1 575216400 1 575216400 1 28267 BLJCK-DIAJ.EFFICIENCY ENP :

0 4

| 1<br><559676<br>553262<br>515792<br>515792<br>315792<br>315795<br>551625<br>551625<br>551625<br>551021<br>1139703<br>1139703<br>083512<br>083612 | 4<br>4<br>4<br>4<br>5<br>1<br>1<br>1<br>2<br>5<br>1<br>1<br>2<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5 |
|--------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|
| 4×××××××××××××××××××××××××××××××××××××                                                                                                           | •••••••••••••••                                                                                                      |

•

u C

| EE0790. | - E E 2 O C - | .205770 | • 00000 • | 170000. | · 20202 · |
|---------|---------------|---------|-----------|---------|-----------|
| 11      | 12            | 13      | 1 •       | 15      | 15        |

## CHARACTERISTIC EFFLCIENCIES

| L |          |     |     |     |     |    |     |     |     |     |          |     |     |     |     |     |
|---|----------|-----|-----|-----|-----|----|-----|-----|-----|-----|----------|-----|-----|-----|-----|-----|
| 2 |          |     |     |     |     |    |     |     |     |     |          |     |     |     |     |     |
|   |          |     |     |     |     |    |     |     |     |     |          |     |     |     |     |     |
|   |          |     |     |     |     |    |     |     |     |     |          |     |     |     |     |     |
|   |          |     |     |     |     |    |     |     |     |     |          |     |     |     |     |     |
|   | -        | -   | 9   | m   | -1  | ,  | ~   | ÷   | 4   | ~   | <b>س</b> | 0   | ç   | e   | (JA | ~   |
| • | <br>TJ d | 194 | 835 | 550 | 111 | 44 | 038 | 634 | 185 | 799 | 635      | 669 | 555 | 663 | 533 | 546 |
|   | .97      | 36  | .97 | .*  | 99  | 66 | 56  | 67  | 98  | 86  | 56       | 66  | 35  | 86  | 5   | -   |
|   | ·        | -   | ·   | •   | ·   | •  | •   | •   | •   | •   | •        | •   | •   | •   | •   | •   |
|   |          |     |     |     |     |    |     |     |     |     |          |     |     |     |     |     |
|   | -4       | ~   | m   |     | 5   | ÷  | ~   | ~   | •   | 0   | _        | ~   | ~   |     |     |     |
|   |          |     |     |     |     |    |     |     | -   | -   |          | -   |     | -   |     |     |

### EFFICIENCY VECTOR

| -4 | • 645651 | .110524 | .076175 | .042661 | .030135 | .013567 | .013295 | .012637 | .009776 | • 009558 | .003905 | .002377 | .0007+2 | .000053 | • 0000 • 5 | + 20002+ |  |
|----|----------|---------|---------|---------|---------|---------|---------|---------|---------|----------|---------|---------|---------|---------|------------|----------|--|
|    | -4       | ~       | m       | 4       | Ś       | •       | ~       | no      | Ţ       | 01       | 11      | 12      | 13      | 14      | 15         | 51       |  |

# TJTAL EFFICIENCY = .97217

REL.. EFF..4204L 44721X 7

|        | 1           | 7                        | ñ       | t          | Ś        | ç       | 7        | æ       | ¢,       | 10      | 11      | 12      |
|--------|-------------|--------------------------|---------|------------|----------|---------|----------|---------|----------|---------|---------|---------|
|        | 61          | 14                       | 15      | 16         |          |         |          |         |          |         |         |         |
| 4      | 0000        | 314305                   | .034543 | .002305    | .015475  | 066854  | .015752  | .272357 | . 303463 | 025960  | 112951  | 723335  |
|        | .12,735     | .005163                  | 055759  | .012423    |          |         |          |         |          |         |         |         |
| ~      | 013205      | 007733                   | .005749 | 096254     | 015703   | 055916  | .032822  | .163409 | .034740  | 019593  | .395751 | .2115.  |
|        | .033386     | 036764                   | .175415 | 5 16 20 0° |          |         |          |         |          |         |         |         |
| ٣      | .015323     | 053342                   | .011725 | .003706    | .079038  | 105149  | .047459  | 123028  | .023390  | 033003  | 053517  | 0-2913  |
|        | .262530     | .131643                  | .030403 | .014356    |          |         |          |         |          |         |         |         |
|        | J2397d      | 321536                   | .925653 | 101628     | 157275   | 043702  | .093474  | 114457  | .102953  | 520844  | .735975 | .320927 |
|        | .196272     | 052225                   | 359305  | .000500.   |          |         |          |         |          |         |         |         |
| 5      | 011351      | .023432                  | 037353  | .012731    | 074156   | .154124 | 015639   | 002019  | -90106-  | .053207 | 123247  | .252652 |
|        | .125153     | 647203                   | 531735  | .211037    |          |         |          |         |          |         |         |         |
| \$     | -07370-     | .263677                  | 029902  | 075291     | 135347   | .78746- | 253839   | .003336 | 113978   | .321109 | 11733   | 573506  |
|        | . 436347    | 143000.                  | 011030  | .021630    |          |         |          |         |          |         |         |         |
| 2      | .027713     | C07519                   | 0.3230  | 035544     | .032555  | 059619  | .00.3359 | .623419 | 038705   | .221333 | .033935 | -121040 |
|        | 036832      | .069753                  | 063971  | 003958     |          |         |          |         |          |         |         |         |
| æ      | .129825     | 245335                   | .058232 | .022239    | .209875  | .532012 | .134177  | .203155 | .066425  | 336423  | 053143  | .155547 |
|        | 350334      | .973677                  | 1050102 | 024744     |          |         |          |         |          |         |         |         |
| Ŧ      | .012452     | 004375                   | 306597  | .130054    | 302072   | .001239 | .009190  | 001525  | 030937   | 006523  | .015212 | 000512  |
|        | 677700"     | 035977                   | 16950.  | 036207     |          |         |          |         |          |         |         |         |
| 1.3    | .087092     | 280111                   | .079035 | .052355    | 097637   | 099430  | 010701   | 018203  | .053612  | .9552-5 | .019918 | 00011   |
|        | 034565      | 027545                   | 017034  | 1.110865   |          |         |          |         |          |         |         |         |
| 11     | 023775      | 753915                   | .025373 | .022080    | 016333   | •008400 | .007893  | .000671 | .098655  | .931541 | .003523 | 004311  |
|        | 001550      | 009753                   | 003393  | 033010     |          |         |          |         |          |         |         |         |
| 12     | 333259      | .907569                  | 151347  | .283555    | 050524   | 066780  | 083520   | .007727 | .479437  | .050752 | 1586+C. | 021591  |
|        | .010933     | .325917                  | .011523 | .215526    |          |         |          |         |          |         |         |         |
| 13     | 063939      | .015510.                 | .002115 | 000597     | . 357624 | .029523 | .054433  | 000984  | .009218  | 006839  | .000951 | 003755  |
|        | 000073      | 036489                   | .0000.  | .003119    |          |         |          |         |          |         |         |         |
| 14     | 012333      | .056602                  | 013236  | 673036     | 562753   | 003975  | .9114.51 | .000999 | 141187   | .202265 | 017159  | 10261.  |
|        | 005201      | .122723                  | 033938  | 012585     |          |         |          |         |          |         |         |         |
| 15     | .0.5574     | .014413                  | .018038 | 000137     | 033155   | 001985  | .000623  | -000110 | .013050  | 000037  | 031350  | .003567 |
|        | 000235      | 003130                   | .00241  | 002215     |          |         |          |         |          |         |         |         |
| 16     | 222562      | • 0 8 9 9 0 <del>6</del> | .616535 | .019645    | .909323  | 002463  | 02000    | 001175  | 164356   | .0098~5 | .034219 | .007484 |
|        | .004518     | .032871                  | 000198  | • 000338   |          |         |          |         |          |         |         |         |
|        | IN RJUTIVE  | JESCCHE                  |         |            |          |         |          |         |          |         |         |         |
| EFF    | .NUM CON TR | IJL COST :               | . 266   | 358E+03    |          |         |          |         |          |         |         |         |
| EFF(   | 05404CONTR  | COST :                   | . 273   | 9845+03    |          |         |          |         |          |         |         |         |
| EFFICI | LENCY CHECK | AFTER DECC               | JMP.:   | •972166E+( | 00       |         |          |         |          |         |         |         |

UMERATOR COMPONENT COSTS:

| .739383E+00<br>.121029E+02                        |  |
|---------------------------------------------------|--|
| <b>\$\$9113€+00</b><br>•895129€+31<br>•229473€+03 |  |
| .2429836+01<br>.1930995+01<br>.9251105+31         |  |

ENDMINATOR COMPONENT COSTS:

•

•

| .28+509-2-02<br>.4413495-01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                | 00+357266.<br>00+357266.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| .375775E-02<br>.333473E-02<br>.833473E-01<br>.834933E+03                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                | E + 03<br>2 4 E + 00<br>• 9 4 6 9 5 9 8 E + 00<br>• 9 4 8 5 9 5 E + 00<br>• 9 7 8 5 5 9 E + 00                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| <ul> <li>7411556+00</li> <li>1242365-02</li> <li>2882966-02</li> <li>4417386-01</li> <li>59241396-01</li> <li>59241396-01</li> <li>3345966-01</li> <li>3345966-01</li> <li>10 146 NP</li> <li>10 146 NP</li> <li>10 146 NP</li> <li>10 146 NP</li> </ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | •********                                                                                                                      | COMP. COSTS:<br>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| .123515+31<br>.954715+31<br>.9547715+31<br>.92494.0024960251<br>.32491596-02<br>.32491596-01<br>.3231595+100<br>.3231596-01<br>.3231596-01<br>.3231596-01<br>.3231596-01<br>.3231596-01<br>.3231596-01<br>.3231596-01<br>.3231596-01<br>.3231596-01<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3231596-00<br>.3331596-00<br>.3331596-00<br>.3331596-00<br>.3331596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311596-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00<br>.3311500-00000000000000000000000000000000 | # # # # # # # # # # # # # # # # # # #                                                                                          | <pre>4 C. DENIMINATIOR<br/>* 3949475491<br/>* 3949625401<br/>* 3496025401<br/>* 372315401<br/>* 1133355401<br/>* 1133355401<br/>* 1133355401<br/>* 1133355401<br/>* 1133255400<br/>* 1133255400<br/>* 1133255400<br/>* 11352<br/>* 1135</pre> |
| 1       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4       4                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | MODECIN MCDSEL) CC<br>MDDECIN MCDSEL) CC<br>8 66 9999999<br>7 7<br>7 7<br>7 7<br>8 7<br>7 7<br>8 7<br>7 7<br>8 7<br>7 8<br>7 8 | aLJCK-DIAG NUMERATG<br>-2442696401 E<br>-2442696401 E<br>-7945296400 E<br>-19663796401 E<br>-3269396401 E<br>-1105976401 E<br>-2292736403 G<br>-2292736403 G<br>ELJCK-DIAG EFFDH<br>UNGRDERED COMPEFF<br>UNGRDERED COMPEFF                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |

susciagymasusussussansusussansususususus Methodi Block-diajonal Contributing TJ 4P Efficiency Compovents IN Decreasing Croesi LCCK-JCLY HCDSEL) BLCCK-JCLi, CCHP..5FF.. 7 .983536400 5 .973575400 6 .973575400 8 .978595400 9 482637400 5 .9482637400 7 .9482637400 5 .9482637400 7 .9482637400 7 .9482637400 7 .9482637400 7 .9482637400 7 .9482637400 7 .9035966400 7 .7197655400

N° IA CROER BLCCK-DIAG.EFFICIENCY ENP : . .5733435+00 15-05-33+UCLP+ CE, MITCIM5, 0.332×LNS.

#### EXAMPLE 6.3

L

やいしんじょく の目れいに対しょう かねぞうひねい シストロストロン ロード・メータン・ロントロン きょうしゅう しゅうしょう しゅうりょう

... 141141 THE CONTAGE LESION MIDEL

.1 AUADER DE L'PUTS ME うそうきがる いっついま うてつうていせん しつ おりのアゴを

16 " a Z

```
.
ŝ
```

|           |                                        |                                          | 9 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0           |                                           |                                         |                                             |                                  |                        |                                         |             |
|-----------|----------------------------------------|------------------------------------------|---------------------------------------------------|-------------------------------------------|-----------------------------------------|---------------------------------------------|----------------------------------|------------------------|-----------------------------------------|-------------|
|           | いっておうしのい                               | 11                                       |                                                   |                                           |                                         |                                             |                                  | (N+2)1622.1            | 00422723.E                              | 0045757     |
| 7 ~ 7     |                                        |                                          | -2-474400<br>-1-4-561-00<br>+5-97278-00           | -1.44795+59<br>-1.52395+71<br>-2.30712+71 | 1.25315+30<br>4.44345+00<br>2.22316+00  | -5.2214E+90<br>-5.2214E+90                  | -3,33455+JJ                      | 1-23055+01             | 1.43225+21                              | 2.53566+31  |
| # 14<br>1 | -=5555+00<br>-=7262+00                 | 10+32456+1+                              | -1.13575+01<br>2.74518+00                         | -1.5635E+31<br>-2.3290E+99                | 1.4031E+J1<br>-1.15555+00               | 1.55235+01<br>-1.59535+00                   | 6.5117E-01                       | -1.54855+01            | 5.43926-01                              | -1.75596+00 |
| 1 1 10    |                                        | -4.32235+33<br>-1.+4502+31<br>2.34422+31 | 3, 30376-01<br>3, 94736-00<br>4,05246+00          | -1.73345+31<br>1.5+125+31<br>9.10225+50   | -4.3170E+00<br>1.82325+00<br>1.53C5E+01 | -5.5750 6+00<br>+9.23556 +00<br>2.5607 6+01 | -4,9329£+00<br>- *               | 1.93555+01             | -1.04492+31                             | -1.75586+01 |
|           |                                        |                                          |                                                   |                                           |                                         |                                             | •                                |                        |                                         |             |
|           |                                        | 14                                       | NICCATI MATRIX                                    |                                           |                                         |                                             |                                  |                        |                                         |             |
|           | -                                      | 1 2                                      |                                                   | • •                                       | s<br>S                                  | 1                                           | ¢Û                               | 6 I I I                | 11                                      | 12          |
| -4        | *****                                  | 13 14<br>**-72.555193*                   | 15<br>144444444444444                             | 15<br>**********                          | <del>a44-3</del> 5.251050:              | *****                                       | *******                          | *********              | - +++++++++++++++++++++++++++++++++++++ | 1.922223    |
| ~         | -76.9551                               | 442 8.035575<br>33:44444444444           | 1-33.629375 3<br>19499404194444                   | .205029<br>*******123.923;                | 251 -6.050852                           | 543-361354 -3.                              | .825316332.745                   | 285-50.773355          | 83.729535 -3                            | 11535317    |
| -         | )///////////////////////////////////// | 427 2.924744444<br>4444444444444444      |                                                   | •1333523<br>716049 -2•323                 | 302 32.461375                           | 41 **********                               | 015756-35.705                    | 545372-930544          | -2,563639 21                            | 56015s.     |
| 4         | 1807 · F.5                             | 2-6-20.731744<br>440000000000            | . 40.537435 0<br>33.7156494444                    | .435525<br><del>*******50.953</del>       | 425 I.050345                            | -37.403837 -3.                              | ,133412#=#####                   | ±≠¢ 10°¢5352           | 5 +++++++++                             | C+05551     |
| Ś         | 507.7-J.                               | 443 -0.540354<br>44123.923251            | -1.506233 0<br>-2.320362-50.                      | .254079<br>359436#########                | ≈a≄ 3.083053                            | 203.458737533.                              | .353243 30.957                   | 156-55.700442          | 132.722453 73                           | .952895     |
| -0        | -24.375.                               | 313 -3.312745<br>50 -5.952352            | 32-451375 13                                      | .653170<br>050345 3.032(                  | 953******                               | *********                                   | 735317110.261                    | 652 -0- <u>0</u> 03647 | 1 \$\$\$\$\$\$\$\$                      | .225890     |
| *         | 116-55                                 | 328 -0+022197<br>445+3+00125+0           |                                                   | -521323<br>403907203.4521                 | 737444444444                            | *********                                   | ,72+335-25-23                    | 697444444466           | *****                                   | ****        |
| ~         | 17-357                                 | 537 4.535275<br>J# -3.325515             | .4*44444444444447<br>14.015765 -3.                | .316463<br>133412533.353,                 | 243 3.785317                            | 12.724335444                                | 112-55144444444                  | 512 -J.571512          | 444444444 -3                            | +620T3•1    |
|           | -52-5121                               | 754 0.255719<br>44322.745.257            | 1770.022653 -1<br>·56.705545 <del>494</del> 4     | .452285<br>www.400 30.3571                | 156110.261652                           | -28.235647165.                              | .712512######                    | ### <b>5.2424</b> 52   | 22.556243-53                            | 122666.     |
| 2         | *****                                  | 1121720213000000000000000000000000000000 | 40,735450 1<br>172,933544 23.4                    | . 513?50<br>465355-55. 7954               | ++                                      | 0- *********                                | 571512 5.242                     | *52###########         | 227.747035 -3                           | UTEEC1.     |
| -         | *********                              | 444 C.0133224                            | -14.034245 -3<br>-1.56366444444                   | .222563<br>200400322.7274                 | ***********                             | ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,     | :                                | 245227+7+7555          | **********                              | .117376     |
| • •       |                                        | 375 44-544563<br>11 14.544717            | 1441.59969.4000.000.000.000.000.000.000.000.000.0 | 101000<br>101000                          | 435 I.225390:                           | l- ttottottot                               | ,5162 <i>3</i> 4-33 <b>.</b> 563 | 527 - J. T3370         | 3.117575227                             | ******      |
|           | *****                                  | 101 (111) (111)                          |                                                   | -223313<br>7-7-2-213                      | 113 55.471328                           | 17-357647-52,                               | .5127540000                      | ************           | -63.223375505                           | *****       |
|           |                                        | 10210-0-0000                             |                                                   | 610-11-11-11-11-11-11-11-11-11-11-11-11-1 | 1919-0-971<br>7919-0-972                | 4. 745275 0.                                | 1-5.113 73.541                   | 511 0. <b>33</b> 3663  | 44.521535 9                             | .223T39     |
| •         |                                        |                                          | 792.4275757                                       | 1016150<br>1026150                        |                                         |                                             |                                  |                        | rostiftst . '                           | 20022222    |
| *1        | 1252222                                |                                          |                                                   | 50023 JT.33C                              | 021ataananaaaa                          |                                             | .022533-+=+172                   | コトスナトフォト マレついま         |                                         |             |

•

| 5.111.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1    |                                             |                |                |                |                                 |                                 |                                       |                |                |                |               |                |                                        |               |               |                      |                           | 6 a 1 C 0 0 0 1 K                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|--------------------------------------------|---------------------------------------------|----------------|----------------|----------------|---------------------------------|---------------------------------|---------------------------------------|----------------|----------------|----------------|---------------|----------------|----------------------------------------|---------------|---------------|----------------------|---------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| 50 -J.242035                               |                                             |                | ,              | <b>1</b>       |                                 |                                 | -                                     |                | · •            |                |               |                |                                        |               |               |                      |                           | 5010500.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| 52,33 1.51 <i>0</i> 3                      |                                             |                |                |                |                                 |                                 |                                       |                |                |                |               |                |                                        |               |               |                      |                           | 0°0°235053                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| - 115+5<br>- 12+5                          |                                             |                |                |                |                                 |                                 |                                       |                |                |                |               |                |                                        |               |               |                      |                           | 166900.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| r+-÷21323-€                                |                                             |                |                |                |                                 |                                 |                                       |                |                |                |               |                |                                        |               |               |                      |                           | 0.004715                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| станичения с.115665<br>3 С.+Ссбія 2.254770 | 5 # 8 0 0 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 | 0.355133+1-21  | -3.37517552+71 | 10+31517556+01 | -6.517+1+68+31                  | ].51741458+01<br>.^ 4*644484845 |                                       | -0.7293721E+01 | 0.7253721£+J1  | -3.11-1514540  | 0.11415145402 | -9-21535552+92 | 3+21569552+32                          | -3-1457636432 | 5+2+537595+32 |                      | 23 M3C 55                 | $\begin{array}{c} -36.64 \\ -36.64 \\ -36.64 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.65 \\ -4.6$ |
|                                            |                                             | 10-04102071471 | -0-1011-101-0  |                | 1 3 + 11 + 12 A A A + 14 + 19 + | 00103000100001                  | 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 |                | -0.53322225-31 | -2-27297155+39 |               |                | 10-00000000000000000000000000000000000 | 8019600100101 | 99-94001000   | REALA ROUTINE EFENDE | DA4PING AATIJ 32 CINTRILL |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|                                            |                                             |                |                |                |                                 |                                 |                                       |                |                |                | :             | 236            |                                        |               |               |                      |                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |

.

·

IN AGUTINE FOR CFRICIENCY COMPLETIONS 

.

INITIAL MODAL STATE VECTOR:

| 0.99000005+39<br>0.00030032+00<br>0.0030035+00    |
|---------------------------------------------------|
| -0.45213075-02<br>-3.63552555-02<br>0.33135965-04 |
| 6+00005+30<br>9-3030005+30<br>0+0035000±+30       |
| +0+1388441-1<br>-0+557584-1<br>-0+24675884-84     |

-0.54330425-09 -0.43333226-02

.

PER CENT EFFICIENCIES

EFFLUIENCY CUEFFLCLENT:

Ũ+<u>3725515</u>+01

GLUBAL EFFICIENCY

0-17633332:+33

RELATIVE 4036L EFFICIENCY:

ţ,

0+65713625+00

0.11300225-32 

20-25166020-02

3+5+65023E+0C 

GLGBAL SPILLOVER JUSTIERTS

2.55315235+53

ĉ

| 1<br>-3.203171 | -0.40413 | 1001010- |   | 0.037912 | 124612 | -0.065573 | 0.021395 | 0.122535 | -0.362457 | 0.303325 | -0-004058 | -0.000035 | 0.306337 |
|----------------|----------|----------|---|----------|--------|-----------|----------|----------|-----------|----------|-----------|-----------|----------|
|                | ~ ~      | ·+ ·r    | 0 | 1        | 'n     | •         | 7        | 11       | 77        | 13       | <b>.</b>  | 1.5       | 1 j      |

ພ ບ

| 1 | 0.304437 | +06210-0 | 0.235517 | 0.005321 | 0.153654 | 0.305271 | 0-502232 | 0.92-240 | 0.965113 | 0.000715 | J. J27637 | G. 313615 | 0.00017 | 0.0000.0 | 0.00011.0 | 0-00000 |
|---|----------|----------|----------|----------|----------|----------|----------|----------|----------|----------|-----------|-----------|---------|----------|-----------|---------|
|   | ٦        | 7        | ~        | .+       | ŝ        | 9        | 1        | q        | •        | 10       | 11        | 12        | 13      | 4 T      | 15        | 15      |

|   | .000715 | .327637 | .313615 | 10000 | .303015 | <br> |  |
|---|---------|---------|---------|-------|---------|------|--|
| 5 |         |         | с.      |       |         |      |  |

ストロロションロオールの「ロコンのコイトトロイルオイン

|                                       |     |                                          | 2                                     |             |                 |
|---------------------------------------|-----|------------------------------------------|---------------------------------------|-------------|-----------------|
|                                       |     |                                          |                                       |             | 110             |
|                                       |     |                                          | •                                     |             |                 |
|                                       |     |                                          | 41 6                                  |             | เม่าม           |
|                                       |     |                                          | 0                                     |             | r m             |
|                                       |     |                                          | 200                                   |             | _i m            |
|                                       |     |                                          | 001                                   |             | <u>.</u>        |
|                                       |     |                                          |                                       |             |                 |
|                                       | (*  |                                          |                                       |             |                 |
|                                       |     |                                          | 337                                   |             |                 |
|                                       | -   |                                          | 0.00                                  |             | 00              |
|                                       | U   |                                          | m ô T                                 |             |                 |
|                                       | 11  |                                          |                                       |             |                 |
|                                       | >   |                                          |                                       |             |                 |
|                                       |     |                                          | • •                                   |             |                 |
|                                       | 5   |                                          | 100                                   |             | 4               |
|                                       |     |                                          |                                       |             | 0.50            |
|                                       |     |                                          |                                       |             |                 |
|                                       | P=4 |                                          |                                       |             |                 |
|                                       |     |                                          | 81 5                                  |             | <b>N</b> 1 11   |
|                                       | • • |                                          | 01000                                 |             | 21.00           |
|                                       | 14, |                                          | دە ل. •                               |             | mon             |
|                                       | 11  |                                          | 0 41 1- 1- 11                         | ••          | 3. 10           |
|                                       |     |                                          | 5 S 10                                |             | 10 -4 -3        |
|                                       |     |                                          | 0000                                  | -           |                 |
|                                       |     |                                          | 0000                                  |             | 000             |
|                                       |     |                                          |                                       |             |                 |
|                                       |     |                                          |                                       | 0           |                 |
|                                       |     |                                          | 141 62 12                             |             |                 |
|                                       |     |                                          | HZHHY                                 |             | 1 2 0           |
| 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 |     | **************                           | 04220                                 | 14          | 004             |
| 21402 24256500                        |     |                                          | 2 H M M M                             |             |                 |
|                                       |     | 4 4 N 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0      | 10 D U U I                            | • •         | 14.8 (14.11.12) |
|                                       |     | 000000000000000000000000000000000000000  | •• • • • • • •                        | <u>n</u>    | 1.0 0           |
|                                       |     | 000000000000000000000000000000000000000  | 13 ac + +                             | 5           |                 |
|                                       |     |                                          |                                       |             |                 |
| <u></u>                               |     | 01001000000000000                        |                                       | 0           |                 |
|                                       |     |                                          | <u> </u>                              |             |                 |
|                                       |     |                                          | 1 - 5 -                               | ā           |                 |
|                                       |     |                                          | زنبي الس                              | ÷           | 000             |
|                                       |     |                                          | -1 1                                  | -           |                 |
|                                       |     |                                          | <b>թ. է ս_ էլ</b>                     | C.C.        |                 |
|                                       |     | 0 (1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 |                                       | 10          |                 |
|                                       |     |                                          | ••••••••••••••••••••••••••••••••••••• | ĩ           |                 |
|                                       |     |                                          |                                       | <b>&gt;</b> |                 |
|                                       |     |                                          |                                       |             |                 |

#### /ECT09

.

| 2 | 3 | 9 |
|---|---|---|
|   | - | - |

-----

•

EQUMENATUR CONSTRUCTS

| 0.159141E-11<br>0.784335E-03                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 0.112553E-15<br>0.954674E-02                                                                                       |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------|
| 0.542401E-04<br>0.304599E-03<br>0.243504E-04                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 7 0 - 5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 1555+30<br>0.9275535-02<br>0.1104126-02<br>0.2330836-37                                                            |
| 0.11.3496-03<br>0.3740126-01<br>175)/06M24.) :<br>6.285106-01<br>6.285106-01<br>0.3725946-03<br>0.3725946-03<br>0.3725946-03<br>0.3145346-03<br>0.2145346-03<br>0.2145346-03<br>0.2145346-03<br>0.2145346-03<br>0.2145346-03<br>0.2145346-03<br>0.2145346-03<br>0.2145346-03<br>0.2145346-03<br>0.2145346-03<br>0.2145346-03<br>0.2145346-03<br>0.2145346-03<br>0.2145346-03<br>0.2145346-03<br>0.2145346-03<br>0.2155555<br>0.2155555<br>0.215555<br>0.2155555<br>0.2155555<br>0.2155555<br>0.2155555<br>0.2155555<br>0.2155555<br>0.2155555<br>0.2155555<br>0.2155555<br>0.2155555<br>0.2155555<br>0.2155555<br>0.2155555<br>0.2155555<br>0.2155555<br>0.21555555<br>0.21555555<br>0.21555555<br>0.215555555<br>0.21555555555555555555555555555555555555                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | <                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | CC4P. CCSTS:                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 057 : 0.5493<br>0.1115705-03<br>0.3024455703<br>0.3324455703                                                       |
| С (                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 00000000000000000000000000000000000000                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | CR C DENJHINATOR<br>C - 2534834640<br>C - 523334640<br>C - 134695640<br>C - 134695640<br>C - 134695640<br>C - 134695640<br>C - 134695640<br>C - 21765640<br>C - 2185<br>C - 21 |                                                                                                                    |
| - с.с.<br>- с. | MD06CIN % 1005EL) 6 3<br>MD06CIN % 1005EL) 6 3<br>MD06CIN % 5 5<br>1 2 3<br>1 3 | <b>BLGCK-JIAJ NU4ERAT:</b><br>0.2963345-34 5<br>0.5633716-04 5<br>0.1295496-25 5<br>0.159602-04 5<br>0.1595256-03 5<br>0.1595256-03 5<br>0.4150352-13 5<br>0.6130395-13 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | БГ. (СА-) (120, Б. 8. 8. 8. 9. 1)<br>(20, 2, 15, 2, 2, 2, 2, 3, 3, 3, 3, 3, 2, 2, 2, 2, 2, 2, 2, 2, 2, 2, 2, 2, 2, |

教育教育教育教育社会中主体中的"ALL",在19世纪和19世纪的教育,会生活性的体育和教育教育教育教育

|                 | <b>J.</b> 35+5755-52 | 0 27552-02 | 9.1154116-02 | 0.3054466-33 | 0.1115736-03 | <b>J.2335535-07</b> | 0.9591557-05 | <b>J.11</b> 25535-16 |
|-----------------|----------------------|------------|--------------|--------------|--------------|---------------------|--------------|----------------------|
| Ajus(14 Aljich) | ^                    | 2          | ŗ.           | <b>.†</b>    | L            | T                   | 7            | ÷                    |

04

| 0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.000<br>0.0000<br>0.0000<br>0.0000<br>0.0000<br>0.0000<br>0.0000<br>0.0000<br>0.0000<br>0.0000<br>0.0000<br>0.0000<br>0.000 | 000000000<br>++++0++++++<br>400000000<br>++++0+0000<br>00000000 |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------|
| う 11 年 12 年 12 年 13 年 19 月 19 年 19 月 19 年 19 月 19 月 19 月 19                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                 |

ы С
# CHARACTERISTIC ERFICIENCIES

| ***** |   |   |    |     |     |     |   |    |    |   |   |    |    |    |    |     |    |  |
|-------|---|---|----|-----|-----|-----|---|----|----|---|---|----|----|----|----|-----|----|--|
| 1     |   | h | 4  | in, | ŝ   | .\$ |   | .# | n) | ~ |   | 'n | N  | 'n | -4 | 1.4 | ~  |  |
| ŗ     |   |   | ς. | 5   |     | ž   |   | 3  | 2  | 7 | 2 | 5  | 2  | 2  | 2  | 2   | 2  |  |
|       | - | • | ò  | ń   | en. | N   | m | ÷  | -  | ò | m | -  | 'n | ~  | m  | •   | -0 |  |
|       | - | 1 | 2  | 2   | 2   | -   | 2 | 2  | 0  | 0 | 2 | -  | -  | 3  | 2  | 2   | Ċ  |  |
|       |   | 2 | 7  | ?   | ٦.  | •   | ? | ?  | 2  | 2 | 2 | 7  | 3  | ?  | 3  | 2   | 2  |  |
|       | c | 5 | ò  | ò   | o   | ò   | o | ò  | o  | õ | 0 | ò  | ò  | ō  | ō  | ō   | ő  |  |
|       |   |   |    |     |     |     |   |    |    |   |   |    |    |    |    |     |    |  |
|       |   | • | ~  | Υ   | •#  | ŝ   | • | ~  | ð  | Τ | 2 | 1  | 2  | 2  | 1  | 2   | 9  |  |
|       |   |   |    |     |     |     |   |    |    |   | _ |    |    | _  |    | _   | _  |  |

#### EFFICIENCY VECTOR

-

| - | 0.011550 | 0.301396 | 0.030333 | 0.303607 | 0.300557 | 0.000555 | 0+3005 | U.000135 | 0.303072 | 0.000032 | 0.00019 | 3.000005 | 0.00001 | 0.00000 | 0.00000 | 0.00000    |  |
|---|----------|----------|----------|----------|----------|----------|--------|----------|----------|----------|---------|----------|---------|---------|---------|------------|--|
|   | -        | ~        | m        | 4        | 2        | •        | ~      | -0       | •        | 10       | :1      | 12       | 13      | 14      | ÷ 5     | <b>1</b> 0 |  |

EF4.010401 G4114 44467 11144. 0.5571506-02

せいたいいい たいりょうたまれい とうしきずみそう

| 3+7523332E+11<br>2+531742E+33                |
|----------------------------------------------|
| 0+1575276+6+<br>0+1527524+01<br>0+1253516+03 |
| 4 - 1 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5    |

さられのロワームでリアバルアロワーズロムマメリカバメリ

.....

:

•

•

. -

.

••

......

| 0.121745-10<br>0.11+435-02                                                                                                                                                                      |                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 0.5921365-32<br>0.5340114-32 |
|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------|
| 0.4473395-04<br>0.1730155-02<br>0.1290231-03                                                                                                                                                    |                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                              |
| 0.1224.96-02<br>0.3740126-01<br>0.1175206-10<br>0.1331392046-02<br>0.1331396-02<br>0.1331396-02<br>0.1331396-02<br>0.5241335-04<br>0.5241335-04<br>0.5241335-04<br>0.5241335-04<br>0.5241335-04 | ددمیدوسوهممیدهد<br>6571835-02<br>۲ CC4P. CGSTS:<br>12 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                              |
|                                                                                                                                                                                                 | ######################################                | 0.115136:0<br>0.134:5952*0<br>0.134:252:40<br>0.391172:2-40<br>0.5517055-0<br>0.170755-0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                              |
| III III III     8 0 2 10       III III III     8 0 2 10       III III IIII     8 0 2 10       III IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII                                                             | ##vauaeeeeeeeeeeeeeeeeeeeeeeeeeeeeeeeeee              | 0 - 7969078-11<br>0 - 6990128-10<br>0 - 1975108-00<br>0 - 1975108-00<br>0 - 2205158-00<br>0 - 2205158-00<br>0 - 100<br>0 - 1000<br>0 - 100<br>0 - 1000<br>0 | <ul> <li></li></ul>          |

, T

.

οκοκίσει το γιατογιστικό του που ποιαστατομούς Γεξαμομίζει Οσημού 100 ΝΕ ΟΟΥΝ του 1010 - 10 Γεγιάτηζεί Ουλαιτικά τον το Οληγούς του το 10 机晶体晶体的 建化物合物 网络外部的复数形式 使用的过去式和过去分词 化乙酰乙酰胺 化乙酰胺乙酰胺 化丁

| 3L C ( C ( 3 4)                                                                 | 0+34524+5104<br>0+33212954+02 |
|---------------------------------------------------------------------------------|-------------------------------|
| ר<br>יי<br>יי<br>יי<br>יי<br>יי<br>יי<br>יי<br>יי<br>יי<br>יי<br>יי<br>יי<br>יי | 7 2                           |

NP TH 5:055 51264-2145.8841618NGY 542 : 0.6553125-02 17.51.52.0002, 62, M173143, 1.6547LNS. ••

į.

#### **EXAMPLE 6.4**

```
VUL06-4 STACTU-4 REFINITERON 4440 VOID
RVALUATIUN MUSEL ORGEN VERVER
```

12 N - : THE CONTRUL DESIGN MODEL DRUER

01 = 2 NUMBER OF INC. NUMBER OF RESIDENT AUTON NABLES

42 = d N AEDJOED PRINCIPLL EFFICIENCY COMPONENTS DRJER DF

# 3 P NUMBER LF CUTPUTS

```
0.625732E-01
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       0.4559315402

0.1247355402

0.1247355402

0.1247355402

0.13037995402

0.13037995402

0.4704415402

0.4704415403

0.7259735475403

0.2530756475404

0.40305475404

0.40305475404

0.4517435476404

0.45124204

0.45124204

0.5594325404

0.5594325404
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           0.439232E+34
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             0 * 9 4 7 5 5 7 E + 0 +
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               0.9309015+04
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                0.1004245+05
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          2
                                                                                                                                                                                                                                                                                                                                                                                                   16
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          ŝ
                                                                                                                                                                                                                                                                                                                                                                                                   15
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         •
                                                                                                                                                                                                                                                                                                                                                                                                          14
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          0.5103475-01
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       \begin{array}{c} 0 & \cdot & \cdot & \cdot \\ 0 & \cdot & \cdot & \cdot & \cdot \\ 0 & \cdot & \cdot & \cdot & \cdot \\ 0 & \cdot & \cdot & \cdot & \cdot \\ 0 & \cdot & \cdot & \cdot & \cdot & \cdot \\ 0 & \cdot & \cdot & \cdot & \cdot & \cdot \\ 0 & \cdot & \cdot & \cdot & \cdot & \cdot \\ 0 & \cdot & \cdot & \cdot & \cdot & \cdot \\ 0 & \cdot & \cdot & \cdot & \cdot & \cdot \\ 0 & \cdot & \cdot & \cdot & \cdot & \cdot \\ 0 & \cdot & \cdot & \cdot & \cdot & \cdot \\ 0 & \cdot & \cdot & \cdot & \cdot & \cdot \\ 0 & \cdot & \cdot & \cdot & \cdot & \cdot \\ 0 & \cdot & \cdot & \cdot & \cdot & 0 \\ 0 & \cdot & \cdot & \cdot & 0 \\ 0 & \cdot & \cdot & 0 & \cdot & 0 \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & \cdot \\ 0 & \cdot & 0 & - & 0 & - \\ 0 & \cdot & 0 & - & 0 & - \\ 0 & 0 & - & 0 & - \\ 0 & 0 & - & 0 & - \\ 0 & 0 & - & 0 & - \\ 0 & 0 & - & 0 & - \\ 0 & 0 & - & 0 & - \\ 0 & 0 & - & 0 & - \\ 0 & 0 & - & 0 & - \\ 0 & 0 & - & 0 & - \\ 0 & 0 & - & 0 & - \\ 0 & 0 & - & 0 & - \\ 0 & 0 & - & 0 & - \\ 0 & 0 & - & 0 & - \\ 0 & 0 & - & 0 & - \\ 0 & 0 & 0 & - & 0 & - \\ 0 & 0 & 0 & - & 0 & - \\ 0 & 0 & 0 & - & 0 & - \\ 0 & 0 & 0 & - & 0 & - \\ 0 & 0 & 0 & - & 0 & - \\ 0 & 0 & 0 & 0 & - \\ 0 & 0 & 0 & 0 & - \\ 0 & 0 & 0 & 0 & - \\ 0 & 0 & 0 & 0 & - \\ 0 & 0 & 0 & 0 & - \\
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          *-
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  0.1010515+05
                                                                                                                                                                                                                                                                                                                                                                                                   5
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         ŝ
                                                                                                                                                                                                                                                                                                                                                                                                   12
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         in
                                                                                                                                                                                                                                                                                                                                                                                                   Ξ
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                217 . 213 214
217 . 213 214
217 . 213 214
217 . 213 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214
217 . 214 . 214 . 214 . 214 . 214 . 214 . 214 . 214 . 214 . 214 . 214 . 214 . 214 . 
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         4
                                                                                                                                                                                                                                                                                                                                                                                                          10
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         0+ 9516756+04
0+1057256+05
0+1170406+05
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               +2+-13-164-0
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 m
                                                                                                                                                                                                                                                                                                                                                                                                   ÷
C1
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         ~
                                                                                                                                                                                                                                                                                                                                                                                                          71
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         -
                                                                                                                                                                                                                                                                                                                                                                                                   ~
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       \begin{array}{c} \mathbf{N} \\ \mathbf{
                                                                                                                                                                                                                                                                                                                                u.
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                1L
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         SELECTED INPUTS:
                                                                                                                                                                                                                                                                                                                                                                                                   100531
                                                                                                                                                                   u.
                                                                                                                                                                                                                                                                                                                                                                                                                  SELECTED
                                                                                                                                                                                                                                                                                                                                       ď.
```

.

0.111423E+05 0.123537E+05

0.11003+E+35

|             |             |             |             |               |             | -3.70775-02  | -2.72036+61                             | 3.45192-23                            | -3.25536-32 |
|-------------|-------------|-------------|-------------|---------------|-------------|--------------|-----------------------------------------|---------------------------------------|-------------|
| 8,90355-04  | 1.15375-02  | -1.22226-03 | 9.87755-03  | 8.12555-03    | 9.93995-03  | 1.31795-32   | a. +5532-ú3                             | 1.53125-03                            | -5.43945-03 |
| 3.8917É-02  | 1.25236-02  | 9.2054E-04  | -7.80055-04 | -2.5170E-03   | 2.29495-04  | 5.42572-32   | 5 - 2 3 9 5 - 9 4                       | 1.20335-32                            | -1.65455-03 |
|             |             |             |             |               |             | 1.00055+33   | 0.03095+30                              | 0.00005+00                            | 0.03036+30  |
| 0.00006+33  | 0.00005+00  | 0.00005+00  | 0.0000E+00  | 0.00006+30    | 0.0000E+00  | 0.430000.00  | 0-70000-0                               | 0.00005+30                            | 0.45020840  |
| 0.00006+03  | 0.0330E+00  | 0.+50000.0  | 0.00005+00  | 0.0000£+30    | 0.0000E+00  | 0.430000.0   | 0.00006+00                              | 0.000000000                           | 6.30005+33  |
|             |             |             |             |               |             | 1.95542-39   | 1.33505-63                              | -5.547~5-09                           | -2.67725+01 |
| 7.34306-10  | 5.19145-11  | 8.4135E-05  | 2.43192-36  | -1.6001E-09   | -5.2107E-09 | 5.03025-09   |                                         | 7.73385-13                            | 1.60972-10  |
| -1.3212E-05 | -4.6452E-09 | 2.23755-13  | 6.51405-10  | -1.2577E-08   | 1.15595-10  | -1.5965E-33  | -2.41546-10                             | -4.01175-29                           | 0.31535-10  |
|             |             |             |             |               |             | 0.00006+00   | 00+30000*30                             | 1.33255=+53                           | C.•≦00CC.0  |
| 0.00366+30  | 0.03036+33  | U.OUJGE+JJ  | 0.430063.0  | 0.00000000000 | 0.430000.0  | 0.00000000   | 00+0000010                              | 0.0000640                             | 0.0300=+33  |
| 0.000055+23 | 0.00006+00  | 0.90906+33  | 00+20000-0  | 0.0000000000  | 0.+30000.0  | 0.05305+30   | 0.+00100+00                             | 0.45000.40                            | 0.60006+00  |
|             |             |             |             |               |             | 4.52335-32   | 12-36363.1                              | 2.35315-25                            | 7.63035-42  |
| -4.56355-91 | -1.77275+01 | 1.73355-72  | -3.13056-02 | -4.13305-01   | 4.22495-31  | -3.70535-34  | -4.523502                               | -5.3531 - 11                          | 7.92335     |
| -5.36925-02 | 3.1870E-32  | -5.1351E-11 | 2.J311E-J3  | -6.88345-02   | -1.97436-02 | -6.95355-32  | 75603                                   | -1.5257:-01                           | -1-717-5-55 |
|             |             |             |             |               |             | 0.430000.0   | 0.43030.430                             | 0.30305.0                             | 0.000000    |
| 1.00006+00  | Q.0009E+30  | 0.00006+30  | 0.430000.0  | 0.00005+00    | 0.00006+00  | 0.00006+00   | 0.+20.00.40                             | 0.90001+00                            | 0.+=0.00.0  |
| 0.90305+30  | 0.00005+00  | 0.000000+30 | 0.30005+30  | 0.00005+00    | 0.00005+00  | 0.95035+33   | 0.+33555+30                             | 6.33342+33                            | 0.0301=+00  |
|             |             |             |             |               |             | 2.35375-32   | 1.25742-51                              | 1.00341-31                            | 4.3571=-01  |
| 5.3030E~03  | -2.J231E-02 | -2.41136-31 | -1.7644E+01 | -1.1515Ê-02   | 4.24306-02  | -3.45-72-31  | 10-20203-3                              |                                       | 20-20101-2- |
| 4.50475-32  | 2.+3935-02  | -3.72305-02 | -1.41056-02 | 5.1329:-01    | -4.80915-33 | -2.32155-32  | BO-90100-1                              | 1.].54545-72                          | -3-64101445 |
|             |             |             |             |               |             | 0.430000.0   | 00+4000000                              | 0(+:0000 °F                           |             |
| 0.30005+30  | 0.000005430 | 1.00305+30  | 0.00005400  | 0.35035+00    | U.+0000E+JD | 0.+200000    |                                         |                                       |             |
| 0.0000=+00  | 0.10005+30  | 0.0355+90   | 0.0000E+00  | 0.00005+33    | 00+30000-6  | 0.0000000    | ·······                                 |                                       |             |
|             |             |             |             |               |             | 3 1 6 - 32   | 1.1.1.5 7.1                             |                                       | ****        |
| -1.52365-31 | -5.62435-01 | -3.04906-03 | -1.65765-02 | -2.31976-01   | -1.50125+01 | -1.34758-93  |                                         |                                       | 3.735.5-1   |
| -4.27525-02 | L.U.035-03  | -1.53545-01 | 9.5595E-03  | -2.23456-02   | -6.42115-33 | - 5.75255-72 | • ( • ; • ; • ; • ; • ; • ; • ; • ; • ; | -1.2007 -                             |             |
|             |             |             |             |               |             | 0.00,00,00   | 0                                       | · · · · · · · · · · · · · · · · · · · |             |
| 0.+2000.+0  | U.JJ.CE+G0  | 0.(1332+30  | 0.03065+53  | 1.0001-+30    | 0.6+30000.0 | 0.355010     |                                         | • • • • • • •                         |             |

CLOSED LOOP EIGEN/ALUES =

|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 0 0                                                                      |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 0 <b>n</b>                                                               |
| * * * * * * * * * * * * * * * * * * * *                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | • •                                                                      |
| The state of the state of the man of the state of the sta | 4. 14.                                                                   |
| ~~~ oucoalessatesserements                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 1- m                                                                     |
| · ユミネアアタタロロタタランショウルエカムシシン                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | ~ ~                                                                      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ~ ~                                                                      |
| • • • • • • • • • • • • • • • • • • • •                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | • •                                                                      |
| 000000000000000000000000000000000000000                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 00                                                                       |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                          |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                          |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 14 61                                                                    |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                          |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ынын<br>                                                                 |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | n 1.                                                                     |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                          |
| a n a a a a a a a a a a a a a a a a a a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | n n                                                                      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                          |
| 444444444444444444444444                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                          |
| **********************                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                          |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                          |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                          |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | <b>•</b> • •                                                             |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                          |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | * *                                                                      |
| the second of the second of the second second at the first of the second s                                                                                                                                                                                                                                             | 1.1.1.1                                                                  |
| ついいトトゥックののののうけいしのうナナトト                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | , ,                                                                      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | N - N                                                                    |
| H + + + U U U N N M M M M M M M M M M M M M M M                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                          |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                          |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                          |
| 000000000000000000000000000000000000000                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                          |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                          |
| <b>??</b> ??????????????????????????????????                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                          |
| <ul> <li></li></ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                          |
| 99999999999999999999999999999999999999                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | н<br>1<br>1                                                              |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                          |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ася<br>1-1<br>1-1<br>1-1<br>1-1<br>1-1<br>1-1<br>1-1<br>1-1<br>1-1<br>1- |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                          |
| COORDONAL CONTRACTOR CONTRAC      |                                                                          |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                          |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                          |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                          |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                          |

.

4.1331E-01 -1.3039E+00 -1.4069E+00 -3.3605E+00
5.6483E+01 1.3117E+C1 3.5005E-01 5.2059E-01 \$+07551-01 1.554505-11 +2.75+1.-02 +4.45513-11 1.34415-01 -2.93725+01 +1.01215+52 +2.42475-14 +1.51765+01 1.94465+01 +4.14975+00 -1.90725-01

.

0.1259296.911350371.2037U1 .. ....

- <u>,</u>
- .,
- ~.
- 5.1
  - -5.J4114152.39%5626-20.542733 21.691403 21.351540101.01415644444444444956.546274444444444444444444444444444444 3
- ακασοιοιος 31.33251μεταροσούσισταναγτοποτοντάταταν 84.0451κμετατοτοτοτοτοτότοτο 54.4116τατατοτοτότιος 356547 ?
- άμερικά ματάμεται ετη ετητητή 10.41545ο πάμεταν τα πάμετα 54.411948-98.418948 τα 94.42494 τα 94.47894 τα 94.478 77
- - 4 r 4
- 3
  - 4

### -6.77713465+06 COMFLEX CLOSED-LJOP EIGENVALUES -0.10433376400 +0.777

0.77713435+03

-0-10+3307E+07

| n          | ς,         | 0       | 0   |      | ч        | -          | ч          |     | -    |            | 1    |          |                                        |                         |
|------------|------------|---------|-----|------|----------|------------|------------|-----|------|------------|------|----------|----------------------------------------|-------------------------|
| 0          | <b>C</b> 1 | 0       | 5   | S    | 0        | • •        | 0          | 0   | 13   | ć.         |      |          |                                        |                         |
| +          | +          | +       | +   | +    | +        | +          | +          | •   | •    | +          |      | •        |                                        |                         |
| ιu.        |            |         | 111 | 111  | 14.7     | <b>U</b> 1 | .11        | 112 | 111  | 111        |      |          |                                        | 111                     |
| ŝ          | ŝ          | C       | 0   | m    | m        |            |            | 3   | ~    |            |      | C 4      | ~                                      | <ul> <li>C •</li> </ul> |
| 3          |            | r.)     | ~1  | 10   | ŝ        | 1.4        | 14         |     |      | 0          |      |          | ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~ | 5                       |
|            | -1         | 0       | • • |      | 4        | -          |            |     |      | r.)        |      |          |                                        | ū                       |
| • 4        | 7          |         | *** | -43  | <b>S</b> | ~          | ŝ          | 5   | \$   | 14.1       | n -  | - 23     | .,                                     | - 10                    |
| C          | 0          | ~~      | ŝ   | Ś    | •0       | ŝ          | ŝ          | •   | +    | · ^        |      |          |                                        |                         |
| ŝ          | 4.1        | •0      | •   | -4   | -        | ŝ          | 6+1        | 5   | - 44 | ~          | ~    | 5        | *                                      |                         |
| (P         | Ē.         | (P)     | / m | • 4  |          | <b>N</b>   | e1         | 3   | ŝ    | <b>C</b> 1 | ŝ    | (1)      | <b>M1</b>                              | <b>T</b> 1              |
| •          | •          | •       | •   | •    | •        | •          | •          | •   | •    | •          | •    | •        |                                        | •                       |
| C          | 0          | $\circ$ | 0   | 0    | 0        | 0          | 0          | 0   | o    | C)         |      | 0        | o                                      | • • •                   |
|            |            | ,       |     | ,    |          | ,          |            | ,   |      |            |      | ,        |                                        |                         |
|            |            |         |     |      |          |            |            |     |      |            |      |          |                                        |                         |
| <i>.</i> , | .,         | 0       | 0   | 0    | 6        | 2          |            |     |      | 0          |      |          |                                        |                         |
| 0          | 0          | ••      | o   | 0    | ō        | 0          | 13         |     | 53   |            |      |          |                                        |                         |
| ٠          | •          | ٠       | •   | 4    | ÷        | +          | •          | •   | •    | •          | ¥ .  | 1        | 1                                      | 1                       |
| 114        | 141        | 14.8    | 101 |      | 147      | 112        | 10         | 112 |      |            |      |          | i                                      | . i.                    |
| 0          | 2          | *       |     | *    | *        | 5          | ŝ          |     | ÿ    | 1.1        |      | ,        |                                        |                         |
| -+         | -1         | m       | -n  | ~    | ~        | - <b>P</b> | (P         | 1.4 | 1.1  | ń          |      |          | . n                                    | - 14                    |
| -          |            | e 1     | 2   | . 17 | m        |            |            | 0   | -    |            | - 4  | 11       |                                        |                         |
| ÷0         | -0         | · 4     | e-1 | 5    | v٦       | ~          | ~          | •   | 10   | . 7        | ,    | <b>.</b> |                                        | •                       |
| ŝ          | -0         | *       | - 4 | ~    | /h       |            |            | ~   | *    | ~          | ~    | -4       | - 4                                    | 7 B                     |
|            | *          | ~       | 1.4 |      |          | ŝ          | 143        | •   |      | 10         | ·· • |          |                                        |                         |
| -4         | ÷-4        | -4      |     | 2    | ~        | m          | <b>C</b> 1 |     | - i  | +          | •    |          | ~                                      |                         |
| •          | •          | •       | •   | •    | •        | •          | •          | •   | •    |            |      |          |                                        |                         |
| o.         | 0          | 0       | ~   | 0    | 0        | 2          | 0          | C   | 0    | 2          | . 7  | o        | 0                                      | "                       |
|            |            | ŧ       |     |      | 1        | •          |            |     |      | i i        |      |          | 1                                      |                         |

0.3+75535+51

....................... -3. 3. 15 . 55 . 51

| 1:*::::::::: | よくチェスをち よるシャン               | さっ チョック ナマナ ちょうしょう  | 10+27777675+0+ | 10+921224+0   |
|--------------|-----------------------------|---------------------|----------------|---------------|
|              | , 1 + 3 / 2 / 1 > 1 / 1 > 1 | <br>といっという71,57にそい。 | +6-1034472-01  | 11-375-252-27 |

0.0134433

0.1155051 0.9529291 0.1223431 0.2032345 0.1272730 DAMPING ARTICLER CONTRULED AND

# 

# うんりゃうしょう ひとうとう ううろうち うくうたいざい

| 0.00000005+00                                   | 0.00000026+00                                               |
|-------------------------------------------------|-------------------------------------------------------------|
| 0.0000005+00                                    | 0.00000036+00                                               |
| -0.13343345-02                                  | 0.56122482*32                                               |
| -0.10085445-01                                  | 0.1667035E-C2                                               |
| 0.2000065400                                    | 3. 9939090€+ 30                                             |
| 0.00390005440                                   | 3. 0633000€+00                                              |
| - 3.752233835-02                                | 0 - 1 3 9 > 7 1 7 E - 0 2                                   |
| - 3.23737325-01                                 | - C - 5 + 3 3 0 + 2 E - 0 3                                 |
| 00000000000000000000000000000000000000          | 00+00000000000000000000000000000000000                      |
| - ( • 5 • 6 • 7 • 7 • 7 • 6 • 6 • • • • • • • • | - 0 + 4 5 5 4 3 6 0 5 7 4 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 |

## PER CENT EFFLUTENCLES

## . . . . . . . . . . . . . . . . .

## されたってりてきょうの。 へいかやすいてもほ

0.15955532+31

#### GLOBAL EFFICIENCY

0.23653112+33

RELATIVE AGUEL EFFICIENCY:

## 0.40365505400

---SLUBALLY MINIMUM COST 54---1 0.51503975-02

---MECAL CEST S4---: 0.372-1716-02

GLOBAL SPILLOVER LUCTIENT: 0.42108365403

| $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | , , , , , , , , , , , , , , , , , , ,       |
|------------------------------------------------------|---------------------------------------------|
|                                                      |                                             |
| ·• ህጣ ፥ ሳለኰ መጠርንቀ/ሀ ጎ ፥ሳ ዐሎ መጠርቀ/ቁማ ፥                | まできょう 4 m m m m m m m m m m m m m m m m m m |
| · · · · · · · · · · · · · · · · · · ·                |                                             |
|                                                      |                                             |

è

u U

# 计设计程序 化林兰林不能用 一口可能已经多过能已没有过来。

| 00000<br>00000<br>000000<br>00000<br>00000<br>00000<br>0000 |        | 0 • 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 |    | 0.00000<br>0.00000<br>0.000000<br>0.0000000<br>0.000000 | 0 • 0 0 0 1 0 4 1 0 0 0 0 0 0 0 0 0 0 0 0 0 |
|-------------------------------------------------------------|--------|-----------------------------------------|----|---------------------------------------------------------|---------------------------------------------|
|                                                             | 11 0 F | n 7 0                                   | 11 | <br>- 6 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7             | ****                                        |

#### ACTOR VECTOR

| <br>.000055 | . 346912 | . JÜÇJIS | - 10001 - |   |       | • 3000 | 5:1:2:1:4 |      |      | <    | 1.0000 | 1    | * |      |         | ] • 3 3 4 4 4 |      |      |            |  |
|-------------|----------|----------|-----------|---|-------|--------|-----------|------|------|------|--------|------|---|------|---------|---------------|------|------|------------|--|
| <br>s<br>0  | 0        | 7 0      | 6<br>()   | د | 10 01 | 11 Ú   | 12 31     | 15 0 | 14 3 | 15 ) | 15 Ū   | . T. |   | C 61 | с<br>г: | 51 O          | C 72 | 23 Ú | 0<br>• • • |  |

0.001113 0.00092

-- ~I

 Tuftac
 Litutac

 Tuftac
 Litutac

 Tuftac
 Litutac

 Effect
 Litutac

UMBRATCR COVPULATE COSTS:

| 0 + 2232748110<br>0 + 2232748110<br>0 + 1240199110<br>0 + 1240199110<br>0 + 1150199110<br>0 + 1150199110                                                              |
|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| 0.1+1+1+C-1+<br>0.1+1+1+C-1+<br>0.0+1+1+1+C-1+<br>0.0+1+1+1+C-1+<br>0.0+1+1+1+C-1+<br>0.0+1+1+1+C-1+<br>0.0+1+1+1+C-1+<br>0.0+1+1+1+C-1+1+1+1+1+1+1+1+1+1+1+1+1+1+1+1 |
| 0                                                                                                                                                                     |

ENGRINATOR COMPONENT COSTOC

| 0.3552385-04<br>0.2552385-04<br>0.2552385-04<br>0.2552385-04<br>0.45217235-04<br>0.423757235-11<br>0.7733355-04                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                   |      |                                            |                 |    |                                           |                 |              |              |             |  |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|------|--------------------------------------------|-----------------|----|-------------------------------------------|-----------------|--------------|--------------|-------------|--|
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 11111111111111111111111111111111111111                                                                            |      | .2305315-02<br>? C34P. C3STS:              | 2               | 10 | n                                         |                 |              | -1           | 1           |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 68.680 (887947)<br>E. 021131601107<br>N15, 12, 060861<br>20088018604688                                           | - 11 |                                            | J.302555+(      |    |                                           | • 3 • 3 • 5 • 5 |              | J.127.Jui-   | しーこのまでいまけょう |  |
| Image: Non-State         Image: Non-State< | 84800480084644440949<br>8678.5015115 ADD<br>678.501462.50147124<br>678.501462.50147124<br>88844884488448844800000 |      | NP TH CROER EFFICIE<br>BLOCK-DIAS NUMERATS | 0.1133518-03 \$ |    | 6 + 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 | 0               | 0.2++1135-03 | 0.1200905-03 | 0.52420255  |  |

 DLUCATION 28-10-10-10-10-10-10-10-10-10-10-10-02-02

|              | 0.3051535-03 | <b>3.541730E-J2</b> | 0.1271475-92                          | 0.9780352-35                   |
|--------------|--------------|---------------------|---------------------------------------|--------------------------------|
| 10+55+5      | 9.3052235-73 | 0.2757545-02        | 0.2134105-03                          | C.121345E-16                   |
| C.ST : 0.195 | 0.2125535463 | 0.3511535-02        | 3.931257E-32                          | ).4538325-52                   |
|              |              | ·                   | 1 - 54117791012400**45400 - 148068726 | 4 301129404440**4200 098404080 |

ФС 1444 ФРАНТА № 00 ПО 14 ОСТИСТИ ОСТИСАТИ В ФОЛИСТ Матадори сосстание истиск состание и посили № 16 № 0 ПРАТОРИКА СОСАРИАТУ № ОВОРНАЦИО ОООСАН

| 31.30<-1.113.40149.4844.4<br>0.93125724-12 | 0.5+273-6+32 | <b>0.4</b> 535325-22 | 0.3511535-32 | 3 . 27 57 6 4 5 - 5 2 | <b>0 - 1 2 7 1 4 7 5 - </b> | J.3725505-33 | 0.3352632-33 | <b>0</b> .3C51535+03 | 0.213+135-33 | 0+97-00=1-03 | <b>0.1113456-10</b> |
|--------------------------------------------|--------------|----------------------|--------------|-----------------------|-----------------------------|--------------|--------------|----------------------|--------------|--------------|---------------------|
| MCDE(I440326)<br>7                         | 0            | 10                   | ł            | 2                     | ¢                           | 1            | 2            | Ē                    | Ŧ            | 16           | 11                  |

0 e

|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                 | 1        |                           |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------|----------|---------------------------|
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                 | 1        |                           |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                 |          |                           |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                 | 0        |                           |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                 |          |                           |
| and the state of t | しょう そうらで うちら ふて ひみきょう ひょう オロロ                   |          | n - 1 +                   |
| • • • • • • • • • • • • • • • • • • •                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ししし しょうしょう しょうしょう ちょうしょう しょうしょう しょうしょ           | г        |                           |
| 1: 0 1: 1 N N N N N N N                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 1 1 1 3 4 7 4 9 9 9 9 9 9 9 9 9 9 9 9 7 7 7 7 7 | <b>.</b> | - <b>1</b> · n - <b>n</b> |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                 |          | 1 4 4 4 1 1               |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 000000000000000000000000000000000000000         |          | 4 (3 (5                   |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 0444000000000000000000000000                    |          | ר היה                     |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                 |          |                           |
| 1100111000                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 000000000000000000000000000000000000000         |          | 000                       |
| · · · ·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                 |          |                           |
| n or non-company and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                 |          |                           |
| ······································                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                 |          |                           |

# LARACTERISTIC EFFICIENCIES

| 0.14.1.7 | 0.3023,5 | 0  | 0.0Jiss5 | 3.315357   | 2.315255 | 0.315127 | 106200.0 | 6.243535 | G.Jusii7 | 0.47.00 | ū.Jūč. |  |
|----------|----------|----|----------|------------|----------|----------|----------|----------|----------|---------|--------|--|
| -4       | ~        | 'n | ,        | , <b>n</b> | 'n       | 7        | 'n       | Ţ        | 1        | 11      | 12     |  |

|                                                                                            |                                                  | 17                   |     |
|--------------------------------------------------------------------------------------------|--------------------------------------------------|----------------------|-----|
|                                                                                            |                                                  |                      |     |
|                                                                                            |                                                  |                      |     |
|                                                                                            |                                                  | L.                   |     |
|                                                                                            |                                                  | 40                   |     |
|                                                                                            |                                                  | 01 H 10              |     |
|                                                                                            |                                                  | 000                  |     |
|                                                                                            |                                                  | 1 + 0                |     |
|                                                                                            |                                                  | 101.00.00            |     |
|                                                                                            |                                                  | FF                   |     |
|                                                                                            |                                                  |                      |     |
|                                                                                            |                                                  |                      |     |
|                                                                                            |                                                  | <b>N</b> -1          |     |
|                                                                                            |                                                  | 1 C 4                |     |
|                                                                                            | -                                                | • •                  |     |
|                                                                                            | <b>*</b>                                         | 100                  |     |
|                                                                                            | د                                                | 1                    |     |
|                                                                                            |                                                  |                      |     |
|                                                                                            | P1                                               | 01 0                 |     |
|                                                                                            | <b>*</b> 4                                       | 2 J 2                |     |
|                                                                                            |                                                  | () ()                |     |
|                                                                                            | 6-4                                              | <b>O</b> I ·····     |     |
|                                                                                            | 1                                                | •                    |     |
|                                                                                            | u.                                               | 0.001+++-13          |     |
|                                                                                            | iu iii iii iii iii iii iii iii iii iii           | 12 12 12             |     |
|                                                                                            |                                                  | <u>n (1 (1 (7</u>    |     |
|                                                                                            |                                                  | 100                  |     |
|                                                                                            |                                                  |                      | 11  |
|                                                                                            |                                                  |                      |     |
|                                                                                            |                                                  |                      | -   |
|                                                                                            |                                                  | <b>H H H H H</b>     | •   |
| 1 1 - 1 1 1 1 - 1 1 1 1 1 - F                                                              |                                                  |                      |     |
| 5 1 1 1 1 1 1 4 5 1 1 1 St F                                                               |                                                  | 27 H (1 (1 (1 (1     | •   |
| E GENERAL AND                                          | <b>FINMMMM4444400000000000000000000000000000</b> | - J D U U F          | 11  |
| きょうしき ちょう きょうりのしません きょぎ                                                                    | 00000000000000000000000000                       | 141 <b>1</b> 4 4 4 4 | fa. |
| 4 4 7 7 7 4 7 4 7 7 7 4 7 7 7                                                              | ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,           | • • > : < •          | r   |
| 0 <b>0 0 0 0 0 0</b> 0 0 0 0 0                                                             |                                                  | 14 4 F F             | 11  |
|                                                                                            | • • • • • • • • • • • • • • • • • • • •          | 1. 2. 5 1 1 2        | • • |
| $\mathbf{v} \mathbf{v} \mathbf{v} \mathbf{v} \mathbf{v} \mathbf{v} \mathbf{v} \mathbf{v} $ | 00000000000000000000000000000000000000           | 12 +4 2 - *          |     |
|                                                                                            |                                                  |                      |     |
|                                                                                            |                                                  |                      | *   |
|                                                                                            |                                                  |                      |     |
|                                                                                            |                                                  |                      |     |
| n + n c > m = 0 = 1 m +                                                                    |                                                  | 51111                |     |
| 4                                                                                          |                                                  | <b>.</b>             | π   |
|                                                                                            |                                                  |                      | 5   |
|                                                                                            |                                                  |                      |     |
|                                                                                            |                                                  |                      |     |

6.2377636-05 0.3524936-03 0.2759675+33 0.527+53675+33

| .466612E-05<br>.133275E-03<br>.112413E-03                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                    |                                                                                                                                       |       |
|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------|-------|
| C.919622E-04<br>0.143655-04<br>0.3515995-04<br>0.243333504<br>0.243333504<br>0.243333504                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                    |                                                                                                                                       |       |
| C 550 2347-02<br>C 72 51 16-01<br>C 72 51 16-01<br>C 718-01<br>C 10 - 10 + 10<br>C 10 - 10 + 10<br>C 10 - 10 + 10<br>C 10 + 10 + 10<br>C 10 + 10 + 10<br>C 10 - 10 + 10<br>C 10 - 10 + 10<br>C 10 - 10 - 10<br>C 10 | 4004000000000<br>5 TC TLE Vo<br>140 OPCEN:<br>4040000000000000000000000000000000000                                                | .40055555-02<br>R CC 4P. CESTS:<br>01<br>01<br>03<br>03<br>01<br>01<br>01<br>01<br>01<br>01<br>01<br>01<br>01<br>01<br>01<br>01<br>01 | 1 1 1 |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 50000100010000000000000000000000000000                                                                                             |                                                                                                                                       |       |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 4844448888888888888888888<br>METNUUL:CCUPLEO MEDI<br>EFFNUKÉR COXPC14<br>EFFNUKÉR COXPC14<br>#848888888888888888888888888888888888 | MCJEKIN MCJSEL)<br>10<br>10<br>12<br>12<br>12<br>12<br>12<br>12<br>12<br>12<br>12<br>12                                               |       |